San Mateo County Parks SAN BRUNO MOUNTAIN PARK

DESIGN-BUILD AGREEMENT for RESTROOM REPLACEMENT PROJECT

THIS AGREEMENT, dated this 13th day of August 2024 by and between Romtec, Inc., whose place of business is located: 18240 North Bank Road, Roseburg, OR 97470 ("Design-Build Entity" or "DBE"), and the County of San Mateo ("Owner"), acting under and by virtue of the authority vested in Owner by the laws of the State of California for preconstruction and construction services ("Work") for the San Bruno Mountain Park Restroom Replacement Project ("Project") in accordance with the Contract Documents. By executing this Agreement, each of the Signatories represents that they have the authority to bind the Party on whose behalf this execution is made.

Owner: County of San Mateo 455 County Center, 4 th Floor Redwood City, CA 94063	Design-Builder: Romtec, Inc. 18240 North Bank Road Roseburg, OR 97470
By: (Signature)	By: Buyanin (soper 7/18/2024 (Signature)
Name:	Name: Ben Cooper, President
Telephone No.:	Telephone No.: 541-496-3541
Facsimile No.:	Facsimile No.:
Email:	Email: sales@romtec.com
	CA License No.: 849246
	DIR Registration No.: 1000002582

THE PARTIES AGREE TO THE FOLLOWING TERMS AND CONDITIONS

E.

EXHIBITS

All Exhibits set forth below are incorporated into the Agreement.

Exhibit 1	Supplemental Conditions
Exhibit 2	Bridging Contract Documents
Exhibit 2A	San Bruno Mountain Park Conceptual
	Design
Exhibit 2B	Criteria Document
Exhibit 2C	Site Survey with underground utilities
Exhibit 2D	Geotechnical Report
Exhibit 2E	Hazardous Material Report
Exhibit 3	Scope of Work
Exhibit 3A	Scope of Work
Exhibit 3B	Exclusions & Assumptions
Exhibit 4	Price Proposal
Exhibit 4A	Price Proposal
Exhibit 4B	Schedule of Values
Exhibit 5	Personnel & Equipment
Exhibit 5A	Staffing Plan
Exhibit 5B	Key Personnel
Exhibit 6	Schedule
Exhibit 6A	Project Baseline Schedule
Exhibit 7	Schematic Design Documents

Table of Contents

1.	GENERAL	1
2.	THE DESIGN-BUILD TEAM AND RELATIONSHIP OF THE PARTIES	1
3.	CONTRACT DOCUMENTS	3
4.	OWNER'S OBLIGATIONS	4
5.	DESIGN-BUILDER'S OBLIGATIONS	5
6.	SUBCONTRACTORS	.15
7.	COMPENSATION	.16
8.	CONTRACT TIME	.18
9.	CHANGES	.20
10.	PAYMENT	.22
11.	INDEMNIFICATION	.25
12.	INSURANCE AND BONDS	.27
13.	WARRANTY OF THE WORK	.30
14.	OWNERSHIP AND USE OF DOCUMENTS	.32
15.	ACCOUNTING RECORDS	.33
16.	TERMINATION, SUSPENSION AND ABANDONMENT	.34
17.	MISCELLANEOUS PROVISIONS	.35

DESIGN-BUILD AGREEMENT

This Design-Build Agreement ("**Agreement**") is executed as of <u>August 13, 2024</u> ("**Effective Date**") by and between the "**Owner**" and "**Design-Builder**" for completion of the "**Project**." The Owner and Design-Builder may collectively be referred to as "**theParties**."

Owner:

County of San Mateo 455 County Center, 4th Floor Redwood City, CA 94063

Design-Builder:

Romtec, Inc. 18240 North Bank Road Roseburg, OR 97470

Project:

San Bruno Mountain Park Restroom Replacement Project No. 24-01

The Owner and Design-Builder agree as set forth below:

1. GENERAL

- **1.1 Definitions.** All defined terms will be capitalized throughout the Agreement. The definitions for this Agreement appear in alphabetical order in Section 1 of the "**Supplemental Conditions**" to the Agreement and may also be set forth herein for convenience as defined terms the first time the term is used.
- **1.2 Project Description.** The Project involves **Preconstruction Stage Services and Construction Stage Services** to construct the replacement of a toilet room building ("Project") according to the San Bruno Mountain Park Map and Project Criteria Documents attached hereto as **Exhibits 2A and 2B** respectively.
- **1.3 Project Delivery.** The Project will be delivered using a design-build delivery method pursuant to Sections 22160, et seq., of the California Public Contract Code.

2. THE DESIGN-BUILD TEAM AND RELATIONSHIP OF THE PARTIES

2.1 Design-Build Team. The Design-Builder is responsible to the Owner for completion of the Project. The Design-Build Team includes all team members providing services for or on behalf of the Design-Builder, and is comprised of, at a minimum, the "General Contractor," the "Design Professionals" and "Design-Build Subcontractors". All "Design Services" will be performed by the Architect of Record and/or other design consultants (collectively, "Design Professionals"). The Design-Builder shall name the Owner as a third-party beneficiary to all design service agreements and/or design-build subcontracts and the parties agree that the Owner is an intended third-party beneficiary of such contracts. The Design-Build Team is currently comprised of the entities identified in Exhibit 5B Key Personnel.

- 2.2 Licensing. Design-Builder must possess a valid California state class "B" General Building Contractor license during the entire term of this Agreement. All members of the Design-Build Team must possess the appropriate California state design licenses for their particular discipline. Subcontractors must all possess the appropriate California state specialty license for their particular trade. Nothing in this Agreement will require a Design-Build Team member, or any of their respective Consultants or Subcontractors, to perform any portion of the Work outside of their respective licenses or contrary to applicable law.
- 2.3 Good Faith. The Design-Builder will perform all Work under this Agreement in compliance with each of the following requirements: (i) use its best skill and judgment in pursuit of the Project; (ii) furnish effective and efficient design, construction administration and supervision; (iii) furnish at all times an adequate supply of "Skilled Labor" and materials; and (iv) perform the Work in the most expeditious and economical manner consistent with the Bridging Contract Documents in Exhibit 2, and good engineering practices.
- 2.4 Standard of Care. The Design-Builder warrants that it possesses the design and construction licenses and expertise required for this Project under Section 2.2 and will use the same degree of care and skill customarily used by California state licensed professionals performing similar services for residential construction projects in the state of California.
- 2.5 Collaboration. Owner and Design-Builder commit at all times to cooperate fully with each other and proceed on the basis of trust and good faith to permit each party to realize the benefits afforded under this Agreement. Design-Builder and its Design Professionals, Subcontractors, suppliers, and equipment vendors will perform their respective portions of the Work using collaborative tools and methods. The Design-Build Team will actively participate and collaborate with Owner to achieve best value, optimal design, increased labor efficiency, and elimination of waste and re-work. The Design-Builder will collaborate with Owner to develop the design within the Contract Price, and to ensure that the design satisfies the Bridging Contract Documents.
- **2.6 Communications.** All communications from the Design-Builder shall be directed to Owner via the Owner's Project Manager, Matthew Estes, and others as designated and directed by the Owner's Project Manager.
- **2.7 Relationship of the Parties.** The Design-Builder's relationship with the Owner is that of an independent contractor whose involvement in the Project is to act solely in the capacity of a California licensed design professional and general contractor and not as an agent, fiduciary, partner, member of, subsidiary of, or otherwise affiliated with the Owner.
- **2.8 Responsibility.** Design-Builder acknowledges and agrees that it is solely responsible to Owner for the sufficiency, quality, adequacy and completeness of the Work, and that Design-Builder is responsible for any acts, errors, or omissions of the Design-Builder's principals, employees, agents, and/or any other parties either directly or indirectly in privity of contract with Design-Builder including, but not limited to, the Architect of Record and other Design Professionals, Subcontractors, suppliers, equipment vendors, and their agents and employees, and other persons performing any portion of the Work on behalf of Design-Builder.

2.9 Conflicts of Interest. Design-Builder warrants that it is not aware of any existing conflicts of interest under Applicable Law that would prevent any member of the Design-Build Team from participating in the Project. Design-Builder has an ongoing obligation to monitor and disclose conflicts or potential conflicts of interest. If an organizational conflict of interest is discovered, the Design-Builder must make an immediate and full written disclosure to the Owner that includes a description of the action that the Design-Builder has taken or proposes to take to avoid or mitigate the conflict. If the contract is terminated due to a conflict of interest that existed at the time of the award, the Owner has no obligation, responsibility or liability to reimburse all or part of the costs incurred or alleged to have been incurred by the Design-Builder.

3. CONTRACT DOCUMENTS

- **3.1 Contract Documents.** The "**Contract Documents**" consist of this Agreement, the Supplemental Conditions, and all other Exhibits attached to this Agreement, all subsequent modifications through amendments and change orders executed by Owner and Design-Builder, and the Construction Documents to be developed by the Design-Builder.
- **3.2** Interpretation and Intent. The intent of the Contract Documents is to include all items necessary for proper completion of all Work within the "Contract Time" and within the "Contract Price." The Contract Documents are intended to be complementary and what is required by any one of them is as binding as if called for by all of them.
- 3.3 Sufficiency of Contract Documents. The Design-Builder acknowledges that all documents and materials submitted by the Owner to the Design-Builder in connection with the process culminating in the execution of this Design-Build Agreement, are complete and sufficient to have enabled the Design-Builder to determine the cost of the Work in order to enter into this Agreement. The Design- Builder confirms that it has examined the site and all physical, legal and other conditions affecting the Work and is fully familiar with the site and with such conditions. The Design-Builder specifically represents to the Owner that it has examined (a) the nature, location, and character of the Project and the site, including, without limitation, the surface conditions of the site and subsurface conditions of the site to the extent that such conditions affect the design and constructability of the Project, and all structures and obstructions on the site and thereunder, both natural and man-made, and all surface and subsurface water conditions of the site and the surrounding area; (b) the nature, location, and character of the general area in which the Project is located, including without limitation, its climatic conditions, available labor supply and labor costs, and available equipment supply and equipment costs; and (c) the quality and quantity of all materials, supplies, tools, equipment, labor, and professional services necessary to complete the Work in the manner and within the cost and time required by the Contract Documents. In connection with the foregoing, and having carefully examined all Contract Documents, and having examined the site, the Design-Builder acknowledges and declares that it has no knowledge of any discrepancies, omissions, ambiguities or conflicts in the Contracts Documents and agrees that if it becomes aware of any such discrepancies, omissions, ambiguities or conflicts, it shall promptly notify the Owner thereof.
- **3.4 Order of Precedence.** In the event of inconsistencies between requirements contained in different components of the Contract Documents, the content of each

document listed below prevails over any inconsistent content in any document listed below it:

- **3.4.1** Amendments of the Design-Build Agreement;
- **3.4.2** Change Orders approved by Owner;
- **3.4.3** The Design-Build Agreement executed between Owner and Design-Builder not including Exhibits;
- **3.4.4** Project Baseline Schedule, as approved by Owner and updated pursuant to Section 8.2 of this Agreement;
- 3.4.5 The Scope of Work set forth in **Exhibit 3** to the Agreement;
- 3.4.6 The Supplemental Conditions included in **Exhibit 1** to the Agreement;
- **3.4.7** The Bridging Contract Documents included as **Exhibit 2**;
- **3.4.8** 100% Construction Documents developed by Design-Builder;
- **3.4.9** All other Exhibits to the Design-Build Agreement and all other Contract Documents not listed above;
- **3.4.10** Proposal submitted by Design-Builder;

4. OWNER'S OBLIGATIONS

- **4.1 Information and Documents.** The Owner may make various Background Documents related to the Project available to the Design-Builder, including but not limited to any surveys and other information that describe the Project Site as well as schedule requirements, budget constraints and other criteria, and procurement schedules. Any Background Documents provided are for information only and will not be included as part of the Contract Documents.
- **4.2 Bridging Contract Documents.** The "Bridging Contract Documents," consisting of the Design Criteria, both included in **Exhibit 2** to this Agreement, were developed by the Owner to provide an understanding of the baseline design requirements for the Project. The Bridging Contract Documents, along with other information provided during the RFP and proposal process, provides a basis for the Contract Price, initial Project Baseline Schedule, and initial design work. The Design-Builder must conduct all Work in accordance with the Bridging Contract Documents.
- **4.3 Ownership of Facilities.** The Owner will own the Facilities, and control easements on which certain Facilities are to be built. Owner will provide Design-Builder with access to the Work site and easements for the purpose of fulfilling its obligations under this Agreement.
- **4.4 Governmental Approvals.** The Design Builder will be responsible for obtaining the permits and approvals for the Project facilities. Owner shall reimburse the Design Builder without mark-up for all permits and fees associated with this work. Design-Builder's responsibility for permits, licenses, and approvals is set forth in Section 5.5.

- **4.5 Owner's Project Manager.** Owner's interests on the Project will be represented by the Owner's Project Manager, Matthew Estes, as well as any other individuals identified from time to time by the Owner. The Project Manager is authorized to act on the Owner's behalf with respect to the daily operations of the Project, including, without limitation, review of Work, invoices, claims, change orders, and submittals, or may delegate authority to another representative. For simplicity, where this Agreement refers to the Owner, Design- Builder may assume that the Project Manager is the appropriate point of contact. Where necessary, the Project Manager will elevate issues to the County Board of Supervisors or to appropriate executives.
- **4.6 Stop Payment Notice.** The Owner will comply with all stop payment notices submitted in compliance with applicable laws by withholding appropriate amounts from payments otherwise due to Design-Builder or otherwise responding consistent with legal requirements.
- 4.7 Separate Contracts. The Owner reserves the right to perform construction, maintenance, and operations related to the Project with the Owner's ownforces, and to award contracts for work that lies outside of the Design-Builder's Project Scope of Work. The Design- Builder and Owner will coordinate to allow for any other separate contracts to be performed, and to minimize interference with the Work and the work that the Owner is having performed through separate contract or contracts. Design-Builder shall perform all Work in such a manner as to avoid any material interruption of Owner's existing operations, including, without limitation, use of the athletic fields. When performing construction, maintenance, or operations related to the Project, the Owner agrees that its separate contractors will be subject to the same obligations as the Design- Builder with respect to insurance, indemnification, safety, protection, inspections and non-conforming work. The Owner will remain responsible to the Design-Builder for any delays to the Contract Time or cost impacts resulting from work performed by its separate contractors. Any cost and/or time impacts will be addressed through the Change Order process set forth in Section 9.
- **4.8 Timeliness.** In order to avoid any impacts to the Contract Time, information or services under the Owner's control, including reviews and approvals, will be furnished within the timeframes set forth in the Contract Documents.
- **4.9 Owner Direct Payments.** In case of a material breach by the Design-Builder, the Owner hereby retains the right to make direct payment to Subcontractors and Design Professionals, less retention, and to deduct the amounts from future payment requests from Design-Builder. Owner shall give Design-Builder notice and a reasonable opportunity to cure the material breach before exercising any rights described in this Section 4.9.

5. DESIGN-BUILDER'S OBLIGATIONS

5.1 Design-Build Services. Design-Builder will provide all labor, materials and equipment necessary to complete the Work in compliance with the Contract Documents as described in further detail in the Scope of Work included in **Exhibit 3**. Unless otherwise provided in the Contract Documents, the Design- Builder shall provide or cause to be provided, and shall pay for services, labor, materials, equipment, tools, construction equipment and machinery, water, heat, utilities, transportation and other facilities and services necessary for proper execution and completion of the Work,

whether temporary or permanent and whether or not incorporated or to be incorporated in the Work.

- 5.2 Preconstruction Stage Services Preparation of Design and Construction Documents. Design-Builder will work collaboratively with Owner to validate Conceptual Design (Exhibit 7), propose alternatives where such alternatives create value to the Owner, evaluate design and construction phasing alternatives, and potential early work packages. Design Builder to produce Schematic, Design Development, and Construction Documents for the Project in accordance with: Criteria Documents (Exhibit 2), its attachments, and supporting documents: Site Survey, Geotechnical Report, and Hazardous Material Report; and Scope of Work (Exhibit 3). Also complete design per meetings with the County for input, conformance with Criteria Documents, and approval at each design phase for the schematic design, design development, construction phase documents.
 - **5.2.1 Conduct of Design Services.** Design-Builder must perform all designservices, including architectural, engineering and other design professional services, consistent with applicable state licensing laws, and through qualified, licensed design professionals employed by Design-Builder, or procured from qualified, independent licensed Design Professionals, and in accordance with the standard of care described in Section 2.4. The Design-Builder's Architect shall be the Architect-of-Record.
 - **5.2.2** Schematic Design. Prior to starting design, DBE is to conduct site investigations and familiarize themselves with the existing conditions and supporting documents provided by the County. Prepare Schematic Design Documents based on the Conceptual Drawings, site observations, Criteria Document and supporting documents. Include site layout drawings for approval by the County. Attend review meetings with the County and incorporate conformance review comments.
 - **5.2.3 Design Development.** Upon the County's written acceptance of the Schematic Design Documents, prepare Design Development Documents for approval by the County from the Criteria Documents, including related architectural, structural, mechanical, electrical, plumbing, and civil plans. Attend review meetings with the County and incorporate conformance review comments.
 - **5.2.4 Construction Documents**. Upon the County's written acceptance of Design Development Documents, prepare 100% complete construction documents for review and written approval by the County prior to submitting for permitting. Perform quality assurance/quality control and constructability reviews of the documents prior to submitting to the County. The County will perform its own constructability and conformance reviews. The 100% Construction Documents shall consist of complete detailed working drawings and Specifications addressing required materials, products, equipment, their installation and operation, quality assurances, reference standards, product data, warranty data, etc. These 100% Construction Documents shall contain all required drawings and Specifications completed in detail sufficient to construct the Project, confirm conformance with Bridging Contract Documents, and obtain agency approvals. The 100% Construction

such submissions may have been modified in a design review meeting and recorded in the meetings minutes. Design-Builder shall remain responsible for correcting any deviation from the Bridging Contract Documents, whether discovered prior to or during construction work. Attend meetings with the County and incorporate review comments prior to submitting for permitting purposes.

- **5.2.5 Permitting**. Submit completed accepted plans and specifications, and obtain all local jurisdictional agency plan approvals and permitting, necessary to meet the Project Schedule (**Exhibit 6**). DBE to pay and be reimbursed without mark-up for all permitting fees.
- **5.2.6** Any other services that are reasonable and necessary for design and permitting of the Project, including close-out with local jurisdictional agencies.
- 5.3 Construction Stage Notice to Proceed. Prior to commencing any work on the Construction Stage of the Project, the Design-Builder will submit a Final Design Package to Owner that Design-Builder proposes would govern the Construction Stage work. The Final Design Package shall be comprised of the following documents: 1) the 100% Construction Documents; 2) a Construction Stage project schedule; 3) all documents required as part of the Project Manual for Construction Stage Work, including a number of forms and plans (Safety Plan, Traffic Control Plan, Quality Control Plan, various mitigation plans) to be identified by Owner and developed by Design- Builder during the course of the Preconstruction Stage; and 5) any other documents or materials reasonably required by Owner. Design-Builder must submit one (1) electronic set and (5) sets of prints. Owner shall review the Final Design Package in order to determine whether this Design-Build Agreement provides Owner with the best value for completing the Construction Stage of the Project. Owner reserves the right to take any of the following actions in response to the proposed Final Design Package submitted by Design-Builder.
 - **5.3.1** Owner may elect to proceed with the Construction Stage of this Agreement by delivering to the Design-Builder a written Notice to Proceed with Construction (the "NTP"), Builder elect to proceed with the Construction (in reasonably acceptable form), Building Permit and Clearance of Bird Nesting. Design-Builder will coordinate in good faith and in a commercially reasonable manner with respect to securing the Building Permit and the Clearance of Bird Nesting.
 - **5.3.2** The Owner may notify the Design-Builder of any objections to the Final Design Package within fourteen (14) calendar days after their submittal. In the event that the Owner makes objections to the 100% Construction Documents, the Design-Builder may complete, correct and/or modify the 100% Construction Documents in question and resubmit the Final Design Package to the Owner. If the need for re-submittal of the Final Design Package (or any part of it) shall not be due to a change requested by the Owner in the Final Design Package, then the Design-Builder shall have ten (10) calendar days within which to correct, complete and re-submit the Final Design Package, but there shall be no extension of the dates in the Project Baseline Schedule. In the event the Owner shall request any change in the Final Design Package that represents a change in the Scope of Work, such

request may require an adjustment of time and compensation pursuant to a ChangeOrder.

- **5.3.3** The Owner may elect not to proceed with the Construction Stage of this Agreement in its sole discretion, and may take such action without cause and for its own convenience whether or not the Owner elects to have the project constructed, terminate the Agreement with Design- Builder for convenience in accordance with Article 16.3, and take possession of the 100% Construction Documents and all other design documents and related work product developed by Design-Builder for potential award to a separate contractor.
- **5.4 Construction Stage Services.** Design-Builder shall proceed to execute and complete the Construction Stage services only upon issuance by the Owner to the Design-Builder of a NTP and other Documents set for the in Section 5.3.1 of this Agreement with the construction phase of the Work. Design-Builder will provide all Construction Stage Services required for the Project. Design-Builder's construction stage services will also include each of the responsibilities summarized below.
 - **5.4.1** Unless otherwise provided in the Contract Documents to be the responsibility of Owner or a separate contractor, Design-Builder shall provide through itself or Subcontractors the necessary supervision, labor, inspection, testing, start-up, material, equipment, machinery, temporary utilities and other temporary facilities to permit Design- Builder to complete the Construction Stage Services consistent with the Contract Documents. Design-Builder is responsible for demolition and removal of all existing improvements necessary for construction of the Project including Hazardous Materials and Substances identified in **Exhibit 2E** in the Work.
 - **5.4.2** Design-Builder shall perform all construction activities efficiently and with the requisite expertise, skill and competence to satisfy the requirements of the Contract Documents. Design-Builder shall at all times exercise complete and exclusive control over the means, methods, sequences and techniques of construction.
 - **5.4.3** Design-Builder shall coordinate the activities of all Subcontractors. If Owner performs other work at the Site with separate contractors under Owner's control, Design-Builder agrees to reasonably cooperate and coordinate its activities with those of such separate contractors so that the Project can be completed in an orderly and coordinated manner without unreasonable disruption to the Work or the work that the Owner is having performed by separate contractors.
 - **5.4.4** Design-Builder shall fully comply with all environmental and permit mitigation requirements set forth in these Contract Documents, including, without limitation, remediation of all hazardous materials, including preexisting hazardous materials, at the Project Site.
 - **5.4.5** Design-Builder must promptly remove from the Project Site, or from property adjacent to the site of the Work, all unused and rejected materials, surplus earth, concrete, plaster, and construction waste, including waste from demolition of existing structures and improvements, to permit Design-Builder

to perform its Construction Stage Services efficiently, safely and without interfering with the use of adjacent property. In particular, the Design-Builder shall keep the Project Site clean to maintain safe access and to avoid fire hazard. Upon Substantial Completion of the Work, or a portion of the Work, Design-Builder shall remove all debris, trash, construction waste, materials, equipment, machinery and tools arising from the Work or applicable portions thereof to permit Owner to occupy the Project for its intended use. Upon Substantial Completion of the Work, Design-Builder shall return the premises to its pre-existing condition or better, based on a preconstruction survey to be performed by Design-Builder.

- **5.5 Governmental Approval Services.** Design- Builder is responsible for obtaining all permits, licenses, and approvals necessary for the completion of the Work.
 - **5.5.1 Review of Approvals.** Owner reserves the right to review any submittals and final terms and conditions of permits, licenses, and approvals obtained pursuant to this Section 5.5, and to deal directly with any agencies responsible for the approvals. Design- Builder will be entitled to an extension of time to the extent that a delay is caused by Owner's unreasonable delay in reviewing and/or approving such approvals.
 - **5.5.2 Permit Documents.** Only documents prepared for or by Design-Builder for this project may be used for obtaining building permits for construction. No drawings or specifications prepared by the Owner or by the Project Manager or by their representatives shall be used for permits or construction without the Owner's and the Project Manager's prior written permission in each instance.
- **5.6 Project Support Services**. Owner may require Design-Builder to provide other incidental services relating to the Project, including, without limitation, public outreach, and presentations at Board of Supervisor meetings. The parties acknowledge and agree that, to the extent that the services described in this Section 5.6 are, in fact, incidental, the Design-Builder shall perform them without additional compensation. In the event that the Owner requests services pursuant to this Section 5.6 and the Design-Builder asserts that such services are not incidental, the Design-Builder shall promptly, and before performing any such work, notify the Owner of its contention and shall provide any supporting documentation. Owner shall promptly respond to any such notice from the Design-Builder and the parties shall meet and confer in good faith regarding any disagreements with respect to services under this Section 5.6.
- **5.7 Site Investigations.** By executing this Agreement, the Design-Builder represents that it has visited the Project premises, and is familiar with the local conditions under which the Work is to be performed.
- **5.8 Test and Inspections.** The Design-Builder shall be responsible for and coordinate any and all inspections required by any governmental body that has jurisdiction over the Project. Failure to obtain any permits, licenses, or other approvals because of the failure of the Design-Builder to conform to this paragraph will not extend the Contract Time and the contractor shall not be entitled to an increase in the Contract Price therefore. Further, the Design-Builder shall be liable to the Owner for any financial damage such failure may cause the Owner. The Owner will pay for all testing and inspection including the special inspections, structural, mechanical, chemical, air and

water pollution tests, tests for hazardous materials, and other laboratory and environmental tests, inspections and reports required by law or the Contract Documents, however, the Design-Builder shall be responsible for costs related to any tests required for corrective work.

- **5.9 Coordination with Owner and Owner's Separate Contractors.** The Design- Builder will coordinate its Work with any of Owner's employees or contractors performing work in the vicinity of the Project Site.
- 5.10 **Sole Responsibility.** The Design-Builder acknowledges and agrees that it is solely responsible to the Owner for the sufficiency, quality, adequacy and completeness of all services performed by the Design-Builder, including, without limitation, design work (whether during the Preconstruction Stage or Construction Stage), and construction services. Design- Builder is responsible for any acts, errors, or omissions of the Design-Builder, its Design Professionals, its Subcontractors, employees, agents, and/or any other parties either directly or indirectly in privity of contract with Design-Builder including, but not limited to, the "Design-Build Team" identified in Section 2.1, second tier-subcontractors, and vendors who are performing any portion of the Work. The Design-Builder's design must meet the minimum design requirements as defined by the Bridging Contract Documents, and all other design requirements included in the agreement. The Owner may review (at its sole discretion) and as it may deem necessary or desirable, the design at specific design development milestones for consistency and compliance with such design requirements. If the Owner shall elect to review any such documents, it shall be entitled (but not obligated) to limit its review to a cursory review or to such review as may be required to enable the Owner to determine rate of progress. Owner's review and/or approval submittals, including, without limitation, interim and final design submittals, does not reduce Design-Builder's obligations under this section.
 - **5.10.1** Nothing in this Agreement shall relieve the Design-Builder of its obligations to complete the Project in full accordance with all applicable laws and regulations and suitable for the Owner's intended purposes.
 - **5.10.2** The Design-Builder, on behalf of itself and its design Team, specifically acknowledges and agrees that the Owner shall have the discretion to determine whether the 100% Construction Documents comply with the requirements of the Contract Documents.

5.11 Applicable Laws.

- **5.11.1** Statutory Authority. The Owner is awarding this Project pursuant to the design-build authority provided under Sections 22160, et seq., of the California Public Contract Code. The Design-Builder and the Owner acknowledge that they have reviewed this statutory authority, are familiar with all requirements, and will comply with applicable requirements and duties.
- **5.11.2** Compliance with All Applicable Laws. The Design-Builder and the Owner agree to comply with all Federal, State, Municipal and local laws, ordinances, rules, regulations, building codes and standards, orders, notices and requirements applicable to proper design and construction of this Project.

- **5.12** Staffing Plan and Key Personnel. The Design-Builder agrees that it will staff this Project in accordance with the staffing plan included in **Exhibit 5A**. The Staffing Plan will include a staff-level organizational chart indicating hierarchy and reporting responsibilities, as well as all Key Personnel.
 - 5.12.1 Key Personnel. The Design-Builder will identify all "Key Personnel" in Exhibit 5B and will not remove any of its Key Personnel from this Project without the express written consent of the Owner, except for death, disability or departure of person from employment. The Owner's Project Manager will be able to request the removal of any person employed by the Design-Builder whom it believes is incompetent, improper or a hindrance to the design-build process. If any Design-Builder personnel become unavailable to work on the Project, or if the Owner requests that an employee be removed, the Design-Builder will propose a replacement person within 10 business days for approval by the Owner. The recommended replacement person will have similar or better qualifications and experience, and must be approved in writing by the Owner. Additional Staffing and Key Personnel requirements are set forth in the Scope of Work in Exhibit 3.
- **5.13 Safety.** The Design-Builder is the "Controlling Employer" as defined by Cal/OSHA and will be responsible for initiating, maintaining and supervising all safety precautions and programs in connection with the Work and in accordance with the Design-Builder's Health and Safety Program. The Design-Builder shall comply with all statutory safety requirements.
- **5.14 Project Baseline Schedule.** The initial Project Baseline Schedule, is attached as **Exhibit 6A**. This initial Project Baseline Schedule will be regularly updated in conjunction with Section 8.2 of this Agreement (at least monthly) and refined as project development and design proceeds. The updates to the initial Project Baseline Schedule will coordinate and integrate the progress of Design Services with milestone dates for design development and Construction Documents, procurement of long lead items and Subcontracts, and construction of major components of the Project. The initial Project Baseline Schedule, and all subsequent versions, shall include all major components of the Work and the Owner's occupancy requirements projecting milestones for each of the following:
 - Completion of Schematic Design
 - Completion of 100% Construction Documents
 - Submission of Final Design Package
 - Issuance of Construction Stage Notice to Proceed
 - "Substantial Completion" of the Project including Issuance of Final Certificate of Occupancy for the Project
 - "Final Completion" of the Project

The Project Baseline Schedule must be prepared utilizing CPM scheduling software. The schedule must be broken down by activity and duration highlighting the critical path.

The schedule submitted by the Design-Builder and agreed to by the Owner shall be referred to as the Project Baseline Schedule, and shall not be adjusted except through Change Order under Section 9 for permitted delays as defined in Section 8. The

Project owns all float in the Project Baseline Schedule and monthly updates. Therefore, there will not be any adjustments to the Contract Time until all Project float is exhausted and the critical path is impacted.

- **5.15 Taxes and Fees.** The Design-Builder will pay all sales, consumer, use, gross receipts, and other similar taxes legally related to the work enacted at the time of Contract Award.
- **5.16 Consultants and Laboratories.** The Design-Builder will make recommendations to the Owner regarding selecting, retaining and coordinating any additional professional services, special consultants and testinglaboratories required for the Project.

5.17 Construction Administration.

- **5.17.1 Preliminary Notices.** Within 5 business day's receipt, Design-Builder will forward to the Owner all Preliminary Notices (Civil Code sections 9300, 9500 and 9560) served on it by any person or entity entitled to assert a payment bond or stop payment notice claim. Design-Builder will maintain a written record of all Preliminary Notices received by it including the manner of receipt, date of receipt, and name and address of person or entity serving the Preliminary Notice. This written record will be turned over to the Owner at Project Close-out.
- 5.17.2 **Supervision.** The Design-Builder will manage, supervise and direct the Work using its best skill and judgment. The Design-Builder shall identify the ("Design- Builder's Manager") who will represent the Design-Builder with respect to the daily operations of the Project, or may delegate authority to another representative. For simplicity, where this Agreement refers to the Design-Builder, Owner may assume that the Design-Builder's Manager is the appropriate point of contact. The Design-Builder's Manager may not be changed without Owner's written consent. The Design-Builder will also provide a gualified superintendent and assistant superintendent at the Project Site to properly supervise all of Design-Builder's employees, Subcontractors and their agents and employees, and other persons performing construction work and to ensure that the construction work is carried out in strict accordance with the Contract Documents. Both the superintendent and assistant superintendent shall be approved by Owner in its sole discretion.

Neither the superintendent nor the assistant superintendent may have other project or business responsibilities or calls on his or her time other than this Project.

5.17.3 Discipline. The Design-Builder will enforce strict discipline and good order at all times among Design-Builder's employees, Design Professionals and Subcontractors and will not employ or contract with any unfit or unskilled person(s) or entities on this Project. The Design- Builder and its Design-Professionals and Subcontractors will comply with all Owner policies, standards, and procedures throughout the duration of this Project.

- **5.17.4 Construction Coordination.** Before starting each portion of the construction work, the Design-Builder will: (i) coordinate with Owner's Project Manager, or his designee, to coordinate any necessary shut- down of equipment, and related operations issues; (ii) review and compare the various Contract Documents relative to that portion of the construction work, as well as the information furnished by the Owner, Design Professionals and Subcontractors that may affect proper installation of the work; (iii) field measure existing conditions related to that portion of the work; and (iv) observe any conditions at the site directly affecting that portion of the work.
- **5.17.5 Field Measurements**. The Design-Builder will take field measurements to ensure proper matching and fitting of new construction with existing conditions at the Project Site.
- 5.17.6 **Submittals**. The Design-Builder and its Subcontractors shall provide timely submittal of all "Shop Drawings," "Product Data," "Samples" and similar submittals (collectively referred to as "Submittals") required by the Contract Documents, to the Architect of Record for review and approval. After approval by the Architect of Record, all Submittals will also be provided to the Owner for Owner's information. All Submittals will be submitted in a sequence that avoids delays in the Project Schedule. Design-Builder will not submit any Submittal that is merely a tracing or copy of any of the Construction Documents. Each Submittal will be prepared by the Design-Builder and/or its Subcontractors or suppliers and will be submitted according to the Project specifications. No construction work will be performed without approval by the Design- Builder, as required. Regardless of the Submittal process, the Design- Builder remains responsible to the Owner for proper design and construction in compliance with all requirements set forth in this Agreement. Refer to OMR for additional provisions on Submittals.
 - **5.17.6.1 Response Times to Design-Builder Submittals.** Unless otherwise provided in the Contract Documents, Owner shall respond to reviews, approvals, and data needs to Design-Builder within 14 calendar days, provided that the Design-Builder shall, at Owner's request, act reasonably in allowing Owner an extension of time to respond to review, approvals, and data based on a high volume of submittals or complexity of submittals under review at a given time.
 - **5.17.6.2 Design-Build Subcontractors.** All Submittals prepared by Design-Builder and its Subcontractors shall be reviewed and approved by the Design-Builder's Architect of Record who will remain responsible to the Design-Builder and Owner for the design.
- **5.17.7 Coordination of MEP.** Mechanical, electrical, plumbing, fire protection and fire and life safety work will be coordinated, as appropriate, to avoid obstructions, conflicts, keep openings and other passageways clear, overcome interference with structural, framing, and equipment conditions, and coordinate with other trades.
- **5.17.8** Layout and Protection. The Design-Builder is responsible for all layouts and will preserve and protect all line and grade benchmarks. Any additional surveying or layout caused as a result of Design-Builder or any of its

Subcontractor's failure to take the necessary precautions to protect the data will be performed at Design-Builder's own cost and expense.

- **5.17.9 Materials and Equipment.** All materials and equipment required under the Contract Documents will be new and of good quality. Once the Construction Documents are complete, no substitutions will be accepted on this Project unless: (i) the specified materials or equipment have been discontinued; or (ii) the Owner has approved the substitution through written Change Order. Materials and equipment will be furnished in ample quantities and procured in time to ensure uninterrupted progress of construction. All materials and equipment will be properly stored and protected as required by the Contract Documents and any loss or damage due to improper storage or protection will be borne by the Design- Builder.
 - **5.17.9.1 Long Lead Items.** The Design-Builder will collaborate with Owner to establish a program to expedite ordering and delivery of materials and equipment requiring long lead time.
 - **5.17.9.2 Shipment and Deliveries.** Prior to shipment, delivery and installation of materials and equipment, the Design-Builder will verify the stage of completion of the Project with Owner to determine the availability of facilities for access, delivery, transportation and storage, and to correlate these observations with the requirements of the Contract Documents. All shipments and deliveries will be scheduled and coordinated in accordance with the most current approved site logistics plan and the most current approved Project Schedule.
 - **5.17.9.3 Storage of Materials and Equipment.** Storage of equipment and materials will be coordinated through the Design-Builder and the Owner. Design-Builder will maintain, or cause its Subcontractor's to maintain, all storage areas and will keep storage areas clean, safe, and secure. Storage areas shall also provide for proper protection of all stored materials and equipment from all forms of corrosion. Design-Builder must request and receive Owner's approval, granted at Owner's sole discretion, for offsite storage. Any materials or equipment stored offsite will be insured or stored in a bonded warehouse. The risk of loss will remain on the Design-Builder for all materials and equipment stored off-site.
 - **5.17.9.4 Risk of Loss.** All construction work stored at the Project Site, or work related to the preparation or delivery of materials or equipment to the Project Site, will remain at the risk of the Design-Builder or appropriate insurance carrier until Final Completion of the Project.
 - **5.17.9.5 Maintenance.** The Design-Builder will provide all maintenance for systems and equipment at its own costs and expense until Substantial Completion.
- **5.17.10 Correction of Work.** At any time prior to Final Completion, Owner may require Design-Builder to correct work that does not comply with the Contract Documents. Design-Builder must correct such defective work immediately

(unless otherwise approved by Owner), at its sole cost and expense, and in a manner that does not delay the completion of the Project.

5.17.11 Covering and Uncovering Work. Design-Builder must provide notice to Owner as to the schedule for covering Work so that the Owner has adequate time to observe Work to be covered. Owner may require any Work to be uncovered, whether or not prior information was provided as to the schedule for covering. Should Work so uncovered prove to be in non-compliance with the Contract Documents, the cost of uncovering, correction of the Work and re-covering shall be borne by the Design-Builder and the Owner is not be liable for any schedule recovery costs Design-Builder may incur. If Design-Builder provided adequate notice of covering and the work is compliant with the Contract Documents, Design-Builder is entitled to a change order for any extra cost caused Design-Builder, including any cost of schedule recovery. Design-Builder may comply with the notice requirements of this section as part of the project schedules described in Section 8.2 of the Design-Build Agreement.

6. SUBCONTRACTORS

- 6.1 **Procurement.** Those portions of the Work that the Design-Builder will not selfperform, or that will not be performed by Design Professionals or Subcontractors named in Design-Builder's proposal, will be performed by Subcontractors added during the course of the Work in accordance with the bidding process in Public Contract Code Sections 4100, et seq. All subcontracted work shall be performed under written subcontracts or purchase orders. The Design-Builder must furnish to the Owner in writing the names of the persons or entities the Design-Builder proposes to engage as subcontractors at least ten (10) days before said entity shall start any Work as a subcontractor. The Design-Builder may not contract with any subcontractor to whom the Owner has made reasonable and timely objection.
- 6.2 Written Agreements. All subcontracts will be in writing and will bind the Subcontractor to the Design-Builder by the terms of the Contract Documents, and Subcontractor will assume toward the Design-Builder all the obligations and responsibilities that the Design- Builder assumes toward the Owner. Each subcontract agreement will preserve and protect the rights of the Owner and Design-Builder under the Contract Documents with respect to the portion of the Work to be performed by the Subcontractor so that subcontracting the Work does not prejudice the Owner's rights. Where appropriate, the Design-Builder will require Subcontractors to enter into similar agreements with its tier- subcontractors. The Supplemental Conditions to this Agreement and all necessary Exhibits to this Agreement will be a part of each Subcontract Agreement.
- **6.3** Licensing Requirements. All Subcontractors will be properly licensed for their respective portion of the Work.
- **6.4 Standard of Care.** All Subcontractors will warrant that they possess the design and/or construction licenses and expertise required for this Project and will use the same degree of care and skill customarily used by California state licensed professionals and contractors performing similar services for residential facilities construction in the state of California during the same time frame.

- **6.5 Responsibility.** Design-Builder assumes responsibility to Owner for the proper performance of the Work of Subcontractors and any acts and omissions in connection with such performance. Nothing in the Contract Documents is intended or deemed to create any legal or contractual relationship between Owner and any Subcontractor or Sub-Subcontractor, including but not limited to any third-party beneficiary rights. Design-Builder shall coordinate the activities of all Subcontractors.
- **6.6 Conflicting Terms.** All conflicts arising out of any subcontract agreement will be resolved in accordance with the order of precedence set forth in Section 3.4, and this Agreement will take precedence over any terms and provisions in a subcontract.
- **6.7 Assignment.** Each subcontract agreement will include an assignment provision. The assignment provision will allow for assignment of subcontracts to the Owner upon termination of the Design-Builder for cause or convenience provided: (i) Owner accepts assignment by written notification to the Subcontractor and Design-Builder; and (ii) assignment is subject to the rights of the surety, if any, obligated under a bond or bonds relating to this Agreement. The Design-Builder will not be responsible for acts and omissions of the Subcontractors that occur after the effective date of assignment.
- **6.8 Claims and Dispute Resolution.** The Subcontractor will be bound to the same claims and dispute resolution procedures as set forth in Section 14.12 of the Supplemental Conditions.
- **6.9 Insurance.** The Design-Builder may, at its discretion, require its Subcontractors, through written subcontract, to carry appropriate insurance and bonding. Design-Builder's insurance must satisfy all requirements set forth in **Paragraph 12** regardless of any subcontractor coverage.
- **6.10 Indemnity.** The Design-Builder will cause its Subcontractors, through written subcontract, to include the indemnification provisions set forth in Section 11 and to indemnify and defend the Owner and its board of trustees, the Owner, and Design-Builder from all claims, damages and liability pursuant to the provisions in Section 11, except to the extent that such subcontractors cannot legally be required to indemnify (e.g., with respect to design-related claims).
 - **6.10.1 Third Party Beneficiary.** The Owner will be an express third party beneficiary to all design-build subcontracts.
 - **6.10.2 Subcontracts.** Subcontracts may be awarded on a lump sum or best value basis.
 - **6.10.3 Contract Time.** Subcontractors will be tied to similar provisions governing Contract Time under Section 8.

7. COMPENSATION

7.1 **Preconstruction Stage Compensation**. During the Preconstruction Stage of the Project, the Design-Builder will complete all Preconstruction Stage Services summarized in Section 5.2 of this Agreement and elaborated in more detail in the Scope of Work in **Exhibit 3.** Compensation for all Preconstruction Stage Services,

including labor, materials, overhead, and profit of Design-Builder and all of its Design Professionals, and design-assist Subcontractors, will be on the basis of a lump-sum amount of Ninety-nine Thousand One Hundred Five Dollars (\$99,105) as full compensation to the Design-Builder for the Work called for in Step One ("Preconstruction Stage Price"). Payments will be made based on monthly invoices, with monthly invoices/payments based on the percentage complete of the scope of work for the Preconstruction Stage services.

- **7.2 Construction Stage Compensation**. During the Construction Stage of the Project, the Design-Builder will perform all Construction Stage Services, as summarized in Section 5.4 of this Agreement. Compensation for the Construction Stage Services will be on the basis of a lump sum of Eight Hundred Ninety-one Thousand Nine Hundred Forty-nine Dollars (\$891,949) which will cover all labor, equipment, materials, profit, overhead, taxes and any other expenses to be incurred by the Design-Builder ("Construction Stage Price"). Design-Builder will be paid pursuant to monthly invoices based on a Schedule of Values and percentage of completion of the Work. The Preconstruction Stage Compensation and the Construction Stage Compensation shall, in the aggregate, constitute the Contract Price.
- **7.3 Construction Contingency.** The Contract Price includes a Construction Contingency in the amount of Fifty Thousand Dollars (\$50,000). The Construction Contingency is Owner controlled. Use of the Construction Contingency requires Owner's prior approval. All unspent funds in the Contingency at Final Completion shall accrue to Owner. The Construction Contingency is available for Design-Builder to cover cost of the Work unanticipated by Design-Builder on the effective date of the Design-Build Agreement, such as unanticipated field conditions or differing site condition, resequencing the Work for the good of the Project, acceleration in the Schedule for improvement in the overall Contract Time, and Owner requested changes.
- **7.4 Contract Price.** The Contract Price is the sum of the Preconstruction Stage Price, Construction Stage Price, and Constriction Contingency, and shall represent the sum total of all compensation due to the Design-Builder for all design and construction services under the Agreement. The Contract Price for this Agreement is the lump sum of One Million Forty-one Thousand Fifty-five Dollars (\$1,041,055).
- **7.5 Design-Builder's Fee.** The Design-Builder's Fee is included in the Contract Price. However, the Design-Builder's Fee of 15% can be applied to the direct cost of the construction in the event of contingency work and approved change orders. Design-Builder will not be entitled to Design-Builder's Fee for work necessitated by its own substandard workmanship, errors or omissions.
- 7.6 Design-Builder Allowances. Design-Builder may propose additional allowance, including appropriate allowances for defined items of Work that cannot be appropriately quantified and estimated at the time the Design-Build Agreement is executed, only with specific approval of Owner. Each such item of Work will be covered in a separate line item and have a clear description of what is covered by such allowance. Allowance items will be converted to, and included as, typical contract work by Change Order once conditions exist that allow them to be properly quantified and priced. Allowance items that cannot reasonably be quantified and estimated before the allowance work begins will be reconciled based on the actual cost of the allowance Work. If the actual cost of the Work for any item of Work covered by an

allowance will be greater than the amount of the allowance, Design-Builder will so notify Owner and if Owner authorizes the allowance Work in a Change Order, the Construction Stage Price will be increased by such difference with an additional agreed upon amount for overhead and profit. If the cost of any item to which such an allowance applies is less than the amount of the allowance, Owner may issue a Change Order decreasing the Construction Stage Price by the sum of the amount of such difference and the mark-up for overhead and profit on the difference.

8. CONTRACT TIME

- 8.1 **Contract Time.** The Design-Builder must achieve Final Completion of the Work using best practical safe speed to achieve Final Completion as soon as reasonably possible. The Contract Time is the time allotted for the Design- Builder to achieve Substantial Completion and Final Completion of the Work. Completion of the Preconstruction Stage must be achieved within (289) calendar days from issuance of the Notice to Proceed for the Preconstruction Services. Substantial Completion must be achieved within (146) calendar days from issuance of the Notice to Proceed for the Construction Stage Services. Final Completion must be achieved within (5) calendar days of achieving Substantial Completion. The Design-Builder must also achieve all specific milestone completion dates as set forth in the Project Baseline Schedule in Exhibit 6A.
- 8.2 Monthly Project Schedules. The Design-Builder will create monthly updates of the initial Project Baseline Schedule, referred to as "Monthly Project Schedules," incorporating activities and schedule updates of the Design Professionals and Subcontractors on the Project as necessary to reflect the status of design and construction and projected milestone dates for Substantial Completion and Final Completion. The Design-Builder will provide for Owner's approval information in an agreed upon format, and as requested by the Owner, for the scheduling of times and sequences of operations required for its Work in coordination with the work of Owner's employees and separate contractors, if any.
 - 8.2.1 3 Week Look-Ahead Schedules. The Owner will require the Design-Builder, with the assistance of its Design Professionals and Subcontractors. to create 3 Week Look-Ahead Schedules for the performance of upcoming Work and document all Work performed during the prior 3-week period. The Design-Builder will require its Subcontractors and Design Professionals to continuously monitor the Monthly Project Schedule and 3 Week Look- Ahead schedules to understand the timing, phasing and sequencing of operations of their respective work with other Work being performed at the Project. The 3 Week Look-Ahead Schedules are to be used as a working tool to evaluate any schedule slippages and collaborate on methods for labor efficiency. Work flow will be scheduled based on providing information, material and resources as required by the user of the information, material or resources, optimizing the flow of Work and reducing bottlenecks and activity that will not advance the Project Schedule. The Design-Builder will provide Owner with copies of the 3 Week Look- Ahead Schedules and will meet with Owner to review and coordinate with any work being performed by Owner's separate contractors.

Design-Builder shall request input from Owner on 3 Week Look- Ahead Schedules, as necessary, no less than one week before the 3 Week Look-Ahead Schedule submission date.

- **8.3 Prosecution of the Work.** The Design-Builder will commence the Work within 10 days of receipt of a Notice to Proceed by the Owner and will diligently prosecute and complete its Work pursuant to the most approved current Monthly Project Schedule.
 - 8.3.1 Schedule Slippage. The Design-Builder will notify Owner and the Owner within 72 hours of any slippage in the Monthly Project Schedule as a result of its Work and must submit a detailed recovery plan for evaluation and approval by Owner. All costs associated with the recovery, which shall provide for completion of the Project within the Contract Time, will be the responsibility of the Design-Builder unless the Design-Builder is entitled to an extension of time under Section 8.4
 - 8.3.2 Acceleration. The Owner may direct the Design-Builder and its Subcontractors and Design Professionals to work overtime in order to accelerate the Project schedule. If the Design-Builder and its Subcontractors and Design Professionals are not in default under any of the terms or provisions of this Agreement, their respective subcontracts and/or agreements, or any of the other Contract Documents, the Owner will pay the Design- Builder, its Subcontractors and Design Professionals for actual additional wages and/or billable rates paid, if any. All additional wages and billable rates paid will be subject to audit.
- 8.4 Permitted Delays. If the Design-Builder is delayed, obstructed, hindered or interfered with in the commencement, prosecution or completion of the Work by:
 (i) any negligent act or omission of the Owner, or Owner's separate contractors;
 (ii) "Owner Elected Changes;" (iii) delay caused by a "Force Majeure Event;"
 (iv) "Unforeseen and Differing Site Conditions;" and/or (v) "Owner's Suspension of the Work," such that the critical path of the most current, approved Project Baseline Schedule is impacted extending the Final Completion Date, the Substantial Completion Date, or any specific milestone completion dates, then the Design-Builder will be entitled to an extension for the same period of time that the Design-Builder was delayed provided that the delay, obstruction, interference or hindrance was not caused, in whole or in part by any fault, neglect, act or omission of the Design-Builder, its employees, Design Professionals, Subcontractors or suppliers.

Notwithstanding the above, the Design-Builder will not be entitled to an extension of time unless the Design-Builder provides the Owner with notice in writing of potential delay, obstruction, hindrance or interference within 72 hours of the discovery of the potential delay. Design-Builder shall follow up with all practical speed, but not later than 7 days after the initial notice, to summarize the cause or causes of the delay, and demonstrates that it could not have anticipated or avoided the delay, obstruction, hindrance or interference and has used all available means to minimize the consequences of the delay. The Design-Builder may also be entitled to an adjustment in the Contract Price based on demonstration that the delay and resulting adverse material effect in the cost of completing the Work, after implementation of all reasonable mitigation, materially adversely affected Design-Builder's cost of completing the Work.

8.5 Liquidated Damages. The Owner and Design-Builder acknowledge and agree that if Design-Builder fails to complete Work within the time set forth in section 8.1, the Owner will suffer damages that are both extremely difficult and impracticable to ascertain. Therefore, Owner and Design-Builder agree that, liquidated damages shall be enforced on failure to achieve Substantial Completion in the amount of \$3,000 per day, and failure to achieve Final Completion in the amount of \$1,000 per day.

Payment of liquidated damages represents a reasonable estimate of fair compensation for the losses that reasonably may be anticipated as a result of Design-Builder's delays in completing the Work. Owner and Design- Builder acknowledge and agree that these liquidated damages provision will be Owner's sole remedy for delay damages caused by Design- Builder's failure to achieve Substantial Completion or Final Completion, within the time set forth in Section 8.1, and/or any of the specific milestone completion dates. Nothing contained in this Section 8.5 shall preclude Owner from recovery for actual damages unrelated to Design-Builder's delays, including, but not limited to, claims for actual losses incurred due to breach of contract, negligence, defective work, injury to persons or property or third-party claims.

9. CHANGES

- **9.1 Change Orders.** A Change Order is a mutually agreed upon written order adjusting the Design-Builder's Scope of Work, Contract Price, Contract Time or any combination. A Change Order may come through an Owner Elected Change, or Design-Builder's request. All changes in the Work will only be authorized by an Owner Elected Change, or Owner executed Change Order and performed under the applicable conditions of the Contract Documents. A Change Order signed by the Design-Builder and Owner indicates an agreement to any adjustment in the Contract Time, and/or Contract Price, which includes all Costs of Work plus Fee, and that the adjustments in the Change Order fully and completely resolves any claim by Design-Builder for additional compensation or time arising from or related to the subject of the Change Order. Change Orders for additional Work that was not considered as part of the Contract Price are limited to the following circumstances and, therefore may impact the Contract Price, and may or may not impact Contract Time:
 - 9.1.1 Owner Elected Changes
 - 9.1.2 Force Majeure Events
 - 9.1.3 Unforeseen and Differing Site Conditions
 - 9.1.4 Owner's Suspension of the Work as defined in Section 16.2
 - 9.1.5 Changes in applicable law
- **9.2 Owner Elected Change.** The Owner will initiate a Change Order by providing the Design-Builder with a written summary of the Owner Elected Change. Within 10 business days of receipt of an Owner Elected Change, or such other mutually-agreed upon period for more complex or extensive Owner Elected Changes, the Design-Builder must submit a complete cost proposal for the revised scope to the Owner, as well as any proposed change in Contract Time under Section 8. The Owner will review and evaluate the Design-Builder's cost proposal and any proposed change in Contract

Time, before presenting the Design-Builder with a proposed Change Order at either its regular weekly meeting or a special meeting.

- **9.3 Design-Builder Initiated Changes.** The Design-Builder must provide the Owner written notice of a proposed change within 5 business days of discovery of the facts or circumstances giving rise to the proposed change order. The Owner will meet and discuss the proposed change either at its regular weekly meeting or at a special meeting.
- **9.4 Submission.** All claims for additional compensation or extensions in Contract Time will be presented in writing to the Owner for review. The Owner will either discuss the proposed change at its regular weekly meeting or will call a special meeting to meet and review the proposed change. At the conclusion of the meeting an Owner Elected Change may be issued. Consistent with Owner's internal procedures, a change order request may require approval from Owner's Board of Trustees. All Change Orders must be approved by the Owner before the expense is incurred. Additive Change Orders will affect Contract Price and may affect Contract Time, subject to Section 8. Deductive Change Orders will affect Contract Price and may affect Contract Time.
- **9.5 Continued Performance.** No Work will be allowed to lag pending the adjustment through Change Order, but will be promptly executed as directed through Owner Elected Change, even if a dispute arises. Disputes will be resolved in accordance with Section 14.12. Failure of the Design-Builder to provide the Owner with notice of its disputed claim and to submit the written claim within 10 business days of completion of the Work in dispute constitutes an agreement on the part of the Design-Builder that it will not be paid for its Work. No claim will be considered after the Work in question has been performed unless a written Change Order has been executed or timely written notice of claim has been made by the Design-Builder. The Design-Builder will not be entitled to claim or to bring suit for damages, whether for loss of profits or otherwise, on account of an omission of any item or portion of Work covered by the executed Change Order.
- **9.6 Omitted Work.** If the Design-Builder omits any portion of the Work that is included in the Contract Documents, the Owner will have the right to withhold from payments due or to become due to the Design-Builder in an amount which, in the Owner's opinion, is equal to the value of portion of the Work that was omitted until the omitted Work is performed.
- **9.7 Contract Price Reduction.** The Owner may also reduce the Contract Price to reflect back-charges or payments withheld pursuant to the Contract Documents upon written notice, and 48 hours opportunity to cure.
- **9.8 Contract Time Impacts and Extended Costs.** The Design-Builder will not reserve a right to assert impact costs, extended job site costs, extended overhead, constructive acceleration and/or actual acceleration beyond what is allowable under Section 8 and claimed in a proposed change order under Section 9.3. No claims will be allowed for impact, extended overhead costs, constructive acceleration and/or actual acceleration due to a multiplicity of changes and/or clarifications. Nothing contained in this Section will be construed as restricting the rights and remedies of Design-Builder in violation of Civil Code section 2782 or Public Contract Code section 7102. If this provision is determined to conflict with Public Contract Code section 7102 or Civil Code section

2782, this provision will be reformed to provide the greatest protection to the Owner under the law.

9.9 Surety. All changes, additions or omissions in the Work ordered through an Owner Elected Change, or Change Order are part of the Work and will be performed and furnished in strict accordance with all of the terms and provisions of the executed Change Order and the other Contract Documents. The Design- Builder will keep its surety informed of all modifications to this Agreement. The obligations of Design-Builder's surety are not to be reduced, waived or adversely affected by the issuance of Change Orders even if the Design-Builder fails to inform the surety of the Change Order(s) and the Owner will not be required to obtain consent of the surety to the Design-Builder or any of itsSubcontractors.

10. PAYMENT

- 10.1 Progress Payments. In accordance with Public Contract Code section 20104.50, the Owner will make monthly progress payments on all undisputed Work performed within 30 calendar days of receipt of a monthly invoice and a monthly progress report that were properly submitted pursuant to the procedures set forth in this Section and as further established by the Owner. Each invoice will be submitted on the forms provided by the Owner, will include an itemized list of the work performed, be based on the percentage of the Work completed, and provide a level of detail to allow Owner to make a fair and reasonable estimate of the value of Work completed. The invoice must be certified by the Design-Builder and made out to the Owner. Before making payment, the Owner will review the invoice for accuracy of the Work completed to date. No such payment shall be required to be made when, in the judgment of the Owner, the request for payment is in excess of the percentage of Work completed. Unless otherwise provided in the Contract Documents, payments shall be made on account of materials and equipment delivered and suitably stored at the site for subsequent incorporation into the Work. If approved in advance by the Owner, payment may similarly be made for materials and equipment stored off of the site at a location agreed upon in writing, provided that such off-site location shall be a bonded warehouse or secured by other means acceptable to the Owner.
 - 10.1.1 **Schedule of Values**. Design-Builder must submit a draft schedule of values to Owner for approval prior to the first payment and within ten days after Notice to Proceed. As approved by Owner, the schedule of values shall be used for preparing future estimates for partial payments to the Design-Builder, and shall list the major items of Work, including materials and services, with a cost fairly apportioned to each item so that the total of the prices for all items equal the lump sum price. The schedule of values shall be by area, structure, or other logical division of work. The insurance, bond, Elevator, Car Lift, Doors & Frames, Finish Hardware, Cabinets, Appliances, Windows, Scaffolding and overhead costs will be carried on separate line items and the Design- Builder shall be entitled to bill for reimbursement for costs related to Elevators and Car Lifts as such costs are incurred, subject to documentation of such costs, prior to the Elevators and Car Lifts being installed in the premises. The schedule of values shall not be considered in determining payment or credit for additional or deleted work. The final Schedule of Values will be added to the Design-Build Agreement as Exhibit 4B.

- **10.1.2 Monthly Progress Reports.** Each monthly invoice must include a report providing an overall status of the Project's progress, and any concerns or impacts.
- 10.1.3 **Evaluation of Invoice.** The Owner will review the invoice based on the approved schedule of values, monthly progress report, on-site observations and evaluation of the Work, and on the data and documentation substantiating the invoice. Upon request, Design-Builder must substantiate the cost for any or all items and provide additional level of detail, including quantities of work. Based on that review. Owner will pay all undisputed items. An approval of an invoice is subject to results of subsequent tests and inspections, to correction of minor deviations from the Contract Documents prior to completion of the Work and to specific qualifications expressed by the Owner. Owner is entitled to rely on the accuracy and completeness of the information furnished by the Design-Builder and approval will not be deemed to represent that a detailed examination, audit, or arithmetic verification of the documentation submitted with the Design-Builder's invoice has been made or that exhaustive or continuous on-site inspections have been made to verify that the Work is in accordance with the Contract Documents. A payment by Owner does not represent that Owner has ascertained how or for what purpose the Design-Builder has used money previously paid.
- **10.1.4 Retention.** The Owner will withhold 5% of each progress payment during the Project. Retention will be withheld until the Project achieves Final Completion unless the Owner, in its sole discretion, agrees to release the Design-Builder's retention earlier and provided that the Work has been accepted by the Owner and other necessary agencies with jurisdiction over the Project.
 - **10.1.4.1 Substitution of Securities.** To the extent required by law, Owner will consider and approve reasonable and appropriate requests under Public Contract Code section 22300 for substitution of securities or establishment of an escrow account for retention. Nothing contained in this Section will prevent Owner from withholding payment when grounds exist for doing so under the ContractDocuments.
- **10.1.5 Change Orders.** Applications for payment may include requests for payment on account of changes in the Work that have been properly authorized.
- **10.1.6 Stored Materials and Equipment.** Stored materials and equipment may be included in the invoice provided the materials and equipment are properly stored in accordance with Section 5.17.9.3 and a complete invoice accompanies the invoice. Owner will not pay for materials or equipment storage.
- **10.1.7 Stop Payment Notices and Claims.** Upon submission of an invoice, the Design-Builder warrants that all Work included in the invoice has been performed in accordance with the Contract Documents and to the best of the Design-Builder's knowledge, information and belief, title to all Work covered by the invoice will pass to the Owner free and clear of all stop payment notices, claims, security interests or encumbrances. Design-Builder will provide executed conditional waivers and release of claims for all amounts

included in the invoice. Waivers must comply with the requirements of California Civil Code section 8132.

- **10.2 Owner Payment to Design Professionals, Subcontractors and Suppliers.** The Owner will not have an obligation to pay a Design Professional or Subcontractor for work performed unless required by law. However, if the Owner is not in default of payment provisions and receives a stop payment notice or has reason to believe that the Design-Builder is not paying its Design Professionals or Subcontractors and suppliers, the Owner may make payment of sums due to Design-Builder through joint check or pay Design Professionals and Subcontractors and suppliers directly and withhold those payments from Design-Builder. The Owner shall not exercise any rights granted under this Section prior to issuing a notice to the Design-Builder and granting the Design-Builder a reasonable opportunity to cure.
- **10.3 Final Payment.** Upon Final Completion of the Work, the Design-Builder will submit a final payment application. All prior progress estimates will be subject to correction in the final invoice. If items remain to be completed at that time, then the Design-Builder in conjunction with Owner will revise the Final Punch List and will include 150% of the estimated cost to complete each remaining item. The Owner may withhold from the final payment 150% of the estimated cost to complete the Work. The amount retained by the Owner for Final Punch List items will be released to the Design-Builder as each item is completed. Upon Final Completion of the Project, and submission of Owner's Release of Claims form, by Design- Builder, final payment of all remaining retention, if unencumbered, will be paid on all undisputed amounts no later than 30 calendar days after either Final Completion of the entire Project or recordation of a Notice of Completion (whichever is later), and in no event later than the time prescribed under Section 7107 of the Public ContractCode.
 - **10.3.1 Contract Price Reduction.** Upon Final Completion of the Project, the Owner after written notice to Design-Builder may reduce the Contract Price to reflect costs charged to the Design-Builder, back-charges or payments withheld pursuant to the ContractDocuments.
 - **10.3.2 Evidence.** Before issuance of final payment, Owner may request satisfactory evidence that: (i) all payrolls, materials bills and other indebtedness connected with the Work have been paid or otherwise satisfied; (ii) insurance required by the Contract Documents will remain in force after final payment and will not be canceled or allowed to expire until at least 30 calendar days prior written notice has been given to the Owner; (iii) the Design- Builder knows of no substantial reason that the insurance will not be renewable to cover the period required by the Contract Documents, (iv) surety, if any, has consented to final payment, (v) Owner has received all close-out documents required by the Contract Documents; and (vi), other data establishing payment or satisfaction of obligations, such as releases and waivers of stop payment notices, claims, security interests or encumbrances arising out of this Agreement have been received.
 - **10.3.3 Payment Not Acceptance of Work.** Approval of an invoice (final or otherwise) or partial or entire use or occupancy of the Project by the Owner will not be used as conclusive evidence that the Work was properly

performed or constitute acceptance of Work that is not in accordance with the Contract Documents.

- **10.4 Payments Withheld.** In addition to the 5% retention, the Owner may withhold payments due to the Design-Builder that may be necessary to cover: (i) stop payment notice claims; (ii) defective Work not remedied; (iii) failure of Design-Builder to make proper payments to its Design Professionals, Subcontractors or suppliers; (iv) damage to Design Professionals, Subcontractors or third party caused by Design-Builder; (v) amounts due to the Owner for claims against Design-Builder; (vi) failure to provide Owner with timely schedule updates under Section 8.2; (vii) disputed amounts in the invoice; and (viii) legally permitted penalties.
- **10.5 Waiver of Claims.** Acceptance of final payment by the Design-Builder constitutes a waiver of claims by Design-Builder and its Design Professionals, Subcontractors and suppliers except for those previously made in writing and identified by the Design-Builder as unsettled at the time of final invoice. The Design-Builder must acknowledge this waiver on a form provided by the Owner prior to Design-Builder's receipt of the final payment.

11. INDEMNIFICATION

11.1 Indemnification. Design-Builder shall defend, indemnify, and hold harmless San Mateo County, and its trustees, officers, employees and agents, and their successors and assigns (collectively referred to as "Indemnitees") from and against all third party claims, demands, liability, suits, actions, costs or expenses (including reasonable attorneys' fees) for any and all loss or damage, including, but not limited to, personal injury or property damage, arising out of or resulting from allegations of:

a. The failure or alleged failure by any Design-Build Team member to comply with any applicable law, order, citation, rule, regulation, standard, ordinance or statute, including rules and regulations imposed by Cal-OSHA and caused by the act or omission of the Design-Builder;

b. The negligent act, omission, misconduct, or fault, or the alleged negligent act, omission, misconduct or fault of any Design-Build Team member;

c. Any and all claims by any governmental or taxing authority claiming unpaid taxes based on gross receipts, purchases or sales, the use of any property or income of any Design-Build Team member with respect to any payment for the Work made to or earned by the Design-Build Team member under the Contract Documents;

d. Any and all stop payment notices and/or liens filed in connection with the Work, including all expenses and attorneys' fees incurred by the Owner in discharging any stop payment notice or lien, provided that the Owner is not in default on payments owing to the Design-Builder with respect to suchWork;

e. Failure of the Design-Builder to comply with the Insurance provisions set forth in Paragraph 12;

- f. Any release of hazardous materials:
 - 1. Brought onto the Site by any Design-Build Team member; or

2. Where the removal or handling involved negligence, willful misconduct, or breach of Contract by any Design-Build Team member; and

Nothing in this section shall require the Design-Builder to defend, indemnify or hold harmless the Indemnitees for the Indemnitees' sole negligence, willful misconduct, or active negligence.

11.2 Indemnification for Infringement of Intellectual PropertyRights.

The Design-Builder agrees to fully defend, indemnify, and hold harmless the Indemnitees against any demand, claim, cause of action, suit, proceeding, or judgment that design, service, method, or product called for and provided by the Design-Builder or any Design-Build Team member (herein called "deliverables") that infringes or allegedly infringes any patent, copyright, trademark, service mark, trade dress, utility model, industrial design, mask work, trade secret, or other proprietary right of a third party (collectively "Intellectual Property Right").

The Design-Builder shall pay any and all costs of such defense and settlement (including interest, fines, penalties, costs of investigation, costs of appeals, and attorney 's fees), and will pay any and all costs and damages finally awarded against any of the Indemnitees. The Design- Builder shall have the exclusive right to conduct its legal defense.

In the event that any deliverable furnished hereunder, or called for in any design or services provided under this Agreement, is in any suit, proceeding, or judgment held to constitute an infringement on any third party's Intellectual Property Right, and its use is enjoined, the Design- Builder shall, at its own expense accomplish the following:

a. Procure the fully paid-up, irrevocable, and perpetual right for the Owner to continue using the deliverable;

b. Modify the deliverable; or

c. Provide for the replacement of the deliverable with an alternative product that is functionally equivalent to the deliverable.

If the Design-Builder is unable to provide the Owner with one of the forms of relief described above, the Design-Builder shall also reimburse to the Owner the total paid by the Owner for the deliverable that is held to constitute an infringement.

11.3 Indemnification for Design Defects. To the fullest extent permitted by law, the Design-Builder shall fully defend (with counsel acceptable to the Owner), indemnify, and hold harmless Indemnitees from any and all claims, demands, causes of action, damages, costs, expenses (including legal, expert witness, and consulting fees and costs), losses, or liabilities of whatsoever nature that arise out of, pertain to, or relate to the negligence, recklessness or willful misconduct of the Design-Builder, its employees, any of the Design-Builder's Design Professionals or Subcontractors of any tier, or anyone for whom Design-Builder or any of its Design Professionals or Subcontracts may be liable, in relation to any of their design services, including but not limited to errors, omissions, inconsistencies, inaccuracies, deficiencies, or other defects whether or not contained in the Construction Documents furnished by the Design-Builder, and whether or not such errors, omissions, inconsistencies, inaccuracies, inconsistencies, inconsistencis, inconsistencis, inconsistencies, inconsistencies, inconsist

inaccuracies, deficiencies, or other defects were also included in the Contract Documents provided by the Owner. The Design-Builder agrees that, because the Bridging Contract Documents are preliminary and conceptual in nature and are subject to review and modification by the Design-Builder, such documents shall not be deemed a "design furnished" by the Owner or any of the other Indemnitees, as the term "design furnished" is used in Civil Code Section 2782, and that this clause is governed by Civil Code Section 2782.8. In addition, Design-Builder shall defend the Owner, or pay for the costs of such defense, to the extent of Design-Builder's proportionate percentage of fault for the underlying claim. In addition, Design-Builder shall defend the Owner, or pay for the costs of such defense, to the extent of Design-Builder Builder's proportionate percentage of fault for the underlying claim. In addition, Design-Builder Builder's proportionate percentage of fault for the underlying claim.

- 11.4 **Exception.** The indemnification provisions in this Section 11 will extend to claims occurring after this Agreement is terminated as well as while it is in force. However, Design- Builder will not be obligated to indemnify or provide a defense to the Indemnitees from claims arising from the active negligence or willful misconduct of Indemnitees. If any of the Indemnitees are actively negligent, the Design-Builder will continue to indemnify and provide a defense to Indemnitees but only to the extent and in proportion to the degree that the Indemnitees were not actively negligent. Nothing contained in Section 11 will be construed to impose any obligation in conflict with the provisions of Civil Code section 2782 and/or Insurance Code section 11580.04. In the event of a conflict, the provision conflicting with Civil Code section 2782 and/or Insurance Code section 11580.04 will be modified to limit Design-Builder's obligations to the greatest extent permitted by law. The section does not apply to the duty to defend claims arising from design defects, which obligation is addressed separately in Section 11.3. The section does not apply to the duty to defend claims arising from design defects, which obligation is addressed separately in Section 11.3.
- **11.5 Duty to Defend.** Except as otherwise provided herein, the Design-Builder will defend all claims defined in Section 11.1 at its own cost, expense and risk and pay and satisfy any judgment or decree that may be rendered against any Indemnitee arising out of a claim, or reimburse Indemnitee(s) for any and all legal expenses incurred by any of them in connection with the claim or in enforcing the indemnity granted in this section. The duty to defend will apply, and Design-Builder will be required to furnish a defense, regardless of whether the matter has been adjudicated. The Owner shall have the right to approve counsel defending it, which approval will not be unreasonably withheld. Following the resolution of any such dispute, the Owner shall reimburse Design-Builder for the costs incurred by Design-Builder for any Indemnitees' defense, to the extent of the Owner's proportionate responsibility or fault, as determined by court or arbitrator or as agreed by settlement or otherwise.

12. INSURANCE AND BONDS

12.3 Design-Builder's Insurance Requirements. The Owner and its trustees, officers, employees, agents, and volunteers will be a named additional insured under all of Design-Builder's insurance policies except errors and omissions policies and workers' compensation policies. Likewise, the Design-Builder will require all Subcontractors to name the Owner, and its trustees, officers, employees, agents, and volunteers as additional insured on all Subcontractor policies except errors and omissions policies and workers' compensation policies. Before commencement of the Work, the Design-Builder will provide certificates of insurance and endorsements per the following as

evidence of insurance and Owner's, and Owner's additional insured status under those policies.

12.4 DBE Provided Insurance

12.4.1 General DBE shall procure and maintain for the duration of this Contract at its sole cost and expense, insurance against claims which may arise from, or in connection with, the performance of the Work by, or on behalf of (whether directly or indirectly), the DBE.

Each insurance policy required by this Contract shall be endorsed to state that coverage shall not be suspended, voided, canceled, or reduced in coverage or in limits except after thirty (30) days' prior written notice has been given to the Owner, except that ten (10) days' prior written notice shall apply in the event that cancellation for non- payment of premium.

- **12.4.2 Commercial General Liability**: \$1,000,000 per occurrence for bodily injury, personal injury and property damage. If Commercial General Liability Insurance or other form with a general aggregate limit is used, either the general aggregate limit shall apply separately to this project/location or the general aggregate limit shall be twice the required occurrence limit; and
- **12.4.3** Workers' Compensation and Employers' Liability: Workers' Compensation limits as required by the California Labor Code and Employers Liability limits of \$1,000,000 per accident; and
- **12.4.4** Automobile Liability: \$1,000,000 combined single limit per accident for bodily injury and property damage; and
- **12.4.5 Builder's Risk/Course of Construction:** DBE shall procure and maintain in effect a Builders' Risk (course of construction) insurance for completed value of the Work. No deductible shall exceed \$100,000, per occurrence except for earthquakes, earth movement or flood. Builder's Risk Policies shall contain the following provisions:
 - **12.4.5.1** Owner and Subcontractors of every tier shall be named as an additional insured loss payee; and
 - **12.4.5.2** Coverage shall contain a mutual waiver of subrogation in favor of the Design Build Entity, Subcontractors at every tier, and the Owner, its officials, employees, agents, and only to the extent of onsite activity, design or engineering professionals.
- **12.4.6** Owner and Design Build Entity will share equally in payment of all deductibles from a covered event due to act of God events including earthquake, earth movement, and flood.
- **12.4.7 Contractor's Pollution Liability Insurance** on an occurrence basis, with limits of at least \$2,000,000 per occurrence and \$2,000,000 policy term aggregate for bodily injury, property damage, cleanup costs and claim expenses, arising at or emanating from the Project Site arising from all

operations performed on behalf of the Design Build Entity. Subcontractors will provide Pollution Liability coverage as required by their specific Subcontract.

Such insurance shall provide liability coverage for both sudden and gradual releases arising from the Work. CPL policy shall name Owner, Design-Build Entity and all Subcontractors of all tiers as insureds.

Contractor shall be responsible at its own expense for an obligation for each loss payable under this insurance that is attributable to the Design-Build Entity's acts, errors, or omissions, or the acts, errors, or omissions of any of its Subcontractors, or any other entity or person for whom Design-Build Entity may be responsible. The amount of the obligation shall be based on the amount of the initial Contract Price, as follows:

12.4.7.1 The portion of the obligation applying to the Design-Build Entity or Subcontractor shall be the responsibility of the Design Build Entity and shall remain uninsured. Design Build Entity shall promptly pay its charge pertaining to any loss. The Owner, in addition to its other remedies, may back charge Design-Build Entity for the obligation and deduct the back-charged amount from Design-Build Entity's next progress payment or final payment.

12.4.8 Professional Liability Errors and Omissions Insurance: \$1,000,000 per claim/\$2,000,000 aggregate limit.

12.4.9 Waivers

12.4.9.1 Owner and Design-Build Entity waive all rights against each other and any of their consultants, including Construction Manager, Bridging Architect and their consultants to the extent of their onsite exposure, separate contractors, if any, Subcontractors, Designers, agents and employees, each of the other, and any of their contractors, subcontractors, agents and employees, for damages caused by fire or other causes of loss to the extent covered by Builder's Risk insurance obtained pursuant to paragraph 1.2 above, or other property insurance applicable to the Work, except such rights as they have to proceeds of such insurance held by the Owner in good faith. Owner or Design-Build Entity, as appropriate, shall require of the separate contractors, if any, and the Subcontractors, Designers, agents and employees of any of them, by appropriate agreements, written where legally required for validity, similar waivers each in favor of other parties enumerated herein. The policies shall provide such waivers of subrogation by endorsement or otherwise. A waiver of subrogation shall be effective as to any individual or entity even if such individual or entity (a) would otherwise have a duty of indemnification, contractual or otherwise, (b) did not pay the insurance premium directly or indirectly, and (c) whether or not such individual or entity had an insurable interest in the property damaged. The only exceptions to this waiver of subrogation are for claims that may be covered by any Professional Liability insurance to the extent that insurance responds to any loss.

- **12.4.9.2** Owner waives subrogation rights under the Contractor's Pollution Liability Policy, to the greatest extent permitted by law, against all other project participants, including Design-Build Entity and Subcontractors of any tier.
- 12.5 Performance and Payment Bonds. Prior to commencement of the Construction Stage, the Design-Builder will furnish a payment bond and a performance bond to the Owner, each in the amount equal to 100% of the amount of the Construction Stage Price, covering all construction work and warranties, on the forms acceptable to the Owner. The payment and performance bonds will be provided prior to commencement of Work. The surety supplying the bonds must be an admitted surety insurer, as defined in Code of Civil Procedure Section 995.120, authorized to do business as a surety in the State of California and satisfactory to the Owner. Failure to furnish the required payment and performance bonds to the Owner constitutes a default under this Agreement and the Owner will have all of the rights and remedies provided under the Contract Documents and afforded by law including, but not limited to, forfeiture of the bidder's bid deposit or bond to the Owner and the Owner may award this contract to another responsive and responsible bidder, or may call for new bids. Full compensation for furnishing the payment and performance bonds are included in the Contract Price.
- **12.6 Payment of Subcontractors.** Without limiting the responsibilities of Design- Builder and its surety under the terms of this Agreement, the Design-Builder and its surety agree to promptly pay all lawful claims of Subcontractors, materialmen, laborers, persons, firms or corporations for labor or services performed or materials, supplies, machinery equipment, rentals, fuels, oils, tools, appliances, insurance and other items furnished, used, or consumed in connection with the prosecution of the construction work including Change Orders, and will indemnify and save harmless the Owner, and Owner from and against all liability loss, damage and expense, including interest, costs and attorneys' fees, which the Owner, and Owner and/or its surety may sustain by reason of Design-Builder's or its surety's failure to do so.

13. WARRANTY OF THE WORK

13.3 Design-Builder Warranty. The Design-Builder shall provide a two-year warranty on all furnished labor and materials, commencing on the date of Final Completion of all Work under the Agreement. Design-Builder shall perform all required corrective work, and shall be responsible for the cost of all labor, materials, equipment, transport, installation and re-testing required for the corrective work. Moreover, in the event that corrective work is required under the Design-Builder Warranty, a one-year warranty shall apply to the corrected work covering any discrepancies and defects in the corrected work that are discovered after the corrected work is accepted.

The Design-Builder Warranty shall warrant that:

a. The Work conforms to the requirements of the ContractDocuments;

b. All Design Services furnished under the Agreement conforms to all professional engineering principles generally accepted as standards of the industry in the State of California and complies with the standard of care of a reasonable professional that is performing the same or similar work, at the same time and locality and under the same or similar conditions;

The construction Work furnished under the Agreement is free from defects in workmanship, and was performed in a workmanlike manner and conforms to the standards of care and diligence normally practiced by recognized construction firms performing construction of a similar nature in the State of California, and conforms to the requirements of the Bridging Contract Documents and the100% Construction Documents, as these documents may have amended during the course of Work under the Agreement;

c. Materials and equipment furnished under the Contract Documents are of good quality and new;

- d. The Facilities are fit for the purposes intended in the Contract Documents;
- e. The Facilities shall be free of defects in design, material, and workmanship; and

f. The Facilities shall function up to the standards set forth in the Bridging Contract Documents and all other reliability standards established in the Contract Documents.

13.4 Subcontractor Warranties. The Design-Builder shall obtain one year warranties commencing on the date of Final Completion of all Work under the Agreement from all Subcontractors and Design-Build Team members providing design services, labor, equipment, materials, supplies and maintenance equipment; require all such warranties to be executed in writing for the benefit of the Owner and enforce all warranties for the benefit of the Owner, if so directed by the Owner. Warranties by subcontractors are in addition to, and do not replace or reduce, any other warranty obligations stated in the Contract Documents, including but not limited to the Design-Builder Warranty. All such Subcontractor warranties from Design-Build Team members shall run directly to and be enforceable by the Design-Builder and the Owner, and their respective successors and assigns.

The Design-Builder hereby assigns to the Owner all of the Design-Builder's rights and interest in all warranties that are received by the Design-Builder from any Subcontractor or Design-Build Team members unless necessary for enforcement. All such warranties shall survive Final Completion, acceptance, final payment, and termination of the Agreement if the stated warranty period extends beyond the Final Completion, acceptance, final payment, and termination of the Agreement if a payment, and termination of the Agreement.

- **13.5 Manufacturers' Warranties.** The Design-Builder shall obtain manufacturers' warranties for all equipment procured and installed on the Project and shall assign all such warranties to the Owner prior to Final Completion. Owner and Design-Builder shall agree upon acceptable warranty periods for each item of equipment prior to the procurement of the equipment of not less than one year from Final Completion. Manufacturers' warranties shall all commence on Final Completion.
- **13.6 Remedy.** The Design-Builder shall remedy, at its own expense, any failure to conform to the warranty requirements set forth in this Section 13 Warranty of the Work. If the Design-Builder fails to remedy any such failure within a reasonable time after receipt of notice (or immediately in the case of an emergency), the Owner shall have the right in its sole discretion to replace, remove, or otherwise remedy the failure at the Design-Builder's expense.
 - **13.6.1** Notification to Design-Builder. The Owner shall notify the Design- Builder, in writing, within a reasonable time after the discovery of any failure to

conform to the warranty requirements set forth in this Section 13, Warranty of the Work.

- **13.6.2** Warranties Do Not Limit the Design-Builder's Liability. The foregoing warranties are in addition to all rights and remedies available under the Agreement or applicable law, and shall not limit the Design- Builder's liability or responsibility imposed by the Agreement or applicable law with respect to the Work, including:
 - a. Liability for design defects;
 - b. Latent construction defects;
 - c. Strict liability;
 - d. Negligence; and
 - e. Fraud.
- **13.7 Assignment of Warranty.** Upon providing written notice to the Design-Builder, the Design-Builder's warranties, including all warranties from Subcontractors and Design-Build Team members that have been assigned to the Design-Builder, shall be immediately assignable by the Owner to any entity, in the Owner's sole discretion.

14. OWNERSHIP AND USE OF DOCUMENTS

- **Ownership of Construction Documents**. The drawings, specifications and other 14.3 documents prepared by or on behalf of the Design-Builder pursuant to this Agreement (including, without limitation, the Construction Documents), including all drafts, and the copyright thereto, shall at all times be and remain the property of the Owner, whether or not the Project for which they are made is commenced, so long as the Owner shall not be in default of its obligations under this Design- Build Agreement. Neither the Design-Builder nor any subcontractor or material or equipment supplier shall own or claim a copyright in such drawings, specifications and other similar or related documents, and Owner shall retain all common law, statutory, and other reserved rights with respect thereto. All copies of such documents shall be delivered by the Design-Builder to the Owner upon completion of the Work or upon the prior termination of this Agreement. Such drawings, specifications and other documents shall be used by the Design- Builder solely with respect to this Project and shall not be used by the Design- Builder or any subcontractor or material or equipment supplier on other projects or for additions to this Project outside the scope of the Work without the specific written consent of the Owner in each instance.
- **14.4 Ownership under Termination.** Should the Owner elect to terminate this agreement, the Design-Builder agrees that (1) for itself and its Architect, that the Owner will be the sole and absolute owner of the 100% Construction Documents, and shall have the right to use or to modify the 100% Construction Documents in any manner it wishes, including, without limitation, using the 100% Construction Documents in construction contracts with third parties; and (2) there shall be no limitation on the Owner by the Design-Builder should the Owner subsequently engage the Design-Builder's Architect of Record for services on this project or for other services.
- **14.5 Confidentiality.** The documents, materials and information prepared by or on behalf of, or furnished to the Design-Builder in connection with the Work, including, without

limitation, the RFP, the Contract Documents, the 100% Construction Documents and any other plans, specifications, drawings, shop drawings or details relating to the Project and the terms and provisions of this Agreement, shall be kept strictly confidential by the Design-Builder. The Design- Builder shall not disclose, furnish or make known or accessible to or use for the benefit of anyone, any such documents, materials or information or make available any reports, recommendations and/or conclusions which the Design- Builder may make for the Owner to any person, firm or corporation or use such documents or information in any manner whatsoever without obtaining the Owner's prior written approval in each instance unless such disclosure is required by law. The Design-Builder acknowledges that the Owner will incur significant damages in the event of a breach by the Design-Builder of its obligations under this Section 14.3. The provisions of Article 14 shall survive the expiration or prior termination of this Agreement.

- **14.6** Licensing. The Design-Builder, its Design Professionals and Design-Build Subcontractors are granted a limited, non-exclusive, license to use and reproduce applicable portions of the Design and Construction Documents and other documents prepared by the Design-Builder for use in the performance of the Design-Builder's Work under this Agreement. Additionally, the Owner grants the Design-Build Team members a non-exclusive, perpetual license for use, or display of the Project information solely for either educational or promotional purposes.
- **14.7 Exception.** Nothing contained in Section 14.1 will be construed to limit the Design-Builder, its Design Professionals and Design-Build Subcontractors rights, title and interest to continue to use their respective general design details that each of them uses or has used on multiple projects, or new standard design details that were developed during design of this facility.
- **14.8 Copies**. All copies made under this license will bear the statutory copyright notice, if any, shown on the Design and Construction Documents and any other documents prepared by the Design-Builder, its Design Professionals and Design-Build Subcontractors. Submittal or distribution to meet official regulatory requirements or for other purposes in connection with this Project will not be construed as publication in derogation of the Owner's copyright or other reserved rights and interests.

15. ACCOUNTING RECORDS

- **15.3** Audit. In accordance with Government Code Section 8546.7, records of both the Owner and the Design-Builder will be subject to examination and audit by the State Auditor General for a period of 10 years after final payment. Design-Builder will make available to the Owner any of the Design-Builder's other documents related to the Work immediately upon request of the Owner as set forth in Section 15.2.
- **15.4** Records. The Design-Builder will keep full and detailed accounts and exercise controls as may be necessary for proper financial management under this Agreement. In addition to the State Auditor rights above, the Owner will have the right during normal business hours to audit and copy the Design-Builder's documents related to this Project including, but not limited to, records, books, estimates, correspondence, instructions, drawings, receipts and invoices for materials, supplies and equipment, temporary facilities, etc., contracts, purchase orders, vouchers, memorandums, Change Orders and all substantiating documentation, certified payroll, and other data
relating to the Cost of Work, the Contract Price in order to evaluate accuracy and completeness of Design- Builder's billing. The Design-Builder will preserve all Project records for a period of at least 3 years after final payment, or for such longer period as may be required by law. The Design-Builder will incorporate Section 15 accounting and auditing provisions into all Design Professional agreements and Subcontracts and require Design Professionals and Subcontractors to keep detailed and accurate accounting records for their portion of the Work for a period of at least 3 years.

16. TERMINATION, SUSPENSION AND ABANDONMENT

- 16.1 **Termination for Fault.** The Owner may terminate this Agreement upon not less than 7 calendar days' written notice and an additional 7 calendar days to commence curing upon the Design-Builder's failure to perform any material obligation under the Agreement. The Design-Builder will have 7 days after receiving reasonably detailed written notice thereof from the Owner, provided that, if the nature of the breach is such that it will reasonably require more than 7 days to commence curing, the Owner may not terminate so long as Design-Builder (1) promptly, upon receipt of notice to cure, submits a plan to initiate all actions reasonably necessary to correct the default and prevent its reoccurrence, and (2) Owner accepts Design-Builder's plan, and (3) Design-Builder commences and continuously implements the plan to Owner's satisfaction. The Owner may also terminate this Agreement without notice or opportunity to cure upon the occurrence of the following Design-Builder events of default: 1) the failure to obtain and maintain any contract security instrument. 2) the failure to achieve acceptance of the Facilities through the acceptance process, and 3) the insolvency or bankruptcy of the Design- Builder. The notice will set forth the reason for termination and the effective date of termination. If the Owner terminates this Agreement for cause, the Design-Builder will not be entitled to any further payments except for work already completed. Unless otherwise limited herein, nothing stated in this paragraph will prevent the Owner from pursuing and recovering any damages allowed by law from Design-Builder arising out of a breach of this Agreement. If a court of competent jurisdiction deems that termination of the Design-Builder was wrongful or otherwise improper, the termination will be deemed a termination for convenience under Section 16.3.
- **16.2 Suspension By Owner.** If the Project is suspended by the Owner and not due to any fault of the Design-Builder or any of its Design Professionals or Subcontractors, the Design- Builder will be entitled to receive payment for all Work performed as of the effective date of the suspension, plus any documented reasonable direct costs incurred by Design-Builder to implement the suspension. The written notice of suspension will set forth the reason for suspension and the effective date of suspension. If the Project is resumed, and provided that the suspension was not caused or due to any fault or neglect of the Design-Builder or any of its Design Professionals or Subcontractors, then the Design-Builder's compensation will be equitably adjusted through Change Order under Section 9.1.4 and the Contract Time will be equitably adjusted for the additional time required to achieve Final Completion.

Termination For Convenience. The Owner may terminate this Agreement, in whole or in part, for convenience upon 30 calendar days' written notice at any time. The notice will state the extent of the termination and effective date of termination. For convenience termination during the Preconstruction Stage and through the issuance of a Notice to Proceed for Construction Stage, the Design-Builder will be entitled to receive payment for the percentage of Preconstruction Stage work completed, not to

exceed the Preconstruction Stage lump sum amount. For convenience termination following the issuance of a Notice to Proceed for Construction Stage Services, the Design-Builder will be entitled to payment for all Work performed as of the effective date of termination based on the compensation provisions set forth in Section 7 of this Agreement, as well as reasonable demobilization costs and unmitigable costs incurred by termination. In the event that the Owner terminates the Design-Build Agreement for convenience, Design-Builder may assign all subcontracts executed pursuant to the performance of the Design-Build Agreement to Owner promptly upon request. Design-Builder is entitled to compensation for all authorized payments made to any subcontractor prior to termination, which payments will be credited to Owner under the respective subcontracts, plus Design-Builder's approved costs that are incurred prior to any termination. In addition, in the event Owner terminates the Design-Build Agreement for convenience, Design-Builder must execute any documents establishing Owner's ownership of completed Work upon request. Any dispute over the amount to be paid upon termination will be resolved in accordance with the claim's procedures set forth in Section 14.12 of the Supplemental Conditions.

17. MISCELLANEOUS PROVISIONS

- **17.1 Governing Law**. This Agreement will be governed and construed in accordance with the laws of the State of California without regard to the principles of the conflict of laws. The Parties agree that any claim or enforcement of a judgment or alternative dispute award will be filed with the appropriate court of law in San Mateo County.
- **17.2 No Solicitation of Employees**. Owner will not solicit or employ any of Design-Builder's Project personnel for the duration of the Project.
- **17.3 Assignment.** The Owner and Design-Builder, respectively bind themselves, their partners, successors, assigns and legal representatives to the other party to this Agreement and to the partners, successors, assigns and legal representatives of the other party with respect to all covenants of this Agreement. Neither Owner nor Design-Builder will assign this Agreement without the written consent of the other, and such consent will not be unreasonably withheld or delayed.
- **17.4 Severability**. The terms and conditions of this Agreement will be interpreted in accordance with their plain meaning, and not strictly for or against either party. Any rule of construction or interpretation to the contrary will be of no force or effect with respect to this Agreement. If a court of competent jurisdiction finds any term or provision of this Agreement to be void or unenforceable for any reason that term or provision will be deemed severed, and the remainder of the Agreement will remain in full force and effect according to its terms and provisions, to the maximum extent permitted by law.
- **17.5 No Third-Party Beneficiaries**. Nothing contained in this Agreement creates a contractual relationship with, or a cause of action in favor of any third party against, either the Owner or Design-Builder. Owner and Design-Builder acknowledge and agree that the obligations of the Design-Builder are solely for the benefit of the Owner and are not intended in any respect to benefit any other third parties.
- **17.6 Waiver**. No action or failure to act by the Owner or Design-Builder will constitute a waiver of a right or duty afforded them under this Agreement, nor will such action or

failure to act constitute approval of or acquiescence in a breach of this Agreement, unless specifically agreed to in writing.

- **17.7 Time is of the Essence.** Time is of the essence with respect to each and every provision of the Agreement and any subsequent Change Orders.
- **17.8 Notice.** Any notice required to be given by this Agreement will be in writing and deemed effective upon personal delivery, or 1 business day after being sent via registered or certified mail return receipt requested or by overnight commercial courier providing next business day delivery and addressed to the following respective parties:

To Owner:	County of San Mateo Procurement Attention: Nicholas Calderon 455 County Center, 4 th Floor Redwood City, CA 94063
Copy to:	Matthew Estes Project Manager Capital Program Management, Inc. 1851 Heritage Lane, Suite 210 Sacramento, CA 95815
	and
	Office of the San Mateo County

Office of the San Mateo County Counsel Attn:, Mario Nastari 400 County Center, 4th Floor Redwood City, CA 94063

To Design-Builder:	Romtec, Inc.
Ū.	18240 North Bank Road
	Roseburg, OR 97470

- **17.9 Counterparts.** This Agreement may be executed in counterparts, each of which will be deemed an original, and all of which when taken together will constitute one instrument. The counterparts of this Agreement, and all amendments, must be manually executed, but the exchange of copies of this Agreement and of manually executed signature pages by facsimile or by electronic mail as an attachment in portable document format (.pdf) to the addresses provided in this Agreement shall constitute effective delivery of this Agreement as to the Parties and may be used as a fully binding original in lieu of the original Agreement for all purposes.
- **17.10 Modifications.** All modifications to the terms and conditions set forth in this Agreement must be in writing and signed by an authorized representative of both parties.

- **17.11** Section Headings. The Section headings contained in this Agreement are for reference purposes only and will not in any way affect the meaning or interpretation of this Agreement.
- **17.12 Legal Citations.** Legal citations to statutory requirements are included in the Agreement for convenience and an omission of any statutory requirement will not relieve the Design-Builder or its Design Professionals and Subcontractors from compliance with the law.
- **17.13 Exhibits.** The Supplemental Conditions and following Exhibits are incorporated by reference into the Agreement as though set forth infull.
- **17.14 Entire Agreement.** This Agreement represents the entire integrated agreement between the Owner and Design-Builder and supersedes all prior oral and written negotiations, representations or agreements by the parties with respect to this subject matter. This Agreement is entered into as of the Effective Date first written above.

COUNTY OF SAN MATEO

DESIGN-BUILDER

Signature

Signature

Printed Name

Printed Name

Exhibit 1 SUPPLEMENTAL CONDITIONS

DESIGN-BUILD AGREEMENT

TABLE OF CONTENTS

<u>Page</u>

1.	DEFINITIONS	2
2.	WORK RESTRICTIONS	6
3.	SOILS INVESTIGATIONS AND HAZARDOUS MATERIALS	6
4.	SAFETY	7
5.	QUALITY ASSURANCE AND QUALITY CONTROL	7
6.	TEMPORARY FACILITIES	9
7.	SURVEYING	11
8.	DEMOLITION	11
9.	PROTECTION OF WORK AND PROPERTY	12
10.	WORKERS AND WORKERS' COMPENSATION	13
11.	CHANGE IN NAME OR LEGAL ENTITY	13
12.	PROHIBITED INTERESTS	13
13.	LAWS AND REGULATIONS	13
14.	STATUTORY PUBLIC WORKS CONTRACT REQUIREMENTS	

SUPPLEMENTAL CONDITIONS TO DESIGN-BUILD AGREEMENT

1. **DEFINITIONS**

1.1 "Agreement" means the Design-Build Agreement between County of San Mateo and Design-Builder, dated August 13, 2024 including the Supplemental Conditions to the Agreement and all Exhibits.

1.2 "Allowance" is a non-binding, good faith, estimate of all Cost of Work attributable to the Allowance item carried in the Contract Price. An Allowance is necessary in certain circumstances because the item, components and/or systems are anticipated but undefined at the time that the Contract Price is set, and may require further development by the Architect of Record, Owner, or other Design Professionals "

1.3 Architect of Record" is the entity retained by the Design-Builder as the lead Design Professional that is responsible for the design of the Project. The Architect of Record for this Project is Romtec.

1.4 "Background Documents" means any existing drawings Owner provides to Design-Builder for reference. Background Documents are not part of the Contract Documents.

1.5 "Bridging Contract Documents" is defined in Section 4.2

1.6 "Change Order" is defined in Section 9.1.

1.7 "Construction Documents" means the documents developed by Design-Builder that establish all requirements for work during the Construction Stage of the Project. The Construction Documents will be informed by, and be consistent with the Bridging Contract Documents.

1.8 "Construction Stage Services" means all labor, materials, equipment and appurtenances provided by the Design-Builder and its Subcontractors to complete construction of the Project in strict accordance with the 100% Construction Documents and other components of the Contract Documents, ensure that all mechanical and support systems, as applicable, are properly and fully operational, and obtain all required certificates, approvals, and temporary or permanent permits for occupancy, use and completion of the Project have been issued by appropriate governmental authorities.

1.9 "Contract Documents" are set forth in Section 3.1 of the Agreement and include the Agreement inclusive of Exhibits, the Supplemental Conditions to the Agreement, and the Construction Documents, to be developed by the Design-Builder, and all subsequent contract modifications issued after execution of the Agreement, such as ChangeOrders.

1.10 "Contract Price" is set forth in Section 7 of the Agreement and reflects the sum total of all compensation due to the Design-Builder for all design and construction services under the Agreement. The Contract Price is the sum of the Preconstruction Price and the Construction Price.

1.11 "Contract Time" is the time within which the Design-Builder must achieve Final Completion of all Work on the Project. The Contract Time is set forth in Section 8.1.

1.12 "Daily Construction Reports" means the daily log kept by the Design-Builder that describes the weather, each Subcontractor's work on the site, the number of workers per trade, identification of equipment, construction work accomplished, problems encountered, and other similar relevant data such as accidents, service connections or disconnections, construction work stoppage, delays, material and labor shortages, and any applicable orders or requests from governing authorities.

1.13 "Design-Builder, Design-Build Entity" the entity that will enter into the Agreement with Owner and that will be the single point of accountability to Owner for delivering the services and the Project.

1.14 "**Design-Build Subcontractors**" means all Subcontractors that contract directly with the Design-Builder to perform design and construction services related to a specific trade or discipline.

1.15 "Design-Build Team" includes the General Contractor, Design Professionals and Design-Build Subcontractors members of the Design-Builder performing Preconstruction Stage Services and Construction Stage Services for the Project.

1.16 "Design Guide Illustrations" mean the drawings prepared by Owner and made a part of the Bridging Contract Documents.

1.17 "**Design Professionals**" means the Architect of Record, the structural engineer, and any other design consultants who are performing design services for the Project on behalf of Design-Builder but do not perform any Construction Stage Services.

1.18 "Design Services" includes all required design work required to complete the Project, consistent with the Bridging ContractDocuments.

1.19 "Effective Date" means the date that the Design-Builder and Owner entered into the Agreement, which is set forth on page 1 of the Agreement.

1.20 "Facilities" means all equipment, products, materials, controls, software, both individually and collectively as a completed system.

1.21 "Final Completion" occurs on the date when Design-Builder has achieved Substantial Completion; all Final Punch List items have been completed and accepted by the Owner; all close-out documentation required under the Project specifications has been transmitted to the Owner's Project Manager.

1.22 "Final Completion Date" is set forth in Section 8.1 of the Agreement.

1.23 "Final Design Package" is defined in Section 5.3 of the Design-Build Agreement.

1.24 "Final Punch List" is the punch list prepared by the Design-Builder in conjunction with the Owner after completing a Project walk-through upon Substantial Completion.

1.25 "Force Majeure Event" means an Act of God as defined under Public Contract Code section 7105, civil disobedience, an act of terror, or unavoidable casualties beyond the Design-Builder's control, and not due to any act or omission of the Design-Builder or its Design Professionals and/or Subcontractors, that necessarily extends the Final Completion Date.

1.26 "Hazardous Materials and Substances" means any substance, product, waste, or other material of any nature that is or becomes listed, regulated or addressed under one or more of the following Environmental Laws: (1) CERCLA, (2) Hazardous Materials Transportation Act, (3) RCRA, (4) the Clean Water Act, (5) the Toxic Substance Control Act, (6) HSAA, (7) the California Porter-Cologne Water Quality Control Act, (8) the California Hazardous Waste Management Act, (9) the California Safe Drinking Water Act, (10) the California Waste Management Act, and (11) any other Federal or State law or local ordinance concerning hazardous, toxic or dangerous substances, wastes, or materials.

1.27 "**Key Personnel**" means the Design-Builder's personnel identified as key to the overall success of the Project, and, at a minimum, including those positions defined as Key Personnel in the RFQ and RFP. The Design-Builder's Key Personnel are specifically identified in **Exhibit 5B**.

1.28 "Owner" means San Mateo County - Parks.

1.29 "**Owner-Elected Changes**" are changes in the Work directed by the Owner that may impact the Contract Price, and Final Completion Date and are not: (i) reasonably inferable from the Bridging Contract Documents or Contract Documents; or (ii) required as a result of design errors and omissions.

1.30 "Owner's Minimum Requirements" mean the performance specifications and prescriptive specifications prepared by Owner and made a part of the Bridging Contract Documents.

1.31 "Owner's Suspension of Work" is when the Owner elects to suspend progress of Work on the Project under Section 16.2 of the Agreement.

1.32 "Party" or **"Parties**" means the Design-Builder or the Owner in the singular or the Design-Builder and Owner collectively who have executed the Agreement.

1.33 "Preconstruction Stage Services" means all services, labor, materials, equipment and appurtenances provided by the Design-Builder and its Subcontractors to prepare a Preliminary Design for the Project, as further defined in Exhibit 3.

1.34 "Preconstruction Survey" Design-Builder's deliverable comprised of a comprehensive preconstruction survey of the Work site as described in more detail in the Scope of Work.

1.35 "Product Data" includes illustrations, standard schedules, performance charts, instructions, brochures, diagrams and other information furnished by the Design-Builder's Subcontractors illustrating materials or equipment for some portion of the construction work.

1.36 "Project" means the completion of the Preconstruction Stage Services and Construction Stage Services.

1.37 "Project Site" means that certain real property located at 555 Guadalupe Canyon Pkwy, Brisbane, CA 94005 commonly known San Bruno Mountain Park.

1.38 "Project Baseline Schedule" means the approved critical path schedule prepared by the Design-Builder for performance of all Work within the Contract Time, as approved by the Owner.

1.39 "Punch List" is a list prepared by the Owner's Project Manager, when Design-Builder considers a portion of the construction work substantially complete, that includes all items that are incomplete or unsatisfactorily finished and a schedule for their completion.

1.40 "**Request for Information**" ("RFI") means written requests prepared by the Design-Builder and/or its Subcontractors requesting clarification about design or raising coordination issues that impact design, cost or schedule. RFIs will be handled internally by the Design-Builder, with copies to the Owner.

1.41 "Samples" means physical examples of materials, equipment or workmanship required by the Construction Documents that are used to establish standards by which the construction work will be judged.

1.42 "Shop Drawings" means drawings, diagrams, and other data specially prepared by the Design-Builder and/or its Subcontractors, manufacturers, suppliers or distributors to demonstrate the way in which materials and equipment will perform in accordance with the design illustrated in the Construction Documents. Shop Drawings will be approved by Design-Builder, but Owner will also review.

1.43 "Site Logistics Plan" will provide phasing, establish the areas of the site that will be used for trailers, deliveries, staging, ingress and egress, location of major pieces of equipment, storage containers, stockpiles of materials, clearways used for emergency access, environmental controls, trailers for Design-Builder and Owner, parting facilities for Design-Builders, employees, and Owner, access road, fence line, etc.

1.44 "Specifications" means the component of the Construction Documents separate from the drawings, addressing all required materials, products and equipment, their installation and operation, quality assurances, reference standards, submittal requirements etc., not already addressed in the OMRs. The Specifications shall be developed in conjunction with the Construction Specifications Institute ("CSI") 16 Division/Three Part Format, as established in the CSI Manual of Practice.

1.45 "Staffing Plan" means the plan submitted by the Design-Builder with its proposal.

1.46 "Subcontractor" means all contractors under direct contract with Design-Builder for performance of a portion of the construction work as well as any lower tier-subcontractors. The term subcontractor includes Design-Build Subcontractors.

1.47 "Submittals" includes Shop Drawings, Product Data, Samples and similar documentation required by the Project specifications or other Construction Documents.

1.48 "Substantial Completion" means completion of all Preconstruction Stage Services and Construction Stage Services in accordance with the Contract Documents, and sufficient for the Owner to occupy and use the Facilities for their intended purpose; notwithstanding the foregoing, incomplete minor Punch List work that does not affect Owner's ability to occupy and use the Facilities for their intended purpose shall not prevent achievement of Substantial Completion.

1.49 "Substantial Completion Date" The Substantial Completion Date is set forth in Section 8.1 of the Agreement.

1.50 "Supplemental Conditions" means the Supplemental Conditions to the Agreement.

1.51 "Unforeseen Site Conditions" or **"Differing Site Conditions**" means discovery of unknown, unforeseen or differing site conditions, as defined in Public Contract Code section 7104, any unknown existing conditions in concealed spaces of the renovated portions of the Project.

1.52 "Work" means all work, including all services, labor, materials, equipment, tools, and appurtenances, necessary to complete the Preconstruction Stage Services and Construction Stage Services, as described in, or reasonably inferable from, the Contract Documents.

2. WORK RESTRICTIONS

2.1 Work Hours. All construction work will be performed between 7:00a.m. and 7:00p.m. unless further restricted by permit requirements or compliance with the US Department of Interior Department of Fish and Wildlife protective measures for marbled murrelet for construction during their breeding season. Design-Builder will provide Owner with written notice for any construction work that will need to be performed after hours. All after hour construction work requires Owner's written approval prior to commencement.

2.2 Signs. Design-Builder shall not erect any sign on the Project Site without the prior written consent of the Owner, which shall be at the sole discretion of the Owner.

2.3 Parking. Design-Builder works parking must be in Owner approved area.

2.4 Staging and Storage. Material will be stored only in the areas indicated on the Site Logistics Plan. Limited short-term staging areas will be designated in the Site Logistics Plan.

3. SOILS INVESTIGATIONS AND HAZARDOUS MATERIALS

3.1 Site Inspection. The Design-Builder is required to examine the Project Site before submitting its proposal. Design-Builder may not rely exclusively on Background Documents to determine the status of soil conditions, except for issues involving Unforeseen and Differing Site Conditions.

3.2 Hazardous Materials and Substances. The Design-Builder is responsible for the proper handling, removal and disposal of the "Hazardous Materials or Substances" that were pre-existing at the Project Site before commencement of construction and are part of the Work. The Design-Builder will not be considered the generator of any pre-existing hazardous materials on the Project Site. The Design-Builder is also responsible for all Hazardous Materials and Substances that it either requires through the Project design specifications or that are brought onto the Project Site by its employees and/or Subcontractors.

3.2.1 Unsafe or Hazardous Conditions. If reasonable precautions will be inadequate to prevent foreseeable bodily injury or death to persons resulting from any Hazardous Materials or Substances encountered at the Project Site, the Design-Builder will stop any part of the Work that it deems unsafe until corrective measures have been taken. If the Design-Builder fails to take corrective measures, the Owner may do so. Failure on the part of the Owner to stop unsafe practices, or the Owner's efforts to take corrective measures after the Design-Builder fails to do so, does not relieve or diminish the Design-Builder's safety responsibilities.

3.2.2 Verification. Upon discovery of any Hazardous Material or Substance that has not previously been identified in the Design-Builder's Hazardous Material Survey, the Design-Builder will immediately notify the Owner's Project Manager and stop all construction work in the area if necessary. The Design-Builder will retain the services of a licensed laboratory to verify the

presence or absence of the preexisting Hazardous Material or Substance. If preexisting Hazardous Material or Substance is discovered, the Design-Builder will contact its licensed laboratory to verify that the condition has been rendered harmless before construction work recommences in the affected area. The Design-Builder may be entitled to an adjustment in the Contract Time if the Hazardous Material or Substance is deemed an Unforeseen or Differing Site Condition and impacts the Final Completion Date of the Project. If the Hazardous Material or Substance was pre-existing the Owner will pay for the services of the licensed laboratory. The Design-Builder will reimburse the Owner for the services of the licensed laboratory if the Hazardous Material or Substance was brought on-site by the Design-Builder or any of its Subcontractors or vendors.

4. SAFETY

4.1 Signs. The Design-Builder will erect and maintain, as required by existing conditions and performance of the construction work, reasonable safeguards for safety and protection, including posting danger signs and other warnings against hazards, promulgating safety regulations, and notifying owners and users of adjacent sites and utilities.

4.2 Weekly Safety Meetings. The Design-Builder will hold weekly meetings with its Subcontractors to review Subcontractor compliance with the Design-Builder's Health and Safety Program.

4.3 Daily Jobsite Walks. The Design-Builder will also conduct daily jobsite inspections to verify that the construction work is being performed in a safe and workmanlike manner and in accordance with the Design-Builder's Health and Safety Program. The Design-Builder will provide written notice to its Subcontractors demanding immediate correction of any known safety violation.

5. QUALITY ASSURANCE AND QUALITY CONTROL

5.1 Quality Control Plan. The Design-Builder will prepare and submit to the Owner's Project Manager for approval a plan that describes the procedures and methods the Design-Builder will utilize to control the quality of the construction work. The Quality Control Plan must be approved before the start of construction. The Owner reserves the right to require revisions of the Quality Control Plan that are necessary to ensure the specified quality of the construction work. The Design-Builder will assign appropriate site personnel to oversee quality control. No change in the Quality Control Plan will be implemented without prior Owner approval. At a minimum the Quality Control Plan will provide information regarding the following:

5.1.1 Quality control supervision and document control.

5.1.2 Identification of personnel for required training and qualification activities.

5.1.3 Procedures for testing and inspections that identify individual inspection or testing points and acceptance criteria, and include provisions for recording results and the responsible inspection/test personnel.

5.1.4 Procedures for identifying what applicable technical and quality requirements will be required of vendors supplying materials, parts and services to ensure compliance with the Contract Documents.

5.1.5 Procedures for receiving, inspecting and accepting materials and equipment. The procedures will include, at a minimum, examination of the physical condition for compliance

with the Contract Documents, purchase order and/or subcontract agreement, and identifying and processing any non-conforming goods.

5.1.6 Provisions for identifying and timely remedying non-conforming or defective construction work.

5.1.7 Documentation control to maintain records of the activities included in the Quality Control Plan. All documentation will be submitted to the Owner as part of the close-out documentation for this Project and therefore must be logically organized and indexed for reference.

5.2 Design Quality Control Plan. The Design-Builder will prepare and submit to the Owner for approval a Design-Quality Control Plan that describes the procedures and methods the Design-Builder will utilize to control the quality of the construction work. The Design Quality Control Plan must be approved before the start of construction. The Owner reserves the right to require revisions of the Design Quality Control Plan that are necessary to ensure the specified quality of the construction work. The Design-Builder will assign appropriate site personnel to oversee quality control. No change in the Design Quality Control Plan will be implemented without prior Owner approval.

5.3 Manufacturer's Field Services. To the extent required, the Design-Builder will engage in a factory-authorized service representative to inspect field-assembled components and equipment installation, including service connections. All manufacturers' field service reports must be in writing and included as part of the records turned over to Owner during close- out.

5.3.1 Quality Control Reports. The Design-Builder will keep daily Quality Control Reports throughout the duration of the construction process certifying that the relevant area of the construction work has been inspected. The Quality Control Reports will be prepared, signed and dated by the personnel identified as the supervisor in the Quality Control Plan and will include, at a minimum, the following information: Identification of the material, equipment or component that was inspected and indicate, if applicable, if the Submittals have been reviewed and approved by the Design-Builder.

5.3.2 Indicate that materials and/or equipment comply with the requirements of Section 5.19.9 of the Agreement and are properly stored, if not yetinstalled.

5.3.3 Indicate that the construction work has been coordinated underSections 5.19.4 and 5.19.5 of the Agreement, that all required preliminary work has been inspected by Quality Control personnel, was properly performed, and that the area is ready to receive subsequent construction work. If the construction work is not acceptable, provide a written description of any rework required in the area inspected with an explanation of the cause of the rerework (including which Subcontractors are involved), any cost involved in the required re-work, and the expected completion date of the required re-work.

5.4.4 Results of any off-site testing or quality control work and any required further actions.

5.4.5 Other necessary information including, directions received, quality control problem areas, deviations from the Quality Control Plan, construction deficiencies encountered, Quality Control meetings held, acknowledgement that as-built drawings have been updated (if applicable), corrective direction given by Quality Control personnel, and corrective action taken by the Design-Builder.

5.5 Quality Control Design Reports. The Design-Builder will keep daily Quality Control Design Reports throughout the duration of the construction process certifying that the relevant area of the construction work has been inspected. The Quality Control Design Reports will be prepared, signed and dated by the personnel identified as the supervisor in the Design Quality Control Plan. Quality Control Design Reports should be submitted as part of each design submittal.

5.6 Test and Inspection Logs. The Design-Builder will maintain an on-site inspection log that is accessible by the Owner. The log will document all tests and inspections performed at the Project during construction. In addition, the Design-Builder will prepare a sequentially numbered record of tests and inspections. The record of tests will Include the following information:

- **5.6.1** Request for Inspection.
- **5.6.2** Date test or inspection was conducted.
- **5.6.3** Identity of testing agency or special inspector.
- 5.6.4 Description of the construction work tested or inspected.

5.6.5 Identification of any drawings or applicable details on the Construction Documents or Submittals that were used during testing and inspection.

5.6.6 Date that the test or inspection was concluded and the date that the results were transmitted to Owner.

6. TEMPORARY FACILITIES

6.1 Temporary Electricity. Design-Builder will provide, maintain, and pay for temporary electrical power at the Project Site for construction purposes and trailers.

6.2 Temporary Communications. The Design-Builder will provide, maintain, and pay for all applicable communications and data service connections for field offices pursuant to **Exhibit 6B** of the Agreement, including all installation and connection charges.

6.3 Temporary Water. The Design-Builder will provide, maintain, and pay for all required potable water required for construction field personnel as well as water required for and in connection with the construction operations such as dust control. Unnecessary waste of water will not be permitted. The Design-Builder must use special hydrant wrenches for opening and closing fire hydrants in lieu of pipe wrenches.

6.4 Temporary Fences. The Design-Builder will provide all necessary temporary fencing and gates required for the Project Site. Temporary fencing will be subject to restrictions in the use permit. The Design-Builder will maintain all fences through Final Completion of the Project. Gates are to remain closed and locked duringoff-hours.

6.5 Temporary Sanitary Facilities. Provide and maintain all required temporary toilets for use of all design and construction personnel and field labor at the Project Site through Final Completion of the Project. Location of temporary sanitary facilities will be approved by Owner's Project Manager prior to delivery. The Design-builder will provide at least 1 temporary toilet facility for every 20 persons. The Design-Builder will cause all design and construction personnel (including field labor) to use temporary sanitary facilities rather than Owner's facilities. All temporary sanitary facilities will comply with the Department of Health standards.

6.6 Temporary Barriers and Enclosures. Provide barriers to prevent unauthorized entry to construction areas, to allow for Owner's safe use of the Project premise, and to protect existing facilities and adjacent properties from damage from construction operations per Section 8.

6.7 Water Control. Design-Builder will grade the Project Site as required by the civil design included in the Construction Document. During construction, the Design-Builder will maintain all trenches and excavated areas free from water accumulation and will provide the necessary barriers to protect the Project Site from ponding, running water and soil erosion. The Design-Builder will provide for increased drainage of storm water and any water that may be applied or discharged on the Project Site during performance of the construction work. All drainage facilities will be adequate to prevent damage to the construction work, Project Site, and adjacent property. Design-Builder will construct dikes, if necessary, to divert any increased runoff from entering adjacent property (except in natural channels), to protect Owner's facilities and the construction work, and to direct water to drainage channels or conduits. Design-Builder will provide for any loss or damages resulting from Design-Builder's failure to comply with the provisions of the Dept. of Water Resources Best Management Practices and County requirements.

6.8 Pollution Control. The Design-Builder will provide a plan that meets the requirements of California Storm Best Management Practices (Stormwater Quality Task Force, 1993) to prevent the pollution of drains and watercourses by sanitary wastes, sediment, debris, and other substances and/or soil erosion during construction operations:

6.8.1 No sanitary wastes will be permitted to enter any drain or watercourse other than sanitary sewers. No sediment, debris, or other substance will be permitted to enter sanitary sewers without authorization of the receiving sanitary sewer service, and all possible best management practices will be taken to prevent materials from entering into any drain to watercourse.

6.8.2 In the event that dewatering of excavations is required, Design-Builder will obtain the necessary permits from local governmental authorities for discharge of the dewatering effluent. The Design-Builder will be responsible for assuring that water quality of the discharge meets the appropriate permit requirements prior to anydischarge.

6.8.3 Erosion and sedimentation control practices will include installation of silt fences, straw wattle, soil stabilization, re-vegetation, and runoff control to limit increases in sediment in storm water runoff, including but not limited to, detention basins, straw bales, silt fences, check-dams, geo-fabrics, drainage swales, and sand bagdikes.

6.8.4 The construction work will be scheduled to expose areas subject to erosion for the shortest possible time, and natural vegetation will be preserved to the greatest extent practicable. Temporary storage and construction buildings will be located, and construction traffic routed, to minimize erosion. Temporary fast-growing vegetation or other suitable ground cover will be provided as necessary to controlrunoff.

6.9 Construction Equipment and Aids. Design-Builder will furnish, install, maintain, and operate all construction equipment required by the performance of the construction work. Construction aids include elevators and hoists, cranes, temporary enclosures, swing staging, scaffolding and temporary stairs. When sandblasting, spray painting, spraying of insulation, or other activities inconveniencing or dangerous to property or the health of design or construction

personnel, Owner's staff, or the public are in progress, Design-Builder will enclose the area of activity to contain the dust, over-spray, or other hazard.

6.10 Traffic Control. The Design-Builder will provide a traffic control plan in accordance with the California Department of Transportation Traffic Manual as part of its Site Logistics set forth in **Exhibit 6B**. The Design-Builder will submit its traffic control plan to the appropriate agency for approval, as necessary, before commencement of the construction work:

6.11 Removal of Temporary Facilities and Equipment. The Design-Builder will remove all temporary utilities, equipment, facilities, and materials before final inspection of the Project and clean and repair any damage caused by installation or use of temporary work restoring existing facilities to their original conditions.

7. SURVEYING

7.1 Field Engineering. The Design-Builder will employ a California State licensed civil engineer or land surveyor to provide field engineering services to establish benchmarks and line and grade for horizontal and vertical control.

8. **DEMOLITION**

8.1 Demolition Plan. Prior to commencing any required demolition work, the Design-Builder will submit a plan to the Owner's Project Manager for review and approval. Under no circumstances, can demolition interrupt the Owner's operations. The Design-Builder's plan, at a minimum, will address the following:

8.1.1 Identify areas that will require demolition and provide a schedule for those demolition activities that is coordinated with the Owner's operations and the approved Project Baseline Schedule.

8.1.2 Inventory materials and equipment that will be salvaged during demolition and whether the salvaged materials and equipment will be reused, returned to the Owner, or sold at fair market value on behalf of the Owner.

8.1.3 Document procedures for protecting the existing structure and/or building materials, equipment and components that are remaining, as well as protection plans for adjacent property and persons.

8.1.4 Document procedures for proper ventilation, noise, and dust control during demolition operations and clean-up after demolition is completed.

8.1.5 Document procedures for required disruption of any utility service as a result of demolition activities and a record of any utilities that are capped during the process. Any required shut-off or interruption of service must be approved in writing by the Owner 14 business days in advance, and all necessary water, emergency power, etc., must be in place prior to shut-off or disruption.

8.1.6 Provide for all required temporary sheeting, shoring, bracing or other structural support necessary to ensure stability of the existing structure or adjacent properties and prevent movement, settlement or collapse during demolition operations. All required temporary structural support will be designed by a California licensed structural or civil engineer.

8.1.7 Document procedures to deal with encountering Hazardous Materials or Substances that comply with the requirements of Supplemental Conditions Section 3 and procedures regarding Unforeseen or Differing Site Conditions that comply with Supplemental Conditions Section 14.14.

8.1.8 Document procedures for hauling away and disposal of any demolished materials and equipment. The procedures should include, among other things, requirements for refrigerant recovery under Environmental Protection Agency, a list of all required hauling permits, requirements for hauling and disposing of Hazardous Waste, volatile organic compounds or any other substance that is regulated by Health and Safety Code, the Bay Area Air Quality Management Owner "BAAQMD" or any other governmental agency that regulates the proper hauling and disposal of certain materials and substances.

8.1.9 Document procedures to ensure that removal and replacement of equipment will not void any existing warranties.

8.1.10 Require a survey of existing conditions and video or photographic documentation before commencement of the demolition activity to demonstrate existing conditions of adjacent areas or property.

8.2 Permits and Fees. The Design-Builder will secure all required hauling permits. The Owner will reimburse the Design-Builder at cost, for all permits and dumping fees as part of the Cost of Work.

9. PROTECTION OF WORK AND PROPERTY

9.1 Design-Builder will be responsible for providing a safe place for the performance of the construction work and for the physical conditions and safety of areas affected by the construction work. Design-Builder will take all necessary precautions to provide for the safety and protection of all persons who may come in contact with the construction work and for all property and equipment within or adjacent to the Project Site including adequate precautions to protect existing trees, equipment, materials, utilities, and other adjoining property and structures. Design-Builder will repair any damage caused by its operations at its own expense and will provide protection to prevent damage, injury or loss to:

9.1.1 Owner's employees and other persons at the Project Site.

9.1.2 Equipment, materials, and vehicles stored at the site or off-site if under the care, custody, or control of the Design-Builder or its Design Professionals or Subcontractors.

9.1.3 Existing trees, structures, roads, equipment, property and the work of others when carrying out Design-Builder's Work. Refer to Tree Protection specifications as included as part of the Criteria document includes as **Exhibit 2B**.

9.2 These precautionary measures will apply continuously and not be limited to normal working hours.

9.3 If damage to persons or property occur as a result of the construction work, Design-Builder will be responsible for proper investigation, documentation, including video or photography, to adequately memorialize and make a record of what transpired. The Owner will be entitled to inspect and copy any documentation, video, orphotographs.

10. WORKERS AND WORKERS' COMPENSATION

10.1 Design-Builder will at all times enforce strict discipline and good order among its employees. Design-Builder will not employ on the Project any unfit person or unskilled labor.

10.2 Design-Builder and its Subcontractors are required to secure the payment of compensation of its employees in accordance with Labor Code section 3700. Before commencing the Work, the Design-Builder, its Design-Build Team members, and its Subcontractors will sign and file a certification with the Owner under Labor Code section 1861 stating the following:

I am aware of the provisions of Section 3700 of the Labor Code which require every employer to be insured against liability for workers' compensation or to undertake self- insurance in accordance with the provisions of that code, and I will comply with such provisions before commencing the performance of the any work or services under the Design-Build Agreement or any subcontract or design service agreements.

11. CHANGE IN NAME OR LEGAL ENTITY

11.1 If a change in name or nature of the Design-Builder's legal entity is anticipated, the Design-Builder will notify the Owner to ensure that the change will be properly reflected on the Agreement.

12. PROHIBITED INTERESTS

12.1 No public official or representative of the Owner who is authorized in such capacity and on behalf of the Owner to negotiate, supervise, make, accept, approve, or to take part in negotiating, supervising, making, accepting or approving any engineering, inspection, construction or material supply contract or any subcontract in connection with design and construction of the Project, will be or become directly or indirectly interested financially in this Agreement.

13. LAWS AND REGULATIONS

13.1 Design-Builder will give all notices and comply with all laws, ordinances, rules and regulations bearing on conduct of Work. If Design-Builder observes that the Contract Documents are at variance with any laws, ordinances, etc., Design-Builder will promptly notify the Owner's Project Manager, in writing, and any necessary changes will be adjusted. If Design-Builder performs any Work knowing it to be contrary to such laws, ordinances, rules and regulations, and without notice to the Owner's Project Manager, it will bear all costs associated with any required corrections or repairs.

14. STATUTORY PUBLIC WORKS CONTRACT REQUIREMENTS

The following requirements apply to all public works construction work performed under this Agreement.

14.1 Public Works Registration. Design-Builder or its contractor, and all subcontractors, must be registered with the California Department of Industrial Relations pursuant to Labor Code Section 1725.5. This Agreement is subject to monitoring and enforcement by the DIR pursuant to Labor Code Section 1771.4. Design-Builder shall furnish the records specified in Labor Code Section 1776 directly to the Labor Commissioner on a monthly basis, and in a format prescribed by the Labor Commissioner. Design-Builder must also post notices at the work site pursuant to Title 8 California Code of Regulations Section 16451.

14.2 Use of Subcontractors. Design-Builder shall not subcontract any work to be performed by it under this Agreement without the prior written approval of Owner, which approval will not be unreasonably withheld. Design-Builder shall be solely responsible for reimbursing any subcontractors and Owner shall have no obligation to them. Attention is directed to the requirements of Section 4100 to 4113, inclusive of the California Public Contract Code which are applicable to the work covered by this Agreement.

14.3 Prohibition Against Contracting with Debarred Subcontractors. Design- Builder is prohibited from performing work on a public works project with a subcontractor who is ineligible to perform work on the public works project pursuant to Section 1777.1 or 1777.7 of the Labor Code.

14.4 Prompt Payment to Subcontractors. Design-Builder shall pay any subcontractors approved by Owner for work that has been satisfactorily performed no later than seven (7) days from the date of Design-Builder's receipt of progress payments by Owner. Within thirty (30) days of receipt of retention by Design-Builder and satisfactory completion of all work required of the subcontractor, Design-Builder shall release any retention payments withheld to the subcontractor. In the event Design-Builder does not make progress payments or release retention to the subcontractors in accordance with the time periods in this section, Design-Builder will be subject to a charge of two percent (2%) per month on the untimely or improperly withheld payment. Owner may require Design-Builder to provide documentation satisfactory to Owner of Design-Builder's compliance with this requirement as a condition of final payment and release of contract retentions, if any.

14.5 Payment Bond for Construction Work. Pursuant to Civil Code Section 9550, Design-Builder shall furnish to Owner a Payment Bond in the amount of all equipment and construction costs, to provide Owner with security for Design-Builder's full payment to workers and subcontractors for costs of materials, equipment, supplies, and labor furnished in the course of the performance of the work applicable to this section.

14.6 Labor Code Provisions. In the performance of this Contract, Design-Builder's attention is directed to the following requirements of the LaborCode:

Hours of Labor. Eight hours labor constitutes a legal day's work. Design-Builder shall forfeit, as penalty to Owner, \$25 for each worker employed in the performance of the Agreement by Design-Builder or by any subcontractor under it for each calendar day during which such worker is required or permitted to work more than eight hours in any one day and 40 hours in any one calendar week in violation of the provisions of the California Labor Code and in particular, Sections 1810 to 1815, inclusive. Work performed by employees of the Design-Builder in excess of eight hours per day and 40 hours during any one week shall be permitted upon compensation for all hours worked in excess of eight hours per day at not less than one- and-one-half times the basic rate of pay, as provided in Section 1815.

<u>Prevailing Wages.</u> Design-Builder shall comply with California Labor Code Sections 1770 to 1780, inclusive. In accordance with Section 1775, the Design-Builder shall forfeit as a penalty to Owner an amount as determined by the Labor Commissioner not to exceed \$200 for each calendar day or portion thereof for each worker paid less than stipulated prevailing wage rates for such work or craft in which such worker is employed for any work done under the Agreement by him or by any subcontractor under it in violation of the revisions of the Labor Code and in particular, Labor Code Sections 1770 to 1780, inclusive. In addition to said penalty and pursuant to Section 1775, the difference between such stipulated prevailing wage rates and the amount paid to each worker for each calendar day or portion thereof for which each worker was paid less than the stipulated

Exhibit 1 - Supplemental Conditions

prevailing wage rate shall be paid to each worker by Design-Builder. Pursuant to the provisions of Section 1773 of the Labor Code, Owner has obtained the general prevailing rate of per diem wages and the general prevailing rate for holiday and overtime work applicable to the work to be done from the Director of the Department of Industrial Relations. Copies of the prevailing wage rates are on file at Owner and are available for review upon request.

<u>Payroll Records.</u> The Design-Builder's attention is directed to the following provisions of Labor Code Section 1776. The Design-Builder shall be responsible for the compliance with these provisions by his subcontractors.

- a. Each contractor and subcontractor shall keep an accurate payroll record, showing the name, address, social security number, work classification, straight time and overtime hours worked each day and week, and the actual per diem wages paid to each journeyman, apprentice, worker, or other employee employed by him or her in connection with the public work.
- b. The payroll records enumerated under subdivision (a) shall be certified and shall be available for inspection at all reasonable hours at the principal office of the Design-Builder on the following basis:
 - 1. A certified copy of an employee's payroll record shall be made available for inspection or furnished to such employee or his or her authorized representative on request.
 - 2. A certified copy of all payroll records enumerated in subdivision (a) shall be made available for inspection or furnished upon request to Owner, the Division of Labor Standards Enforcement and the Division of Apprenticeship Standards of the Department of Industrial Relations.
 - 3. A certified copy of all payroll records enumerated in subdivision (a) shall be made available upon request to the public for inspection or copies thereof made; provided, however, that a request by the public shall be made through either Owner, the Division of Apprenticeship Standards or the Division of Labor Standards Enforcement. If the requested payroll records have not been provided pursuant to paragraph (2), the requesting party shall, prior to being provided the records, reimburse the costs of preparation by the Design- Builder, subcontractor and the entity through which the request was made. The public shall not be given access to such records at the principal office of the Design-Builder.
- c. The certified payroll records shall be on forms provided by the Division of Labor Standards Enforcement or shall contain the same information as the forms provided by the Division.
- d. The Design-Builder shall file a certified copy of the records enumerated in subdivision (a) with the entity that requested such records within ten (10) days after receipt of a written request.
- e. Any copy of records made available for inspection as copies and furnished upon request to the public or Owner, the Division of Apprenticeship Standards or the Division of Labor Standards Enforcement shall be marked or obliterated in such a manner as to prevent disclosure of an individual's name, address and social security number. The name and address of the Design-Builder shall not be marked or obliterated. The Design-Builder

shall inform Owner of the location of records enumerated under subdivision (a), including the street address, city and county, and shall, within five working days, provide a notice of a change of location and address.

- f. In the event of noncompliance with the requirements of this Section, the Design-Builder shall have ten (10) days in which to comply subsequent to receipt of written notice specifying in what respects such contractor must comply with this Section. Should noncompliance still be evident after such 10-day period, the Design-Builder shall, as a penalty the State or Owner, forfeit Twenty-five Dollars (\$25) for each calendar day, or portion thereof, for each worker, until strict compliance is effectuated. Upon the request of the Division of Apprenticeship Standards or the Division of Labor Standards Enforcement, such penalties shall be withheld from progress payments then due. The penalties specified in subdivision (g) of Labor Code Section 1776 for noncompliance with the provisions of said Section 1776 may be deducted from any monies due or which may become due to the Design-Builder.
- g. The Design-Builder and each subcontractor shall preserve their payroll records for a period of three (3) years from the date of completion of theContract.

Labor Non-discrimination. Attention is directed to Section 1735 of the Labor Code which provides that Design-Builder shall not discriminate against any employee or applicant for employment because of race or color, religion, physical or mental disability, national origin or ancestry, medical condition, marital status or sex of such persons, except as provided in Section 12940 of the Government Code. Design-Builder further agrees to include a similar provision in all subcontracts, except subcontracts for standard commercial supplies or raw materials.

<u>Apprentices.</u> The Design-Builder and all subcontractors shall comply with the requirements of California Labor Code sections 1777.5, 1777.6 and 1777.7 regarding the employment and of apprentices.

14.7 Skilled and Trained Labor Force Requirements. Design-Builder agrees to comply with all requirements related to providing a skilled and trained workforce, pursuant to Public Contract Code section 22164(c), and Public Contract Code sections 2600-2603, including but not limited to the requirement to submit monthly reports to the Owner.

14.8 Retention on Progress Payments. Owner will deduct and hold in retention five percent (5%) from each progress payment to Design-Builder for construction work, or portion thereof. The remainder, less any other deductions taken in accordance with the Agreement, will be paid to Design-Builder as progress payments.

14.9 Securities in Lieu of Retention. Pursuant to Public Contract Code Section 22300, Design-Builder may elect, in lieu of having progress payments retained by Owner, to deposit in escrow with Owner, or with a bank acceptable to Owner, securities eligible for investment under Government Code Section 16430, bank or savings and loan certificates of deposit, interest bearing demand deposit accounts, standby letters of credit, or any other security mutually agreed to by Design-Builder and Owner. If Design-Builder elects to submit securities in lieu of having progress payments retained by Owner, Design-Builder shall, at the request of any subcontractor performing more than 5% of Design-Builder's total bid, make the same option available to the subcontractor.

14.10 Assignment of Claims. In entering into a public works contract or a subcontract to supply goods, services, or materials, Design-Builder or subcontractor offers and agrees to assign to

Owner all rights, title, and interest in and to all causes of action it may have under Section 4 of the Clayton Act (15 U.S.C. Sec. 15) or under the Cartwright Act (Chapter 2 (commencing with Section 16700) of Part 2 of Division 7 of the Business and Professions Code), arising from purchases of goods, services, or materials pursuant to the public works contract or the subcontract. This assignment shall be made and become effective at the time Owner tenders final payment to Design-Builder, without further acknowledgement by the parties.

14.11 Third-Party Claims. Pursuant to Public Contracts Code Section 9201, Owner shall have full authority to compromise or otherwise settle any claim relating to the Agreement at any time. Owner shall provide for timely notification to Design-Builder of the receipt of any third-party claim, relating to the contract. Notice shall be in writing and will be provided within thirty (30) days.

14.12 Public Contract Code Claims Procedures

14.12.1 <u>Mandatory Prerequisites to Filing a Construction Claim.</u> Prior to filing a construction claim pursuant to Public Contract Code Sections 9203 and 20104-20104.6 and this section, Design-Builder must first complete all Change Order procedures in Section 9 of the Design-Build Agreement. Any claim submitted prior to satisfaction of the Change Order procedures will be rejected as premature and untimely. A construction claim must be submitted no later than (a) 30 days after the completion of all Dispute Resolution Board procedures are completed, or (b) 30 days after the occurrence of the event giving rise to the claim.</u>

14.12.2 <u>Claims Procedures.</u> In accordance with the procedures set forth in Public Contract Code sections 9204 and 20104-20104.6, Design-Builder may submit a claim by registered or certified mail with return receipt requested, for one or more of the following: (a) a time extension, including, without limitation, for relief from damages or penalties for delay assessed by the Owner; (b) payment by the Owner of money or damages arising from work done by, or on behalf of, the Design-Builder pursuant to this contract and payment for which is not otherwise expressly provided or to which the Design-Builder is not otherwise entitled; or (c) payment of an amount that is disputed by the Owner.

14.12.3 Support for Claim. The Design-Builder shall furnish reasonable documentation to support the claim, including but not limited to: 1) a clear, concise recital of the basis upon which the claim is asserted, including a designation of the provisions of the Contract Documents upon which the claim is based, 2) a statement as to the amount of time and/or compensation sought pursuant to the claim; 3) whether the Design-Builder's claim arises from an ongoing occurrence, and if so a description of the specific Work activities affected by the claim, 4) a time impact analysis in the event that Design-Builder requests a time extension, 5) full and complete cost records supporting the amount of any claim for additional compensation, and 6) a notarized certification by the Design-Builder as follows: "Under the penalty of law for perjury or falsification and with specific reference to the California False Claims Act, Government Code Section 12650 et seq., the undersigned hereby certifies that the information contained herein is a true, accurate and complete statement of all features relating to the claim asserted." Failure by the Design-Builder to provide sufficient documentation, or clarification of the documentation provided.

14.12.4 <u>Response to Claim.</u> Upon receipt of a claim, the Owner will conduct a reasonable review and provide a written statement to the Design-Builder identifying what portion of the claim is disputed and what portion is undisputed within 45 days of receipt of the claim. The Owner and Design-Builder may, by mutual agreement, extend the 45 day time period. For any

undisputed portion of a claim, the Owner must make payment within 60 days of its issuance of the written statement.

If the Design-Builder disputes the Owner's written statement, or if the Owner fails to respond, the Design-Builder may demand an informal conference to meet and confer for settlement of the issues in dispute. The Owner will then schedule the meet and confer conference within 30 days of the demand. Within 10 business days following the meet and confer conference, the Owner will provide a written statement identifying the portion of the claim that remain in dispute. Any payment due on an undisputed portion of the claim will be made within 60 days of the meet and confer conference.

After the meet and confer conference, any disputed portion of the claim shall be submitted to non-binding mediation. Alternatively, upon receipt of a claim, the parties may mutually agree to waive, in writing, mediation and proceed directly to the commencement of a civil action or binding arbitration, as applicable. If mediation is unsuccessful, the parts of the claim that remain in dispute shall be subject to applicable procedures set forth below.

Failure of Owner to respond to a claim within the time periods described above shall result in the claim being deemed rejected in its entirety. Additionally, amounts not paid in a timely manner shall bear interest at 7 percent per year.

In the event that the mediation is unsuccessful, Design-Builder must file a government claim pursuant to Government Code section 910 et seq. in order to initiate a civil action.

14.13 Utility Relocation. Pursuant to California Government Code Section 4215, if during the course of the work Design-Builder encounters utility installations which are not shown or indicated in the contract plans or in the specifications or which are found in a location substantially different from that shown, and such utilities are not reasonably apparent from visual examination of the work site, then it shall promptly notify Owner in writing. Where necessary for the work of the Contract, Owner will amend the Agreement to adjust the scope of work to allow Design-Builder to make such adjustment, rearrangement, repair, removal, alteration, or special handling of such utility, including repair of the damaged utility. If Design- Builder fails to give the notice specified above and thereafter acts without instructions from Owner, then it shall be liable for any or all damage to such utilities or other work of the Agreement which arises from its operations subsequent to the discovery, and it shall repair and make good such damage at its own cost.

14.14 Trenching, Shoring, and Differing Site Conditions.

14.14.1 Compliance. Design-Builder will comply with Labor Code sections 6500, 6705, and 6707, and Public Contract Code section 7104 regarding trenching and shoring.

14.14.2 Permit Requirements for Trenches 5'-0" or More in Depth. Design-Builder agrees to comply in full with Section 6500 of the Labor Code and to provide the required permits prior to the initiation of any work, method, operation or process that involves: (i) construction of trenches or excavations that are 5'-0" or deeper and into which a person is required to descend; (ii) the construction of any building, structure, falsework, or scaffolding more than 3 stories high or the equivalent height; (iii) the demolition of any building, structure, falsework, or scaffold more than 3 stories high or the equivalent height; (iv) the underground use of diesel engines in work in mines and tunnels.

14.14.2.1Detailed Plans for Trenches 5'-0" or More in Depth. Incompliance with Labor Code section 6705, the Design-Builder will submit to the Owner's Project

Exhibit 1 - Supplemental Conditions

Manager, in advance of excavation, a detailed plan showing the design of shoring, bracing, sloping or other provisions to be made for worker protection from the hazard of caving ground during the excavation of any trench or trenches 5'-0" or more in depth. If the plan varies from shoring system standards, the plan will be prepared by a registered civil or structural engineer. The plan will not be less effective than the shoring, bracing, sloping, or other provisions of the Construction Safety Orders, as defined in the California Code of Regulations.

14.14.2.2 Separate Bid Items for Sheeting, Shoring, etc. To the extent that Design-Builder's Work involves construction of a pipeline, sewer, sewage disposal system, boring and jacking pits, or similar trenches or open excavations, which are 5'-0" or deeper, Design-Builder will comply with all applicable laws, regulations, and codes and its bid and the Contract Price will contain, as a line item, adequate sheeting, shoring, and bracing, or equivalent method, for the protection of life or limb pursuant to Labor Code section 6707, which will conform to applicable safety orders. Nothing in this section will be construed to impose tort liability on the Owner or any of its employees.

14.14.3 Excavations Deeper than 4'-0". If Work under this Agreement involves digging trenches or other excavation that extends deeper than 4'-0" below the surface, Design-Builder will promptly, and before the following conditions are disturbed, notify Owner's Project Manager, in writing, in accordance with Public Contract Code section 7104, of any:

14.14.3.1 Material that the Design-Builder believes may be hazardous waste, as defined in Section 25117 of the Health and Safety Code, which is required to be removed to a Class I, Class II, or Class III disposal site in accordance with provisions of existing law.

14.14.3.2 Subsurface or latent physical conditions at the site differing from those indicated.

14.14.3.3 Unknown physical conditions at the site of any unusual nature, different materially from those ordinarily encountered and generally recognized as inherent in the construction work of the character provided for in the Bid Documents and under this Agreement.

14.14.4 Differing Site Conditions. Design-Builder's notice to Owner shall be issued by telephone or in person and followed within 24 hours thereafter by written notice, providing a brief description of why the condition encountered is considered a Differing Site Condition. Promptly upon receipt of Design-Builder's notice, Owner will investigate the site conditions. If, during construction, the Design-Builder encounters an alleged Differing Site Condition, the Design-Builder shall immediately give written notice and may continue work; provided however that the following documents and information shall be submitted on a daily basis:

- 1. Digital photographs (paper and electronic copy) that detail the Differing Site Conditions;
- 2. An electronic copy of the pertinent data (e.g. settlement monitoring data, boring logs, dewatering production rates, etc.) for the previous 24 hours;
- 3. As applicable, sample of soil and groundwater in the alleged Differing Site Condition area.
- 4. Design-Builder's applicable daily reports for each day that the alleged Differing Site Condition exists; and

5. Detailed daily records (which shall include, but not be limited to, labor and equipment), describing the alleged Differing Site Conditions and the impact the Differing Site Conditions are having on the progress of the construction.

Immediate written notice shall describe the specific ground conditions encountered and the measures taken to deal with the ground conditions. The Design-Builder will provide the OR with written notice within 5 business days discovery of an Unforeseen and Differing Site Condition. The OR, in conjunction with the Owner and IOR, will promptly investigate the conditions, and if they find that the conditions do so materially differ, or do involve hazardous waste, and cause a decrease or increase in Design-Builder's Contract Price or Contract Time for any part of the Work, the OR will recommend that the Owner issue a Change Order under Section 9 of the Agreement. If it is determined that physical conditions at the site are not materially different from those indicated in Bid Documents or that no change in terms of the Contract Documents is justified, the OR will notify Design-Builder in writing, stating reasons the Design-Builder will not be entitled to an adjustment in the Contract Price or Contract Time. Such reasons may include any of the following:

14.14.4.1 Design-Builder knew of the existence of the conditions at the time Design-Builder submitted its proposal; or

14.14.4.2 Design-Builder should have known of the existence of the conditions as a result of having complied with the requirements of Contract Documents; or

14.14.4.3 The information or conditions claimed by Design-Builder to be latent or materially different consist of information, conclusions, opinions or deductions of the kind that precludes reliance upon; or

14.14.4.4 Design-Builder was required to give written notice of differing site conditions under the Contract Documents and failed to do so within the time required.

The Design-Builder will not be excused from the Contract Time to complete its Work and will proceed with all Work to be performed under the Agreement unless or until it is determined that Design-Builder is entitled to an adjustment under Section 9 of the Agreement. If the Design-Builder disagrees with the decision regarding an alleged Differing Site Condition, Design-Builder may pursue a claim under Section 14.12 of these Supplemental Conditions.

14.15 Design-Builder's License Requirements. Design-Builder and any approved subconsultants (for architectural design, engineering, construction project management services) or subcontractors shall hold such current and valid licenses as required by California Law, including the Department of Industrial Relations (DIR) contractor and subcontractor registration requirements articulated in part by Cal. Labor Code section 1725.5.

14.16 Examination and Audit of Records. Pursuant to Government Code Section 8546.7, Design-Builder shall retain all project-related records for a period of 3 years after final payment on this DBO Contract, which shall be subject to audit or inspection by the Owner or the State Auditor during this period.

14.17 Safety Requirements. The Design-Builder shall promptly and fully comply with and carry out, and shall without separate charge therefore to the Owner, enforce compliance with the safety and first aid requirements prescribed by applicable State and Federal laws and regulations, rules and orders and as may be necessary to ensure that all Construction Work shall be done in a

safe manner and that the safety and health of the employees, agents and the people of local communities is safeguarded. Compliance with the provisions of this Section by subcontractors shall be the responsibility of the Design-Builder. All installed, dismantled, and removed material, equipment and facilities, without separate charge therefore to Owner, shall fully conform with all applicable State and Federal safety laws, rules, regulations and orders and it shall be the Design-Builder's responsibility to furnish only such material, equipment and facilities.

14.18 Notice of Third-Party Claims. Pursuant to Public Contract Code section 9201, the Owner will provide Design-Builder with timely notification of the receipt of any third-party claim relating to the Agreement.

14.19 Assignment of Anti-Trust Actions. Pursuant to Public Contract Code section 7103.5 and Government Code sections 4554 and 4553, in entering into a public works contract or subcontract to supply goods, services, or materials pursuant to a public works contract, Design-Builder, its Design Professionals and Subcontractors offer and agree to assign to the Owner all rights, title, and interest in and to all causes of action it may have under Section 4 of the Clayton Act (15 U.S.C. section 15) or under the Cartwright Act (chapter 2 (commencing with section 16700) of part 2 of division 7 of the Business and Professions Code), arising from the purchase of goods, services, or materials pursuant to this Agreement or any Subcontract. This assignment will be made and become effective at the time the Owner makes final payment to the Design-Builder, without further acknowledgment by the parties.

14.20 Compliance with All Applicable Laws. Design-Builder shall comply with all the applicable requirements of federal, state and local laws, statutes and ordinances relative to the execution of the Work. In the event Design-Builder fails to comply with these requirements, Owner may stop any Work until such noncompliance is remedied. No part of the time lost due to any such cessation of the Work shall be made the subject of a claim for an extension of time or increase in the compensation.

Exhibit 2

BRIDGING DOCUMENTS

Exhibit 2A: San Bruno Mountain Park Conceptual Design
Exhibit 2B: Criteria Document
Exhibit 2C: Site Survey with Underground utilities
Exhibit 2D: Geotechnical Report
Exhibit 2E: Haz-Mat Report



EXHIBIT 2A



Option #1 - Pull Down Garage Door

Square Footage

- Building Footprint 642.53 sf
- Overall Footprint 2,055.33 sf

Option #2 - Sliding Barn Doors



EXHIBIT 2B

San Mateo County – Parks San Bruno Mountain Park Toilet Facility Replacement Project <u>Facility Design</u> Criteria

January 24, 2024

Following is the design criteria for the pre-qualified Design-Build Entities to use as the basis for their proposal during the Request for Proposal (RFP) phase:

Project Scope – Replacement of existing toilet room building. Abate and/or demolish the existing building according to the recommendations in the Hazardous Materials report and replace it with a new facility, including 4 ADA unisex restroom stalls.

General Requirements:

- 1. The building will have all stalls ADA-compliant.
- 2. Buildings shall be aesthetically pleasing but durable and low maintenance.
- 3. Buildings must comply with all fire-resistive construction requirements for Wildland Urban Interface (WUI).
- 4. Buildings to have a minimum of 4 to 12 sloped metal or concrete roofs with 4-foot overhangs around the entire building.
- 5. Roof overhangs must be at least 7 feet from tree trunks. The minimum height of the lowest point of the covered walkway surrounding the buildings is 8 feet and 8 inches. The roof overhang is to be cantilevered with no posts or columns.
- 6. Buildings require ample natural and mechanical ventilation due to high humidity levels.
- 7. Buildings are required to have daylighting in all rooms.
- 8. Concrete floors to be designed to mitigate cracking through expansion/contractor control, mix design, and curing methods.
- 9. Buildings shall have a no-limit, non-prorated, 2-year comprehensive warranty and specific product manufacturer warranties.
- 10. The Parks Department reserves the right to salvage items they want before DBE mobilization onto the site.

General Site Requirements:

- 1. Site grading around new construction is required to ensure smooth and flush transitions to existing grades, walks, and paths.
- 2. Provide positive drainage around buildings to prevent ponding water. DBE to include conceptual design solutions as part of their proposal.
- 3. Parks will perform removal of any trees. However, stumps and roots inside the building pad will be required to be removed by the DBE.
- 4. Buildings shall include a minimum of 6-foot wide reinforced concrete walkway around the entire building.
- 5. All exterior concrete needs to have a medium broom finish and meet ADA-required slopes and cross-slopes requirements.



San Mateo County – Parks Dept. – San Bruno Mountain Day Use Area Restroom Replacement and Picnic Area Upgrade Project – Planning Facility Design Criteria

- 6. All new walkways to be a minimum of 4" concrete reinforced with #4 rebar at 12" o.c. in both directions. Walkways to be located a minimum of 5-feet from tree trunks and be on 6" of ³/₄" aggregate above tree root zones. Transition to natural grade with topsoil.
- 7. ADA-compliant accessible parking and path-of-travel are included in the scope. The existing accessible parking stalls and signage need to be evaluated for current code compliance, and any necessary improvements are included in the scope.
- 8. All site work, grading, and paving must comply with the recommendations in the Geotechnical Report.

Site Utility Requirements:

- 1. Reuse existing site utilities for new building services.
- 2. Provide a new electrical pull box at the Point-of-Connection (POC).
- 3. A new water shut-off valve with a new grade box is to be provided at the Point of Connection (POC). Also, an additional valve and grade box are required at the stub ahead of building the shut-off valve for future connection.
- 4. New sewer cleanouts are required at Point-of-Connection (POC) in grade boxes. Provide separate cleanouts for each direction.
- 5. All new grade boxes are to be concrete with cast-in identification, 24" min. gravel sumps, bolt-down covers, rodent exclusion wire mesh, and 6" min. concrete collars.
- 6. All valves shall include operator tools.
- 7. Provide vehicle access and location within 20 feet of the electrical port for a trailer-mounted generator to be placed during power outages.

Floor Plan:

- 1. The building shall have the required ADA-compliant accessible accommodations.
- 2. Each toilet stall shall be an individual room with heavy-duty metal exterior doors and heavyduty door louvers. The toilet room doors shall face the prevailing wind or windward side of the building to protect the covered area.
- 3. Locate all plumbing on the common wall with the plumbing chase room behind.
- 4. The plumbing chase room is to have a 32" exterior door, interior lighting, an epoxy-coated floor sloped to the drain, and floor drain(s).
- 5. A custodial room with an exterior door shall accommodate an electrical panel. Note that the code required clear space in front of all panels. Also included is a mop sink with a mop rack and a heavy-duty metal storage cabinet 24" d x 48" w x 72" h anchored to the wall.
- 6. A separate storage room with double 36" doors to access from the center facing the covered area is required. The approximate size of the storage room is 6 feet wide and 20 feet long, with doors in the center. Lights, power receptacles, and epoxy-coated concrete floors are required.
- 7. A covered area is required to extend 20 feet from the storage room wall and the width of the building with access to the storage room, open on 3 sides, with support columns at each of the two open corners. It shall be located on the leeward side of the building to provide shelter from the prevailing winds. It is to have lighting and sloped concrete to adjacent grades.



San Mateo County – Parks Dept. – San Bruno Mountain Day Use Area Restroom Replacement and Picnic Area Upgrade Project – Planning Facility Design Criteria

Roof:

- 1. The roof must have a minimum of 4:12 pitch and be a standing seam metal roof or concrete with a smooth surface and grooves for the appearance of metal roofing. Roofing to have 30-year warranty.
- 2. The roofing manufacturer will provide the matching flashings and fascia metal, if not concrete.
- 3. Provide R-30 batt insulation over all rooms in the attic. Not required over covered area.
- 4. No rain gutters.

Exterior Requirements:

- 1. Exterior to have exposed decorative masonry exterior with aesthetically pleasing pattern and texture.
 - Provide alternative pricing for masonry wainscot with stone veneer and painted cementitious siding with the appearance of wood board-and-batt siding or other alternatives and options. Refer to the RFP for additional information on options and pricing.
- 2. Exterior doors and frames should be heavy-duty 16-gauge powder-coated metal with reinforcement and backing to prevent crushing when door hardware bolts are tightened.
- 3. Door louver vents are required at all stall doors. Louvers must be heavy-duty 16-gauge powder-coated metal louvers that match the doors and are fitted with insect screening and wire grid. No vision through louvers is required.
- 4. Provide baby changing station signage and ADA-compliant signage at all doors that identify room use. Fabricate the signs with high-quality, rigid, UV-tolerant materials, and mount them with vandal-resistant fasteners (provide two (2) bits to the owner for each type of vandal-resistant fastener used).
- 5. Provide two (2) 36" x 48" bulletin boards in a protective locking case, keyed alike and keyed to the existing keyway, with vents and a corkboard interior. Place the bulletin boards according to the concept design provided. Provide a doorstop to prevent them from opening beyond 90 degrees. Bulletin boards are to be fully welded and have a powder coat finish.
- Provide accommodations for an 80" flat panel display in a protective, heavy-duty, vandalresistant metal enclosure with Plexiglas cover and ventilation. Include a mounting bracket. Parks will provide the display. Provide power and conduit/junction box for future cable TV service.

Interior Requirements:

- 1. Each toilet stall shall include a wall-hung sink (lavatory), wall-hung toilet, electric hand dryer, and sap dispenser located over the sink.
- 2. Interior stalls are to have non-slip concrete floors coated with high-quality epoxy, with an integral coved base and sloped to drains. Provide a room that may be easily hosed out to clean. Slope the floors so that there is no ponding water.
- 3. The interior finishes of all stalls must be mold-and-mildew-resistant, durable, and easily maintained. Provide one gallon of touch-up paint for each color. Parks will select the paint colors.
- 4. All stainless-steel grab bars to be Type 304.
- 5. Provide heavy-duty tempered glass framed stainless steel 24" x 36" mirrors above sink.



4 Page

San Mateo County – Parks Dept. – San Bruno Mountain Day Use Area Restroom Replacement and Picnic Area Upgrade Project – Planning Facility Design Criteria

- 6. Provide a heavy-duty powder-coated coat hook on the interior face of each restroom's door. Finish to match the door.
- 7. Provide a wall-mounted stainless-steel trash receptacle.
- 8. Provide stainless steel, wall-mounted, manual liquid soap dispenser at each lavatory.
- 9. Provide ADA-compliant toilet paper roll holder to accommodate 2 rolls and have a round bar style.
- 10. Provide a "Koala Kare" baby changing station (or approved equal) in each room, including signage on the exterior to identify the room with a changing station.
- 11. Door Hardware at Restrooms:
 - A licensed specialty-trade professional will fabricate (and/or procure) and install the doors, frames, and hardware. (C-61/D28 or similar)
 - o Toilet Rooms:
 - Schlage Lever set Occupancy indicator with push-button lock function with key override.
 - Vandal lever break-away design.
 - Lock astragal.
 - Schlage Deadbolt with key operation only to lock for seasonal lock-down.
 - LCN long-arm door closures.
 - Heavy-duty 180-degree hinges. Provide stops/bumpers if unable to open 180 degrees due to exterior stone veneer if acceptable.
 - Locks to be keyed alike and keyway to tie into Schlage keying system.
 - Schlage L9050 with Vandlgard 06 lever set
 - ADA interior thumb latch 09-509 x L583-363
 - Occupied/Vacant outside trim indicator L283-722
 - Full size interchangeable core FSIC
 - B600 series deadbolt with B662P double cylinder deadbolt
 - Custodial Room, Plumbing Chase, and Gatehouse:
 - Schlage Lever set
 - Vandal lever break-away design.
 - Lock astragal.
 - Schlage Deadbolt with key operation only to lock for seasonal lock-down.
 - LCN long-arm door closures.
 - Heavy-duty 180-degree hinges. Provide stops/bumpers if unable to open 180 degrees due to exterior stone veneer if acceptable.
 - Locks to be keyed alike and keyway to tie into Schlage keying system.
 - Schlage L9050 with Vandlgard 06 lever set
 - ADA interior thumb latch 09-509 x L583-363
 - Full size interchangeable core FSIC
 - B600 series deadbolt with B662P double cylinder deadbolt
 - Storage Room:
 - Schlage Lever set
 - Vandal lever break-away design.
 - Lock astragal.
 - Schlage Deadbolt with key operation only to lock for seasonal lock-down.
 - LCN long-arm door closures with door sequencer.



San Mateo County – Parks Dept. – San Bruno Mountain Day Use Area Restroom Replacement and Picnic Area Upgrade Project – Planning Facility Design Criteria

- Heavy-duty 180-degree hinges. Provide stops/bumpers if unable to open 180 degrees due to exterior stone veneer if acceptable.
- Locks to be keyed alike and keyway to tie into Schlage keying system.
 - Storage room door to be openable by a standalone key and site master key.
- Schlage L9050 with VandIgard 06 lever set
- ADA interior thumb latch 09-509 x L583-363
- Full size interchangeable core FSIC
- B600 series deadbolt with B662P double cylinder deadbolt

Plumbing Requirements:

- 1. The existing 2" water and 4" sewer lines will be reused for service to the new building.
- The building will have a type 316 stainless-steel exterior-mounted high/low drinking fountain with a water bottle filler station (Elkay or approved equal). Provide a tool for vandal-resistant fasteners.
- 3. The building is to have an overall shut-off valve as described under "site utilities" above.
- 4. Each stall to have a separate shut-off valve located in the plumbing chase room.
- 5. Restroom sinks will receive hot and cold water.
- 6. Toilets must be ADA compliant, commercial grade, vitreous china, elongated bowl, wall-hung with heavy-duty carriers, and 1.28 g.p.f. with exposed top spud bowl American Standard or approved equal.
- 7. Toilet seats are to be commercial grade, heavy-duty, and open front. American Standard or approved equal.
- 8. Toilets are to have exposed manual flush valves with chrome piping and escutcheons. Royal Sloan or approved equal.
- 9. Lavatories (sinks) to be commercial grade, heavy-duty vitreous china, mounted on heavyduty carriers. American Standard or approved equal.
- 10. Lavatories to be provided with Chicago Faucets push button operation. At buildings with water heaters, provide mixing valve for temperate water.
- 11. Custodial closet to have a mop sink with Chicago Faucets bucket hook. Provide hot water.
- 12. Provide a 60-gallon high-efficiency water heater.
- 13. Plumbing chase room to have a hose bib located adjacent to the access door.
- 14. Sewer cleanouts are required at each fixture and the end of lateral lines. All must be easily accessible.
- 15. All above-grade water piping is to be copper.
- 16. All valves to be commercial grade, heavy-duty brass.
- 17. Floor drains in each restroom, plumbing chase, and janitor closet.

Existing Sewer Lift Station:

1. [REQUIREMENTS FOR NEW VAULT AND LIFT STATIONS PENDING REVIEW OF EXISTING UTILITIES.]

Mechanical Requirements:

- 1. Provide a centralized in-line exhaust fan system as follows:
 - Ventilation to be high volume at low velocity to minimize noise.



San Mateo County – Parks Dept. – San Bruno Mountain Day Use Area Restroom Replacement and Picnic Area Upgrade Project – Planning Facility Design Criteria

- Fans to be minimum of 14"
- Quiet operation: 0.6 @ 2.5 Sones
- High performance: Up to 200 CFM
- Energy efficient: 22 @ 64 W
- Fans to be controlled by central timeclock with override switch.
- Ventilation ductwork to be oversized for increase volume at lower velocity and constructed of all sheet metal, externally insulated, and have accessible balancing dampers.
- Designer to include ventilation calculations of air exchanges.
- Orient the exhaust vent on the building so that it does not face picnic areas or other occupied spaces.
- 2. No heating is required.

Electrical Requirements:

- 1. Replace the existing transformer with a 200 Amp device with new conduits and conductors.
- 2. The panel will have a main breaker and separate circuits for the existing lift station, power receptacles, interior and exterior lights, exhaust system, each of the 4 hand dryers, and water heater.
- 3. Remove existing generator manual transfer switch with port and reinstall in new building. Note that the location must be within 20-feet of the generator trailer pad.
- 4. All light fixtures are to be LED with UL label and readily available at major suppliers.
- 5. Provide a minimum of 50 foot-candles inside toilet room stalls and storage room; 30 footcandles in plumbing chase room.
- 6. Interior lights to be controlled with a photocell for daylight harvesting and dual sensor Watt Stopper controller or approved equal motion and infrared detection integrated into the light fixture. Detectors to be easy to adjust sensitivity.
- 7. Exterior recessed soffit mounted light fixtures are to provide a minimum of 30 foot-candles at all door entries.
- 8. Exterior lights are to be controlled with a roof-mounted central photo-cell detector that includes programming capabilities.
- 9. Add power receptacles (double duplex) in the storage room and in custodial room, and as necessary to comply with equipment service requirements. Locate adjacent to light switch at same height as light switch and locate a second at standard height.
- 10. No power receptacles inside restrooms.

Construction Requirements:

- 1. This is a Prevailing Wage Rate Project. Contractor to be DIR registered and comply with all applicable laws.
- 2. Contractor is required to provide and maintain all water pollution prevention and erosion control measures during construction in accordance with County's Best Management Practices (BMP). Please refer to Attachment "B".
- 3. Spill Prevention and Response:
 - Fluid spills shall not be hosed down. The Contractor shall use dry cleanup methods (absorbent materials, cat litter, and/or rags) whenever possible. If water must be used, the Contractor will be required to collect the water and spilled fluids and



San Mateo County – Parks Dept. – San Bruno Mountain Day Use Area Restroom Replacement and Picnic Area Upgrade Project – Planning Facility Design Criteria

dispose of it as hazardous waste. Spilled fluids shall not be allowed to soak into the ground or enter into any watercourse.

- Spilled dry materials shall be swept up immediately. The Contractor shall not wash down or bury any dry spills. Spills on dirt areas shall be removed by digging up and properly disposing of contaminated soil. The Contractor shall report significant spills to the County immediately.
- 4. Contractor is required to provide tree protection per the Parks Department's requirements included as Attachment "C". Prior to any work, Contractor must meet with County Arborist to review proposed tree removal and protection requirements.
- 5. Contractor will be responsible to provide any temporary facilities that may be required on site to implement the work. However, Park water and power can be used at no cost to contractor. Job site trailer is not required. However, DBE may provide if desired and if it is in an approved location. DBE to have generator(s) on site and on standby due to unreliable power from PG&E.
- 6. There is no reliable cellular service or public wi-fi in the Park. Contractor to provide communications to support construction.
- 7. Contractor and its sub-contractors are required to comply with all Park rules while in the Park. Any violation of Park Rules will result in the immediate removal of person(s) from Park and project.
- 8. Contractor to provide notifications at least 48-hours in advanced for any planned power or water shut-offs that may affect areas beyond the building area itself.
- 9. Clean soil and green waste may be disposed of on-site at locations determined by County. 10. Parks will pay all County permitting fees.

Attachments:

- A. Water Pollution Prevention BMP
- B. Tree Protection Requirements



San Mateo County – Parks Dept. – San Bruno Mountain Day Use Area Restroom Replacement and Picnic Area Upgrade Project - Planning Facility Design Criteria

Attachment A – Water Pollution Prevention BMP



Construction Best Management Practices (BMPs) Construction projects are required to implement the stormwater best management practices (BMP) on this page, as

they apply to your project, all year long.

MATER COUNTYWINE Water Pollution **Prevention Program** Clean Water, H

Materials & Waste Management



- Bern md rover stockpiles of sand, dart or other construction material with tarps when rain is forecast or if not actively being used within 14 days.
- Use (but don't overuse) reclaimed water for dust control.

rdous Materials

- Label all hazardous materials and hazardous wastes (such as bed all hardworks meterial and hardwork works (such is incide, pairs, dhardwork, wolvers, field, and and/weap) in container with city, coursy, size and fielders' regulatorial, interactions materials and wasses is ware regulatorial, more any work day or during west weakler or when min is fractual one manufactures? Projections insurverse for hardworks metalls and be careful one to use met than accursary. Do not regulate a provide the size when min is fracture whan 34 hours, mange for appropriate disposed of all hanachous warses.
- mit th

- Waste Management Cover waste disposal comments security with tarps at the end of every work day and during use weather.
- □ Check waste disposal containers frequently for leaks and to make sure they are not overfilled. Never hose down a dumpster on the
- ruction site. a or replace portable toilets, and inspect them frequently for and spills.
- must an spuc. D hypose of all wastes and debris properly. Recycle materials and wastes that can be recycled (such as aphalt, concrete, aggregate base materials, wood, gp bound pipe, etc.) D hypose of logic resident from pairs, thimsen, solvents, ghaes, and closuing finite as hazardous waste.

- struction Entrances and Perimeter
- Countrooke databases and it mattern De Inshibit and maintain efficiency permissier controls and tabilize sil-centraction entraces and exits to sufficiently control erosion and as dimmer discharges from una ward maching of 6th. Sweep or vacuum any strest tracking immediately and secure sediment source to prevent further tracking. Never bose down streets to clean up include:

Equipment Management & Earthmoving



Spill Control

- Maintenance and Parking Designate an area, first with appropriate BMPs, for valicit and equipment parking and storage Perform major maintenance, repair jobs, and vehicle and equipment washing off site. If refusing or valicite maintenance must be doos
- onsite, work in a bernaed area away from storm dra and over a drip pan or drop cloths big enough to co fluids. Recycle or dispose of fluids as hazardous w
- Buds. Factycle of dispose of Buds as hazardous w If vehicle or equipment cleaning must be done onsi clean with water only in a bermed area that will not allow rinke water to nu into gutters, stretts, storm dmins, or surface waters

Do not clean vehicle or equipment onsite using soap solvents, degressers, or iteam cleaning equipment.

- all Prevention and Control [Keep upill clearup materials (e.g., rags, absorbersts and cat latter) available at the construction site at all times.
- Impert validies and equipment frequently for and repair leaks promptly. Use drip pans to carch leaks until repairs are made.
 Clean up spills or leaks immediately and dispose of determined on the spin of the
- Cuento que spano en senso immedantery and onpose or cleanup materials properly. De not hose down surfaces where fluids have spilled. Use dry cleanup methods (absorbent materials, cat liner, and/or rags).
- Sweep up uplied dry materials immediately. Do not try to wash them away with water, or bury them.
 Clean up spills on drat raws by digging up and properly disposing of constrainments duel.
 Report significant spills immediately. You are require
- gnificant spills immediately. You are exposed report all significant releases of hazardous s, including oil. To report a spill: 1) Dial 911 the second second second second second second second second second response number, 2) Call the by law
 - social emergency response number, 2) or 's Office of Emergency Services Wa (800) 852-7550 (24 hours).



- Schedule grading and excavation work during dry workles.
 Stobilize is demaided areas, install and maintain temporary evosion control (size are evosion control folice or bonded flow matrix) unit vegetation is exhibited. *B Ranove* existing vegetation only when abuolitely necessary, and used or plant eveneration for evolution control on ideose by necession for eros
- ned. Prevent sediment from migrating offsite and protect storm drain inlets, gatters, ditches, and drainage courses by installi
- Keep excavated soil on site and transfer it to dump trucks on site, not in the streets.
- Contrainmatted South If any of the following conditions are observed, test for contamination and contact the Regional Water Quality Control Board:
- er odor. Abandoned underground tanks. Abandoned wells Buried barrels, debris, or traih.



Paving/Asphalt Work

- prevent materials dust have not cure from contacting stormwater runoff.
- Cover storm drain inlets and matholes when applying seal cost, tack cost, shary seal, fog seal, etc.
 Collict and netycle or appropriately dispose of excess abraine gravel or sand Do NOT sweep or weals it more gravel.
- Do not use water to wash down fresh asphalt concrete pavement. ing & Asphalt/Concrete Re
- out of the storm drain system. Showel, aboserb, or vacuum saw-car shurry and dispose of all waste as soon as you are disabled in one location or at the end of each work day (whichever is



Store concrete, grout, and morta from storm drains or waterways. pallets under cover to protect the rain, runoff, and wind.

Concrete, Grout & Mortar

- rain, ruisor, and whin. Wish our constret equipment tracks officie or in a designated washour area, where the water will flow into a temporary waste pitt, md in a maanner that will prevent leaching into the underlying solar or easo surrounding an Let concrete harden and dispose of as Whith out of

Landscaping

- landscaping materials by storing them under ups all year-round Stack bagged material on pallets and under cover.
 Discontinue analysis of the second sec DD application of any erodibl
 - material within 2 days before a ain event or during wet weather

Painting & Paint Removal



Painting Cleanup and Ra ovever clean bruches or rinse pai containers into a street, patter, s drain, or stream.

- For water-based paints, paint out brasi to the extent possible, and mice into a drain that goes to the sanitary sewer. Never pour paint down a storm drain. D For to
- ed paints, paint out br
- ent possible and clean with ent in a proper container. Fi hinners and solvents. Dispo licuids as hazardous waste.
- and dast fre
- O Ches
- Pairs chips and dust from non-bazardous day stripping and and blaring any be sweyt up or collected in plastic drop cloths and disposed of as trash. Chemical pairs stripping residue and chip and dust from marine paints or paints containing leafs. Imercury, or throwylin must be disposed of an harardous watte. ed paint res wal requires a st





In press of known or suspected contamination, call your local a determine whether the ground w be tested. Pumped groundwater to be collected and hauled off-to treatment and researchers.

Storm drain polluters may be liable for fines of up to \$10,000 per day!

















San Mateo County – Parks Dept. – San Bruno Mountain Day Use Area Restroom Replacement and Picnic Area Upgrade Project – Planning Facility Design Criteria

Attachment B – Tree Protection Requirements

SECTION 31 1311

TREE PROTECTION

PART 1 - GENERAL

1.1 DESCRIPTION OF WORK

- A. Extent: Furnish all labor, material, equipment, tools, and incidentals necessary for the installation of Tree Protection measures as specified in this Section.
 - 1. The work includes pruning of vegetation to be protected that are affected by temporary or permanent construction.

1.2 **DEFINITIONS**

- A. Vegetation: Shrubs, groundcovers, grass, and other plants.
- B. Plant Protection Zone: Area surrounding individual trees, groups of trees, and other vegetation to be temporarily protected during construction with fencing.
- C. Root zone: The root zone diameter of a tree is determined to be that area located out a distance 15 times the trunk diameter in all directions or the drip line, whichever is greater.
- D. Tree Protection Zone: temporary tree protection shall extend till the edge of the root zone, unless otherwise noted and shall be fenced. At no time shall the fencing be located closer than 3-feet away from the approved foundation, retaining wall, or grade cut, whichever provides the greater distance from the tree trunk.
- 1.3 Tree Pruning Schedule: Submit written schedule detailing scope and extent of pruning of trees to remain that interfere with or are affected by construction.
 - 1. Species and size of tree.
 - 2. Location on site plan. Include unique identifier for each.
 - 3. Reason for pruning.
 - 4. Description of pruning to be performed.
 - 5. Description of maintenance following pruning.


San Mateo County – Parks Dept. – San Bruno Mountain Day Use Area Restroom Replacement and Picnic Area Upgrade Project – Planning Facility Design Criteria

B. Reviews: Prior to proceeding with any tree removal or pruning, the Contractor shall notify the County 72 hours in advance for a review by the County Arborist.

PART 2 - PRODUCTS

2.1 MATERIALS

- A. Fencing: New or re-used chain-link, plywood, wood, or plastic, as approved by the county, minimum 4' high. Fence material shall be mounted on 2" diameter galvanized iron poles, maximum spacing 10' between poles.
- B. Warning Sign: Laminated card, rigid plastic or metal sheet, minimum 8.5"x11", with attachment holes, legibly printed with non-fading letters.
 - 1. Sign shall clearly state "WARNING Tree Protection Zone"
- C. Topsoil: The top layer of existing soil below the grass root zone, containing minerals and organic materials including humus. Depth of topsoil shall be taken to be 2-4 inches deep or as determined by the County at the time of construction.

PART 3 – EXECUTION

3.1 PRE-CONSTRUCTION

- A. Erosion and Sedimentation Control: Examine the site to verify that temporary erosionand sedimentation-control measures are in place. Verify that flows of water redirected from construction areas or generated by construction activity do not enter or cross plant protection zones.
- B. Existing Conditions: Review existing trees and vegetation indicated to remain on site, and document preconstruction conditions that might be misconstrued as damage caused by construction activities.
- C. Documentation: Prepare written report if necessary, endorsed by arborist, listing conditions detrimental to the protection of trees and vegetation.
- D. Preconstruction Meeting: Review methods and procedures related to temporary plant protection including, but not limited to:
 - 1. Construction schedule. Verify availability of materials, personnel, and equipment needed to make progress and avoid delays.
 - 2. Enforcing requirements for protection zones.



San Mateo County – Parks Dept. – San Bruno Mountain Day Use Area Restroom Replacement and Picnic Area Upgrade Project – Planning Facility Design Criteria

- 3. Field maintenance and quality control.
- E. Install all protection fencing for tree and plant protection zones prior to any site preparation, demolition, or grading work.
- F. Identification: Trees to be preserved shall be marked with a spot of paint. The marking is required to notify designated Inspectors that the subject tree or tree(s) are to be fenced at all times during construction.
- G. Verification: Verify in writing that all preconstruction conditions noted herein have been met and are in place. Submit verification to the Architect for approval prior to any site preparation, demolition, or grading work.

3.2 **PROTECTION ZONES**

- A. Tree and plant protection zones shall be maintained in a natural condition and not compacted. The following practices are prohibited within tree and plant protection zones:
 - 1. Storage of construction materials, debris, or excavated materials.
 - 2. Dumping of chemicals or garbage.
 - 3. Parking vehicles or equipment.
 - 4. Foot traffic.
 - 5. Erection of sheds or structures.
 - 6. Impoundment of water.
 - 7. Excavation or other digging unless otherwise indicated.
 - 8. Attachment of signs to or wrapping materials around trees or plants unless otherwise indicated.
- B. Prohibit heat sources, flames, ignition sources, and smoking within or near tree and plant protection zones and mulch.
- C. Signage: Install warning signs in visibly prominent locations in a manner approved by the Architect or Project Arborist in enough quantity so as to be visible from all visible sides.
- D. Fencing:
 - 1. Fencing shall be located at the edge of the tree protection zone, unless otherwise noted on the Drawings or as approved by the Architect or Project Arborist.
 - 2. Fencing shall be rigidly supported and maintained during all construction periods until Final Inspection.
 - 3. Do not remove protection-zone fencing, even temporarily, to allow deliveries or equipment access through the tree protection zone.



San Mateo County – Parks Dept. – San Bruno Mountain Day Use Area Restroom Replacement and Picnic Area Upgrade Project – Planning Facility Design Criteria

- 4. Temporary access is permitted subject to preapproval in writing by Project Arborist if a root buffer effective against soil compaction is constructed as directed by Project Arborist. Maintain root buffer so long as access is permitted.
- 5. Removal of fencing shall be approved by County Arborist.

3.3 EXCAVATION

- A. All cut, fill and/or foundations or walkways shall be located a minimum of 5-feet away from the outside edge of the trunk of all trees scheduled for preservation. However, the minimum distance permitted shall be 6-feet away from the outside edge of the trunk for all trees of a trunk diameter less the 2-feet. The diameter of a tree shall be measured at 4-feet and 6-inches above the surrounding grade (diameter at breast height, (DBH). Where Drawings conflict with this, immediately contact the County Project Manager.
- B. Utility and Drain lines: Shall be located outside the root zone of all trees scheduled for preservation. In cases where alternative routes are not available, utility conduit, pipe, wire and drain lines shall be tunneled under major roots. Major roots are determined to be those that exceed two (2) inches in diameter. In no case shall utility lines be permitted within six

(6) feet of the trunk. Immediately contact the Architect if the Drawings conflict with this.

- C. All approved construction work within the root zone of trees scheduled for preservation shall observe the following minimum tree protection:
 - 1. Hand trench at point or line of grade cuts closest to the trunk to expose major roots 2- inches in diameter or larger. In cases where rock or unusually dense soil prevents hand trenching, mechanical equipment may be approved by the Architect, provided that work inside the drip-line is closely supervised by the Arborist to prevent tearing or other damage to major roots.
- D. Redirect roots in backfill areas where possible. If encountering large, main lateral roots, expose roots beyond excavation limits as required to bend and redirect them without breaking. If encountered immediately adjacent to location of new construction and redirection is not practical, cut roots approximately 3-inches back from new construction and as required for root pruning.
- E. Do not allow exposed roots to dry out before placing permanent backfill. Provide temporary earth cover or pack with peat moss and wrap with burlap. Water and maintain in a moist condition. Temporarily support and protect roots from damage until they are permanently relocated and covered with soil.



3.4 PRUNING

San Mateo County – Parks Dept. – San Bruno Mountain Day Use Area Restroom Replacement and Picnic Area Upgrade Project – Planning Facility Design Criteria

- A. All tree pruning and tree damage repair shall only be performed by a qualified tree care specialist, or certified tree worker. Verify all pruning with County Arborist prior to start of pruning work.
 - 1. Trees shall be pruned to reduce hazards and develop a strong, safe framework of branches. Trees may also be pruned for 'crown cleaning' as defined by the International Society of Arboriculture Pruning Guidelines. Any pruning beyond these activities must be authorized by the County Arborist.

3.5 REPAIR & REPLACEMENT

- A. General: Repair or replace trees, shrubs, and other vegetation indicated to remain or be relocated that are damaged by construction operations, in a manner approved by Architect.
 - 1. Perform repairs within 24 hours.
 - 2. Replace vegetation that cannot be repaired and restored to full-growth pattern, as determined by Project Arborist.
 - 3. Replacement planting shall conform to Specification Section Landscape Planting, and soil amendments shall conform to Specification Section Soil Preparation.
- B Soil Aeration: Where directed by County Arborist, aerate surface soil compacted during construction. Aerate 10 feet beyond drip line and no closer than 36 inches to tree trunk. Drill 2-inch diameter holes a minimum of 12 inches deep at 24 inches o.c. Backfill holes with an equal mix of augured soil and sand.

3.6 REGRADING

- A. Lowering and raising grades: Where new finish grade is indicated below or above existing grade around trees, maintain existing grades within the Tree Protection Zone, and slope grade beyond the Tree Protection Zone.
- B. Lowering grade within Tree Protection Zone: slope grade away from trees as recommended by County Arborist.
- C. Minor Fill within Tree Protection Zone: Where existing grade is 2 inches or less below elevation of finish grade, fill with topsoil. Place topsoil in a single uncompacted layer and hand grade to required finish elevations.

END OF SECTION









AC	ASPHALT PAVED SURFACE
CB	CATCH BASIN

	BACKFLOW PREVENTER		FLAG F
BBQ	BARBECUE	IRR	IRRIGA
\bigcap	BICYCLE RACK	M	MILE P
\₩	BOLLARD	<u> </u>	SIGN -
	CATCH BASIN	o0	SIGN -
C	COMMUNICATIONS PULLBOX	CO	SEWER
DF	DRINKING FOUNTAIN	S	SEWER
	DROP INLET 1	£	VEG –
E	ELECTRIC PULLBOX	$\mathbb{X}_{\mathbb{K}}$	WATER
0	FENCE GATE POST	\wedge	CONTR

TAG #	TYPE	TRUNK DIA (IN)	DRIPLINE DIA (FT)
TAG 623	CLUSTER OF 2 OAK	12	15
TAG 624	OAK	4	5
TAG 625	CLUSTER OF 2 OAK	12	15
TAG 626	CLUSTER OF 4 OAK	36	15
TAG 627	OAK	12	15
TAG 628	CLUSTER OF 3 OAK	12	10
TAG 630	EVERGREEN	18	40
TAG 632	DECIDUOUS TREE	5	10
TAG 633	DECIDUOUS TREE	10	20
TAG 634	DECIDUOUS TREE	24	30
TAG 635	DECIDUOUS TREE	6	15
TAG 636	EVERGREEN	67	55
TAG 637	EVERGREEN	72	60
TAG 638	EVERGREEN	48	20
TAG 639	EUCALYPTUS	54	30
TAG 640	EUCALYPTUS	12	15
TAG 641	EVERGREEN	44	30
TAG 642	EVERGREEN	12	30
TAG 643	EUCALYPTUS	77	50
TAG 644	EVERGREEN	24	30
TAG 645	EVERGREEN	24	30
TAG 646	EVERGREEN	36	25
TAG 647	EVERGREEN	43	35
TAG 648	EUCALYPTUS	28	25
TAG 649	EUCALYPTUS	18	20
TAG 650	EUCALYPTUS	24	45
TAG 651	EUCALYPTUS	18	25
TAG 652	EUCALYPTUS	24	25
TAG 653	EUCALYPTUS	18	25
TAG 654	EVERGREEN	36	40
TAG 655	EUCALYPTUS	8	15
TAG 656	EUCALYPTUS	24	25
TAG 657	EUCALYPTUS	10	30
TAG 658	CLUSTER OF 3 12" EUCALYPTUS	36	50
TAG 659	EVERGREEN	12	30
TAG 660	EVERGREEN	12	25
TAG 661	CLUSTER OF 5 6" EVERGREEN	30	25
TAG 662	EVERGREEN	100	60
TAG 663	EVERGREEN	24	45
TAG 664	EUCALYPTUS	43	50
TAG 665	CLUSTER OF 2 9" EUCALYPTUS	18	30
TAG 666	EUCALYPTUS	12	35
TAG 667	EUCALYPTUS	32	30
TAG 668	EUCALYPTUS	48	40
TAG 670	EUCALYPTUS	24	30
TAG 671	EVERGREEN	30	40
TAG 672	EUCALYPTUS	26	25
TAG 673	EUCALYPTUS	48	40
IAG 674	EUCALYPTUS	10	20
IAG 675	EUCALYPTUS	24	30
IAG 676	EUCALYPTUS	24	20

TAG 677	EUCALYPTUS	20	25
TAG 678	EUCALYPTUS	16	20
TAG 679	EUCALYPTUS	22	30
TAG 680	EVERGREEN	12	20
TAG 681	EUCALYPTUS	30	25
TAG 682	EUCALYPTUS	40	40
TAG 683	EUCALYPTUS	24	30
TAG 684	EUCALYPTUS	36	25
TAG 685	EUCALYPTUS	43	25
TAG 686	EUCALYPTUS	7	10
TAG 687	EUCALYPTUS	17	25
TAG 688	DECIDUOUS TREE	12	25
TAG 689	EUCALYPTUS	30	25
TAG 690	EUCALYPTUS	20	25
TAG 691	EVERGREEN	10	15
TAG 692	EUCALYPTUS	25	30
TAG 693	EUCALYPTUS	12	15
TAG 694	EUCALYPTUS	24	25
TAG 695	EUCALYPTUS	36	35
IAG 696	EUCALYPTUS	18	25
IAG 697	EUCALYPTUS	24	30
IAG 698	EVERGREEN	77	50
TAG 699	EVERGREEN	48	45
TAG 700	EVERGREEN	35	50
TAG 701	EUCALYPTUS	5	15
TAG 702	EUCALYPTUS	30	25
TAG 703	OAK	10	20
TAG 704	OAK	10	20
TAG 705	OAK	12	25
TAG 706	OAK	10	20
TAG 707	OAK	8	25
TAG 708	CLUSTER OF 3 12" OAK	36	25
TAG 709	CLUSTER OF 2 14" OAK	28	30
TAG 710	OAK	6	12
TAG 711	OAK	6	12
IAG 712	OAK	18	30
TAG 713	OAK	36	45
IAG /14	UAK OAK	10	15
IAG /15	UAK	12	20
IAG /16	UAK	13	18
IAG /1/	UAK	13	20
1AG /18	UAK	20	30
1AG /19	UAK	22	30
TAC 701		17	
TAC 701		10	<u>۷</u> 10
TAC 707		1∠ ??	ں ۲0
TAG 723	ΟΔΚ	5	10
TAG 725	ΟΔΚ	9 9	12
TAG 726	FVFRGRFFN	11	15
TAG 727	ΟΔΚ	28	.35
TAG 728	ΟΑΚ	14	20
TAG 729	ΟΔΚ	1.3	15
TAG 7.30	0AK	12	15
	0/11	<u>، د</u>	

TAG 731	OAK	11	15
TAG 732	OAK	18	25
TAG 733	OAK	18	25
TAG 734	OAK	12	15
TAG 735	OAK	15	20
TAG 736	CLUSTER OF 2 10" OAK	20	25
TAG 737	OAK	12	20
TAG 738	OAK	12	20
TAG 739	OAK	12	20
TAG 740	OAK	10	15
TAG 741	OAK	10	15
TAG 742	OAK	24	30
TAG 743	OAK	16	20
TAG 744	OAK	20	25
TAG 745	CLUSTER 2 24" EVERGREEN	48	50
TAG 746	EVERGREEN	17	25
TAG 747	EVERGREEN	14	30
TAG 748	EVERGREEN	15	25
TAG 749	EVERGREEN	24	30
TAG 750	EVERGREEN	45	50
TAG 751	EVERGREEN	17	25
TAG 752	EVERGREEN	13	20
TAG 753	OAK	5	10
TAG 754	EVERGREEN	7	15
TAG 755	EUCALYPTUS	36	40
TAG 756	EVERGREEN	7	10
TAG 757	OAK	10	17
TAG 758	OAK	5	17
TAG 759	OAK	5	8
TAG 760	OAK	10	15
TAG 761	EUCALYPTUS	24	35
TAG 762	EUCALYPTUS	24	35
TAG 763	EUCALYPTUS	24	35
TAG 764	EUCALYPTUS	24	35
TAG 765	EUCALYPTUS	18	25
TAG 766	EUCALYPTUS	36	35
TAG 767	EUCALYPTUS	80	50
TAG 768	EUCALYPTUS	18	30
TAG 769	EUCALYPTUS	24	40
TAG 770	EUCALYPTUS	24	40
TAG 771	EUCALYPTUS	18	30
TAG 772	EUCALYPTUS	6	15
TAG 773	EUCALYPTUS	5	12
IAG 774	EUCALYPTUS	45	40
TAG 775	EUCALYPIUS		35
	UAK	14	20
	UAK	6	/
	UAK	6	10
	UAK OAK	6	/
	UAK	8	10
TAG621	UAK	4	/

	Surveying, Mapping		2300 Clayton Road, Sulte 1200	(Concord, CA 94520-2176	25) 682-6976 - www.towill.com		
	TOPOGRAPHIC SURVEY OF	SAN RRIIND MOUNTAIN DARK		555 GUAUALUPE CANYON PKWY, BRISRANE CA			
PREPARED FOR:		SAN MAIF()		COUNIY PARKS			
BY DATE SCALE $1" = 20'$	1 DAIE 4/8/2024	SURVEYED 7 -7	P.M. JTM	CALCULATED MJI	DRAWN N/A	CHECKED JTM	
REVISIONS							
ISSUE DATE		4/13/2024		JOB NUMBER	17301	TOC/T	
Sł	HEI OF	ET	-		3		

EXHIBIT 2D

Geotechnical Evaluation San Bruno Mountain Day Use Area New Restroom and Picnic Areas Saddle Loop Trail Daly City, California

Capital Program Management, Inc. 1851 Heritage Lane, Suite 210 | Sacramento, California 95815

April 19, 2024 | Project No. 404778001



Geotechnical | Environmental | Construction Inspection & Testing | Forensic Engineering & Expert Witness Geophysics | Engineering Geology | Laboratory Testing | Industrial Hygiene | Occupational Safety | Air Quality | GIS



Geotechnical & Environmental Sciences Consultants





Geotechnical Evaluation San Bruno Mountain Day Use Area New Restroom and Picnic Areas Saddle Loop Trail Daly City, California

Mr. Matthew Estes Project/Construction Manager Capital Program Managements, Inc. 1851 Heritage Lane, Suite 210 | Sacramento, California 95815

April 19, 2024 | Project No. 404778001

Caleb Hodnett Senior Project Engineer

CJH/MKW/rk

Marlene Watson

Marlene Watson, P.E., G.E. Principal Engineer



CONTENTS

1	INTRO	DUCTION	1	
2	SCOPE OF SERVICES 1			
3	SITE D	ESCRIPTION AND BACKGROUND	2	
4	PROJE	CT DESCRIPTION	2	
5	SUBSL	JRFACE EVALUATION AND LABORATORY TESTING	2	
6	GEOLO	OGIC AND SUBSURFACE CONDITIONS	3	
6.1	Region	al Geologic Setting	3	
6.2	Site Ge	ology	3	
6.3	Subsur	face Conditions	3	
6.4	Ground	lwater	4	
7	GEOLO	OGIC HAZARD AND GEOTECHNICAL	4	
7.1	Seismi	Hazards	4	
	7 1 1	Historical Seismicity	4	
	7.1.2	Faulting and Ground Surface Rupture	5	
	7.1.3	Liquefaction and Strain Softening	5	
7.2	Slope S	Stability	6	
7.3	Expans	sive Soil	6	
7.4	Static S	Settlement	6	
7.5	Excava	tion Characteristics	6	
7.6	Corros	ive/Deleterious Soil	7	
8	CONCL	LUSIONS	7	
9	RECO	MMENDATIONS	8	
9.1	Seismi	c Design Criteria	8	
9.2	Earthw	ork Recommendations	9	
	9.2.1	Pre-Construction Conference	9	
	9.2.2	Site Preparation	9	
	9.2.3	Observation and Removals	10	
	9.2.4	Material Recommendations	10	
	9.2.5	Subgrade Preparation	11	

	9.2.6	Fill Placement and Compaction	12
	9.2.7	Temporary Excavations and Shoring	13
	9.2.8	Utility Trenches	14
	9.2.9	Rainy Weather Considerations	15
9.3	Structu	ure Foundations	15
	9.3.1	Shallow Foundation	15
	9.3.2	Slabs-on-Grade	16
9.4	Pavem	ents and Flatwork	17
	9.4.1	Asphalt Pavement	18
	9.4.2	Concrete Pavement and Exterior Flatwork	19
9.5	Concre	ete	20
9.6	Surfac	e Drainage	21
9.7	Review	v of Construction Plans	21
9.8	Constr	ruction Observation and Testing	21
10	LIMITA	ATIONS	22
11	REFE	RENCES	24
TABLE	S		
1 – Calif	ornia Bui	Iding Code Seismic Design Criteria	9
2 – Reco	ommende	ed Material Requirements	11

3 – Subgrade Preparation Recommendations	12
4 – Fill Place Placement and Compaction Requirements	12
5 – OSHA Material Classifications and Allowable Slopes	13
6 – Asphalt Concrete Pavement Sections	18

FIGURES

- 1 Site Location
- 2 Exploration Locations
- 3 Fault Locations and Earthquake Epicenters
- 4 Regional Geology
- 5 Seismic Hazard Zones

APPENDICES

- A Boring Logs
- **B** Laboratory Testing

- C Corrosivity Testing
- D Utility Survey

1 INTRODUCTION

In accordance with your request and authorization, we have performed a geotechnical evaluation for the San Bruno Mountain Day Use Area project located on the north side of Guadalupe Canyon Parkway, east of Saddle Loop Trail in Daly City, California (Figure 1). The purpose of our study was to assess geotechnical conditions and provide recommendations for the design and construction of the proposed improvements.

2 SCOPE OF SERVICES

The services we performed included the following:

- Reviewed readily available geologic and seismic literature pertinent to the project area including geologic maps and reports, regional fault maps, seismic hazard maps, and aerial photographs.
- Performed site reconnaissance to observe the general site conditions and to mark the proposed locations for subsurface exploration.
- Coordinated with Underground Service Alert (USA) to locate and mark underground utilities in the vicinity of the subsurface exploration locations.
- Performed utility mapping within 100 feet radius of the existing restroom area. A map of the located utilities is included in Appendix D of this report.
- Obtained a boring permit from the San Mateo County Environmental Health Services Department.
- Conducted a subsurface exploration consisting of three (3) exploratory borings advanced up to 15 feet below existing grade. A representative of Ninyo & Moore logged the subsurface conditions exposed in the borings and collected bulk and relatively undisturbed soil samples for laboratory testing. The borings were backfilled with cement grout, in compliance with the San Mateo County Environmental Health Services Department.
- Performed laboratory testing of selected soil samples to evaluate the geotechnical properties of the subsurface materials including in-situ soil moisture content and density, particle size distribution, Atterberg limits, unconfined compressive strength, expansion index and soil corrosivity.
- Performed data compilation and engineering analysis of the information obtained from our background review, subsurface evaluation, and laboratory testing.
- Prepared this geotechnical report presenting our findings and conclusions regarding the subsurface conditions encountered at the project site, and our geotechnical recommendations for the design and construction of the proposed improvements.

3 SITE DESCRIPTION AND BACKGROUND

The project site is located on the north side of Guadalupe Canyon Parkway, east of Saddle Loop Trail in Daly City, California. (Figure 1) at about 37.697548 degrees north latitude and 122.433571 degrees west longitude. The site consists of a grassed lot with picnic tables along the west and east sides of the site. The existing restrooms are located in the southeast area of the site. The site is bounded to the east by Saddle Loop Trail, to the west and north by undeveloped land and to the south by the day use area parking lot. The project site is currently occupied by a landscaped park area. The site slopes downward to the south-southwest with elevations ranging from approximately 688 to 683 feet above mean sea level [MSL] (Google Earth, 2023).

4 **PROJECT DESCRIPTION**

Based upon the information provided to us, we understand that the project will consist of a new restroom facility to be constructed at the location of the existing restroom facility and upgrades to the adjacent picnic areas.

5 SUBSURFACE EVALUATION AND LABORATORY TESTING

Our field exploration included a site reconnaissance and subsurface exploration of the project site. The subsurface exploration was conducted on March 18, 2024, and consisted of three exploratory borings. The locations of the borings are presented on Figure 2. Prior to commencing the subsurface evaluation, USA was notified for field marking of existing utilities.

The borings were advanced to depths of up to approximately 15 feet below the existing grade with a truck-mounted drill rig equipped with solid-stem augers. A representative of Ninyo & Moore logged the subsurface conditions exposed in the borings and collected relatively undisturbed and bulk soil samples from the borings. The samples were then transported to our geotechnical laboratory for further visual characterization and testing. The borings were backfilled with grout after completion of drilling in accordance with the boring permit. Soil cuttings generated during drilling were thinly spread across the surface of the site in the vicinity of the borings. Descriptions of the subsurface materials encountered are presented in the following sections. Detailed logs of the borings and a description of sampling procedures is presented in Appendix A.

Laboratory testing of selected soil samples recovered from the borings included tests to evaluate in-situ soil moisture content and dry density, particle size distribution, Atterberg limits, unconfined compressive strength, expansion index, and soil corrosivity. The results of the in-place moisture content and dry density tests are shown at the corresponding sample depths on the boring logs in Appendix A. The results of the other geotechnical laboratory tests performed are presented in

Appendix B. Testing and a brief evaluation were conducted by CERCO Analytical on one sample of near-surface soil for corrosivity. The results of corrosivity testing are presented in Appendix C.

6 GEOLOGIC AND SUBSURFACE CONDITIONS

6.1 Regional Geologic Setting

The site is located near the northern end of the San Francisco Peninsula approximately 2.3 miles west of the western shore of San Francisco Bay in the Coast Ranges geomorphic province of California. The Coast Ranges are comprised of several mountain ranges and structural valleys formed by tectonic processes commonly found around the Circum-Pacific belt. Basement rocks have been sheared, faulted, metamorphosed, and uplifted, and are separated by thick blankets of Cretaceous and Cenozoic sediments that fill structural valleys and line continental margins. The San Francisco Bay Area has several ranges that trend northwest, parallel to major strike-slip faults such as the San Andreas, Hayward, and Calaveras (Figure 3). San Francisco Bay is a northwest trending structural depression that lies between the San Andreas Fault to the west and the Hayward fault to the east. Major tectonic activity associated with these and other faults within this regional tectonic framework consists primarily of right-lateral, strike-slip movement.

6.2 Site Geology

Published geologic maps (Bonilla, 1998) indicate that the site is underlain by Pleistocene slope debris and ravine fill. Bonilla describes this unit as stony, silty to sandy clay; locally silty to clayey sand or gravel; yellowish-orange to medium gray, unstratified or poorly stratified.

The slope debris and ravine fill are underlain by the Franciscan Assemblage, undivided. Bonilla describes this assemblage as mostly greywacke and shale, may be variably sheared. A map of regional geology is presented as Figure 4.

6.3 Subsurface Conditions

In general, the site is underlain by moist to wet, stiff sandy silt with some clay and gravel and moist. stiff silty clay with some sand and gravel, consistent with the regional mapping of slope debris and ravine fill.

Bedrock of the Franciscan Complex was encountered in each of the borings below the slope debris and ravine fill at depths ranging from approximately 2¼ feet to 8 feet below the existing ground surface extending to the maximum depth explored of approximately 15 1/2 feet. The bedrock encountered generally consisted of yellowish-brown and reddish-brown, moist, weathered sandstone.

Detailed descriptions of the materials encountered in our borings are presented on the boring logs in Appendix A.

6.4 Groundwater

Apparent perched groundwater was observed at a depth of approximately 8 feet in boring B-2 at the time of drilling. Groundwater was not observed in borings B-1 and B-3 during our subsurface exploration. However, groundwater may rise to a higher level due to the relatively low seepage rate in clay and the limited time for observation. Fluctuations in the groundwater level may occur due to seasonal precipitation, variations in topography or subsurface hydrogeologic conditions, or as a result of changes to nearby irrigation practices or groundwater pumping. In addition, seeps may be encountered at elevations above the historic high groundwater levels due to perched groundwater conditions, leaking pipes, preferential drainage, or other factors not evident at the time of our exploration.

7 GEOLOGIC HAZARDS AND GEOTECHNICAL CONSIDERATIONS

This study considered a number of potential geologic hazards and geotechnical considerations relevant to the proposed construction on the subject site, including seismic hazards, slope stability, expansive soils, static settlement, excavation characteristics, and soil corrosivity. These issues are discussed in the following subsections.

7.1 Seismic Hazards

The seismic hazards considered in this study include the potential for ground rupture due to faulting, seismic ground shaking, liquefaction, and dynamic settlement. These potential hazards are discussed in the following subsections.

7.1.1 Historical Seismicity

The site is located in a seismically active region. Figure 3 presents the location of the site relative to the epicenters of historic earthquakes with magnitudes of 5.5 or more from 1800 to 2000. Records of historic ground effects related to seismic activity (e.g. liquefaction, sand boils, lateral spreading, and ground cracking) compiled by Knudsen et al. (2000), indicate that ground effects related to historic seismic activity have not been reported for the site; however, a point about 1.2 miles southwest of the site is recorded as having historic landsliding.

7.1.2 Faulting and Ground Surface Rupture

California lies along the boundary between the North American and Pacific tectonic plates. Movement along the plate boundary can generate earthquakes and has created zones of deformation within the Earth's crust. These zones include various types of complex geologic structures and geomorphic features such as folds, faults, sag ponds, shutter ridges, linear valleys, and scarps. During moderate to large magnitude earthquakes, the ground can rupture along well-defined zones of deformation where faults intersect the Earth's surface.

The site is not located within an Alquist-Priolo Earthquake Fault Zone established by the California State Geologist (CGS, 2018) to delineate regions of potential ground surface rupture adjacent to active faults. As defined by the California Geological Survey (CGS), active faults are faults that have caused surface displacement within Holocene time, or within approximately the last 11,700 years (CGS, 2007 and 2018). The closest fault rupture hazard fault zone is the San Andreas fault zone, which is located about 3.7 miles southwest of the site (Figure 3). The nearest mapped fault to the site is the City College fault, which lies about one mile northeast of the project site. The City College fault is not considered active.

Based on our review of the referenced seismic hazard and geologic maps, known active faults are not mapped on the site, and the site is not located within a fault-rupture hazard zone. Therefore, the probability of damage from surface fault rupture is considered to be low.

The site is located near the San Andreas Fault and has a significant potential for seismic shaking due to its close proximity to this major tectonic boundary. The accumulation of stress along the fault's locked segments results in periodic release of energy through earthquakes. As a result, the site faces an increased risk of experiencing ground shaking during seismic events.

7.1.3 Liquefaction and Strain Softening

The strong vibratory motions generated by earthquakes can trigger a rapid loss of shear strength in saturated, loose, granular soils of low plasticity (liquefaction) or in wet, sensitive, cohesive soils (strain softening). Liquefaction and strain softening can result in a loss of foundation bearing capacity or lateral spreading of sloping or unconfined ground. Liquefaction can also generate sand boils leading to subsidence at the ground surface. Liquefaction (or strain softening) is generally not a concern at depths more than 50 feet below ground surface. The site is not located within a liquefaction hazard zone established by the State Geologist (CGS, 2000a). The seismic hazard zones for the site vicinity are presented on Figure 5. Regional studies of liquefaction susceptibility (Knudsen et al., 2000) indicate that the

liquefaction susceptibility at the site is very low. Liquefaction is not considered a design consideration.

7.2 Slope Stability

Regional landslide mapping in the seismic hazard zone report compiled by the California Geological Survey (CGS, 2021) depicts two landslides approximately 1.1 and 1.2 miles west of the subject site. The project site is not depicted as lying within a landslide hazard area. Therefore, landsliding and slope stability are not design considerations.

7.3 Expansive Soil

Some clay minerals undergo volume changes upon wetting or drying. Unsaturated soil containing those minerals will shrink/swell with the removal/addition of water. The heaving pressures associated with this expansion can damage structures and flatwork. Laboratory testing was performed on samples of the near-surface soil to evaluate the Atterberg limits and expansion index. The Atterberg limits test was performed in accordance with American Society of Testing and Materials (ASTM) Standard D 4318. The expansion index test was performed in accordance with ASTM Standard D 4829. The results of our Atterberg limits test indicates the near-surface soils exhibit a plasticity index (PI) of 21, which is indicative of a moderate expansion characteristic. The results of our expansion index test indicate that the expansion index of the sample tested was 50. This result is indicative of a low expansion characteristic.

7.4 Static Settlement

Based on documents provided, we understand that the proposed improvements do not include grading that will raise grades at the site. The sustained loads for the proposed structure are expected to be relatively light. We anticipate, therefore, that the total static settlement of shallow foundations due to sustained loads will be up to 1 inch with differential settlement of ½ inch across the building footprint, provided the recommendations presented in this report are followed.

7.5 Excavation Characteristics

We anticipate that the project will involve excavations for foundations and open pits or trenches for utilities of up to about 5 feet in depth. The near surface materials encountered during our subsurface exploration over this interval generally consisted of stiff sandy silt and silty clay and weathered sandstone. We anticipate that conventional earthmoving equipment in good working condition should be able to make the proposed excavations. Excavations may encounter obstructions consisting of cobbles, over-sized materials, or cemented bedrock that may require special handling or demolition equipment for removal. Gravel and cobbles or other obstructions may impede cutting neat excavations. Use of forms to construct footings should be anticipated.

Near-vertical temporary cuts in the near surface deposits up to 4 feet in depth should remain stable for a limited period of time. However, sloughing of the materials exposed on the excavation sidewall may occur, particularly if the excavation extends near the groundwater level, encounters granular soil, is exposed to water, or if the sidewall is disturbed during construction operations. Excavation subgrade may become unstable if exposed to wet conditions. Recommendations for excavation stabilization are presented. Excavated materials may also be wet and need to be dried out before reuse as fill.

7.6 Corrosive/Deleterious Soil

Corrosivity analysis was performed by CERCO Analytical, Inc. of Concord, California on samples of the near-surface soil. As reported by CERCO Analytical, the samples were determined to be "moderately corrosive" based on resistivity test results. CERCO Analytical's report (see Appendix C) included the following recommendation: "All buried iron, steel, cast iron, ductile iron, galvanized steel and dielectric coated steel or iron should be properly protected against corrosion depending upon the critical nature of the structure. All buried metallic pressure piping such as ductile iron firewater pipelines should be protected against corrosion."

The results for sulfate ion testing indicate that it is not sufficient to damage buried reinforced concrete structures or cement mortar-coated steel.

Please refer to the CERCO Analytical report included in Appendix C for more information regarding their test results and brief evaluation.

8 CONCLUSIONS

Based on our review of the referenced background data, site reconnaissance, subsurface evaluation, and laboratory testing, it is our opinion that the proposed construction is feasible from a geotechnical standpoint. Geotechnical considerations include the following:

- The subsurface conditions encountered during our exploration generally consisted of moist to wet stiff sandy silt with some clay and gravel and moist stiff silty clay with some clay and gravel underlain by weathered sandstone.
- The site could experience a relatively large degree of ground shaking during a significant earthquake event on a nearby fault.
- Apparent perched groundwater was observed at a depth of approximately 8 feet in boring B-2 at the time of drilling. Groundwater was not observed in borings B-1 and B-3 during our

subsurface exploration. Variations in the groundwater level across the site and over time should be anticipated.

- Static settlement should be tolerable for the proposed improvements provided that the proposed structures are supported on foundations that conform to our recommendations and fill placement to raise grades is less than 2 feet in height.
- Expansion Index and Atterberg limit testing indicates that the near-surface soil on site has a low to moderate expansion characteristic.
- Laboratory corrosion testing indicates that the near-surface site soils are considered moderately corrosive. A corrosion engineer should be consulted to provide specific guidance on protective measures to mitigate corrosion.

9 **RECOMMENDATIONS**

The following guidelines should be used in the preparation of the construction plans. The project plans and specifications should be reviewed by Ninyo & Moore prior to construction bidding to check for consistency with these recommendations.

9.1 Seismic Design Criteria

Table 1 presents the Risk-Targeted, Maximum Considered Earthquake (MCER) spectral response accelerations consistent with the 2022 California Building Code and corresponding site-adjusted and design level spectral response accelerations based on the USGS seismic design maps (SEAOC/OSHPD, 2023).

According to the shear-wave velocity in the upper 30 meters of surficial geology map provided by California Department of Conservation (DOC, 2020), the shear wave velocity in the upper 30 meters is about 730 m/s. According to ASCE 7-16, Table 20.3-1, the site is classified as D, defined as stiff soil.

Table 1 – California Building Code Seismic Design Criteria	
Seismic Design Parameter Evaluated for 37.697548° North Latitude, 122.433571°West	Value
Site Classification	D-Stiff Soil
Site Coefficient, F _a	1.2
Site Coefficient, Fv	
Mapped Spectral Response Acceleration at 0.2-second Period, S_s	1.823g
Mapped Spectral Response Acceleration at 1.0-second Period, S ₁	0.746g
Spectral Response Acceleration at 0.2-second Period Adjusted for Site Class, S_{M}	ıs 2.187g
Spectral Response Acceleration at 1.0-second Period Adjusted for Site Class, S	/ 1
Design Spectral Response Acceleration at 0.2-second Period, S_{DS}	1.458g
Design Spectral Response Acceleration at 1.0-second Period, S_{D1}	
Seismic Design Category for Risk Category I, II, or III ¹	D

Notes:

1. Based on simplified design procedure of Section 12.14 and Table 11.6-1 of ASCE 7-16.

9.2 Earthwork Recommendations

Earthwork should be performed in accordance with the requirements of applicable governing agencies and the recommendations presented below. The geotechnical consultant should observe earthwork operations. Evaluations performed by the geotechnical consultant during the course of operations may result in new recommendations, which could supersede the recommendations in this section.

9.2.1 **Pre-Construction Conference**

We recommend that a pre-construction conference be held to discuss the grading recommendations presented in the report. Representatives of the parks department, the architect, the engineer, Ninyo & Moore, and the contractor should be in attendance to discuss project schedule and earthwork requirements.

9.2.2 Site Preparation

Site preparation should begin with the removal of vegetation, utility lines, debris and other deleterious materials from areas to be graded. Tree stumps and roots should be removed to such a depth that organic material is generally not present. Clearing and grubbing should extend to the outside of the proposed excavation and fill areas. Rubble and excavated materials that do not meet criteria for use as fill should be disposed of in an appropriate landfill. Existing utilities in the work area should be relocated away from the proposed structures. Existing utilities to be abandoned should be removed, crushed in place, or backfilled with grout.

Excavations resulting from removal of existing building foundations, buried utilities, tree stumps, or obstructions should be backfilled with compacted fill in accordance with the recommendations presented in the following sections.

9.2.3 Observation and Removals

Prior to placement of fill, or the placement of forms or reinforcement for foundations, the client should request an evaluation of the exposed subgrade by Ninyo & Moore. Materials that are considered unsuitable shall be excavated under the observation of Ninyo & Moore in accordance with the recommendations provided in this section or supplemental recommendations by the geotechnical engineer. Some moisture conditioning of soils may be required for the near-surface wet soils encountered in boring B-2.

Unsuitable materials include, but may not be limited to dry, loose, soft, wet, expansive, organic, or compressible natural soil, and undocumented or otherwise deleterious fill materials. Unsuitable materials should be removed from trench bottoms and below bearing surfaces to a depth at which suitable foundation subgrade, as evaluated in the field by Ninyo & Moore, is exposed.

9.2.4 Material Recommendations

Materials used during earthwork, grading, and paving operations should comply with the requirements listed in Table 2. Materials should be evaluated by the geotechnical engineer for suitability prior to use. The contractor should notify the geotechnical consultant 72 hours prior to import of materials or use of on-site materials to permit time for sampling, testing, and evaluation of the proposed materials. On-site materials may need to be dried out before re-use as fill. The contractor should be responsible for the uniformity of import material brought to the site.

Table 2 – Recommended Material Requirements				
Material and Use	Source	Requirements ^{1,2}		
General Fill: Import - for uses not otherwise specified	Import	Close-graded with 35 percent or more passing No. 4 sieve and either: Expansion Index of 50 or less, Plasticity Index of 12 or less, or less than 10 percent, by dry weight, passing No. 200 sieve		
	On-site borrow	No additional requirements ¹		
Aggregate Subbase	Import	Class II; CSS ⁴ Section 25-1.02		
Aggregate Base	Import	Class II; CSS ⁴ Section 26-1.02		
Controlled Low Strength Material (CLSM)	Import	CSS ⁴ Section 19-3.02F		
Pipe/Conduit Bedding and Pipe Zone Material -material below pipe invert to 12 inches above pipe	Import	90 to 100 percent (by mass) should pass No. 4 sieve, and 5 percent or less should pass No. 200 sieve		
Trench Backfill - above bedding material	Import or on-site borrow	As per general fill and excluding rock/lumps retained on 4-inch sieve or 2-inch sieve in top 12 inches		

Notes:

1 In general, fill should be free of rocks or lumps in excess of 4 inches in diameter, trash, debris, roots, vegetation or other deleterious material.

2 In general, import fill should be tested or documented to be non-corrosive³ and free from hazardous materials in concentrations above levels of concern.

3 Non-corrosive as defined by the Caltrans Corrosion Guidelines (Caltrans, 2018a).

4 CSS is California Standard Specifications (Caltrans, 2018b).

9.2.5 Subgrade Preparation

Subgrade in trenches and below flatwork or fill should be prepared as per the recommendations in Table 3. Prepared subgrade should be maintained in a moist (but not saturated) condition by the periodic sprinkling of water prior to placement of additional overlying fill. Subgrade that has been permitted to dry out and loosen or develop desiccation cracking, should be scarified, moisture-conditioned, and recompacted as per the requirements above.

Table 3 – Subgrade	Preparation Recommendations
Subgrade Location	Source
Below fill, slabs, and flatwork	 After clearing per Section 9.2.2, check for unsuitable materials as per Section 9.2.3. Scarify top 8 inches then moisture-condition and compact as per Section 9.2.6. Keep in moist condition by sprinkling water.
Utility Trenches	 After clearing per Section 9.2.2, check for unsuitable materials as per Section 9.2.3. Remove or compact loose/soft material.

9.2.6 Fill Placement and Compaction

Fill and backfill should be compacted in horizontal lifts in conformance with the recommendations presented in Table 4. The allowable uncompacted thickness of each lift of fill depends on the type of compaction equipment utilized, but generally should not exceed 8 inches in loose thickness.

Table 4 – Fill Placement and Compaction Recommendations						
Fill Type	Location	Compacted Density ¹	Moisture Content ²			
Subgrade	Flatwork, or fill and in locations not already specified	90 percent	+ 2 percent or above			
General Fill	In locations not already specified	90 percent	+ 2 percent or above			
Bedding and Pipe Zone Fill	Material below invert to 12 inches above pipe or conduit	90 percent	Near Optimum			
Trench Backfill	Top 18 inches below finish subgrade for areas subject to vehicular loading	95 percent	+ 2 percent or above			
	In locations not already specified	90 percent	+ 2 percent or above			
Aggregate Base	Below hardscape	95 percent	Near Optimum			

Notes:

1 Expressed as percent relative compaction or ratio of field density to reference density (typically on a dry density basis for soil and aggregate). The reference density of soil and aggregate should be evaluated by ASTM D 1557.

2 Target moisture content at compaction relative to the optimum as evaluated by ASTM D 1557.

Compacted fill should be maintained in a moist (but not saturated) condition by the periodic sprinkling of water prior to placement of additional overlying fill. Fill that has been permitted to dry out and loosen or develop desiccation cracking, should be scarified, moisture-conditioned, and recompacted as per the requirements above.

9.2.7 Temporary Excavations and Shoring

Trench excavations shall be stabilized in accordance with the Excavation Rules and Regulations (29 Code of Federal Regulations [CFR], Part 1926) stipulated by the Occupational Safety and Health Administration (OSHA). Stabilization shall consist of shoring sidewalls or laying slopes back.

Dewatering pits or sumps should be used to depress the groundwater level (if encountered) below the bottom of the excavation. Table 5 lists the OSHA material type classifications and corresponding allowable temporary slope layback inclinations for soil deposits that may be encountered on site. Alternatively, an internally-braced shoring system or trench shield conforming to the OSHA Excavation Rules and Regulations (29 CFR, Part 1926) may be used to stabilize excavation sidewalls during construction. The lateral earth pressures listed in Table 5 may be used to design or select the internally-braced shoring system or trench shield. The recommendations listed in this table are based upon the limited subsurface data provided by our exploratory borings and reflect the influence of the environmental conditions that existed at the time of our exploration. Excavation stability, material classifications, allowable slopes, and shoring pressures should be re-evaluated and revised, as-needed, during construction. Excavations, shoring systems and the surrounding areas should be evaluated daily by a competent person for indications of possible instability or collapse.

Table 5 – OSHA Materia	5 – OSHA Material Classifications and Allowable Slopes					
Formation	OSHA Classification	Allowable Temporary Slope ^{1,2,3}	Lateral Earth Pressure on Shoring⁴ (psf)			
Slope debris and ravine fill (above groundwater)	Туре С	1½ h:1v (34°)	80×D + 72			

Notes:

1 Allowable slope for excavations less than 20 feet deep. Excavation sidewalls in cohesive soil may be benched to meet the allowable slope criteria (measured from the bottom edge of the excavation). The allowable bench height is 4 feet. The bench at the bottom of the excavation may protrude above the allowable slope criteria.

2 In layered soil, layers shall not be sloped steeper than the layer below.

3 Temporary excavations less than 5 feet deep may be made with vertical side slopes and remain unshored if judged to be stable by a competent person (29 CFR, Part 1926.650).

4 'D' is depth of excavation for excavations up to 20 feet deep. Includes a surface surcharge equivalent to two feet of soil.

The shoring system should be designed or selected by a suitably qualified individual or specialty subcontractor. The shoring parameters presented in this report are preliminary design criteria, and the designer should evaluate the adequacy of these parameters and make appropriate modifications for their design. We recommend that the contractor take

appropriate measures to protect workers. OSHA requirements pertaining to worker safety should be observed.

Excavations made in close proximity to existing structures may undermine the foundation of those structures and/or cause soil movement related distress to the existing structures. Stabilization techniques for excavations in close proximity to existing structures will need to account for the additional loads imposed on the shoring system and appropriate setback distances for temporary slopes. The geotechnical engineer should be consulted for additional recommendations if the proposed excavations cross below a plane extending down and away from the foundation bearing surfaces of the adjacent structure at an angle of 2:1 (horizontal to vertical).

The excavation bottoms may encounter wet, loose material, which may be subject to pumping under heavy equipment loads. The contractor should be prepared to stabilize the bottom of the excavations. In general, unstable bottom conditions may be mitigated by using a stabilizing geogrid, overexcavating the excavation bottom to suitable depths and replacing with compacted fill, or other suitable method. Additionally, aeration of wet soils should be anticipated.

9.2.8 Utility Trenches

Trenches constructed for the installation of underground utilities should be stabilized in accordance with our recommendations in Section 9.2.7. Utility trenches should be backfilled with materials that conform to our recommendations in Section 9.2.4. Trench backfill, bedding, and pipe zone fill should be compacted in accordance with Section 9.2.6 of this report. Bedding and pipe zone fill should be shoveled under pipe haunches and compacted by manual or mechanical, hand-held tampers. Trench backfill should be compacted by mechanical means. Densification of trench backfill by flooding or jetting should not be permitted.

To reduce potential for moisture intrusion into the building envelope, we recommend plugging utility trenches at locations where the trench excavations cross under a building perimeter. The trench plug should be constructed of a compacted, fine-grained, cohesive soil that fills the cross-sectional area of the trench for a distance equivalent to the depth of the excavation. Alternatively, the plug may be constructed of concrete or CLSM.

9.2.9 Rainy Weather Considerations

Earthwork and foundation construction should be performed during the period between approximately April 15 and October 15 to avoid the rainy season. In the event that grading is performed during the rainy season, the plans for the project should be supplemented to include a stormwater management plan prepared in accordance with the requirements of the relevant agency having jurisdiction. The plan should include details of measures to protect the subject property and adjoining off-site properties from damage by erosion, flooding or the deposition of mud, debris, or construction-related pollutants, which may originate from the site or result from the grading operation. The protective measures should be installed by the commencement of grading, or prior to the start of the rainy season. The protective measures should be maintained in good working order unless the project drainage system is installed by that date and approval has been granted by the building official to remove the temporary devices.

In addition, construction activities performed during rainy weather may impact the stability of excavation subgrade and exposed ground. Temporary swales should be constructed to divert surface runoff away from excavations and slopes. Steep temporary slopes should be covered with plastic sheeting during significant rains. The geotechnical consultant should be consulted for recommendations to stabilize the site as-needed.

9.3 Structure Foundations

Based on the results of our field and laboratory evaluations, it is our opinion that the proposed restroom can be supported on spread footings or mat foundation. Foundations should be designed in accordance with structural considerations and the following recommendations. In addition, requirements of the appropriate governing jurisdictions and applicable building codes should be considered in design of the structures.

9.3.1 Footing Foundation

The proposed structure may be designed for a net allowable bearing capacity of 2,000 pounds per square foot (psf) presuming that the foundation is constructed per the recommendations in this report. This allowable bearing capacity includes a factor of safety of 2 and may be increased by one-third when considering wind or seismic loading combinations. Shallow foundations should have a minimum width of 24 inches and a minimum depth of the foundation base below adjacent grade of 24 inches.

The proposed structure should be designed for a total settlement of 1-inch due to sustained loads and a differential settlement of $\frac{1}{2}$ -inch across the structure, estimated to be up to a 40-foot span.

The footing and slab reinforcement should be designed and detailed by the structural engineer based on the anticipated loading and usage. Recommendations for concrete and concrete cover over reinforcing steel are presented in Section 9.5. Masonry briquettes or plastic chairs should be used to aid in the correct placement of slab reinforcement.

A friction coefficient of 0.35 and a lateral bearing pressure of 350 psf per foot of embedment depth may be used to evaluate foundation resistance to lateral loads. The lateral bearing pressure should be neglected to a depth of 12 inches where the ground adjacent to the foundation is not covered by a slab or pavement. The lateral bearing pressure may be increased by one-third when considering loads of short duration such as wind or seismic forces.

9.3.2 Mat Foundation

The proposed restroom may be founded on a mat foundation supported on native material or newly placed fill, using an allowable bearing capacity of 2,000 pounds psf. This allowable bearing capacity may be increased by one-third when considering loads of a short duration such as wind or seismic forces. Thickness and reinforcement of the mat foundation should be in accordance with the recommendations of a structural engineer.

Mat foundations typically experience some deflection due to loads placed on the mat and the reaction of the soils underlying the mat. A design coefficient of subgrade reaction, K_{v1} , of 110 pounds per cubic inch (pci) may be used for evaluating such deflections at the building site. This value is based on a unit square foot area and should be adjusted for the planned mat size. The coefficient of subgrade reaction K_b for a mat of a specific width, may be evaluated using the following equation, where b is the width of the foundation.

$$K_b = K_{v1}[(b+1)/2b]^2$$

9.3.3 Slabs-on-Grade

Building floor slabs should be designed by the project structural engineer based on the anticipated loading conditions. Building floor slabs should be underlain by compacted

engineered soil prepared in accordance with the recommendations presented in this report. We recommend that slabs be, at a minimum, 6 inches thick and reinforced with No. 3 steel reinforcing bars placed 24 inches on-center (each way) placed near the mid-height of the slab. The placement of the reinforcement in the slab is vital for satisfactory performance. The floor slab and foundations should be tied together by extending the slab reinforcement into the foundations. The slab should be underlain by a 4-inch-thick capillary break (consisting of either sand or crushed rock) overlain by a polyethylene vapor retarder (with a thickness of 15 mils or more). The steel reinforcements for the floor slab shall be placed on the vapor retarder using chairs, as appropriate. The vapor retarder is recommended in areas where moisture-sensitive floor coverings are anticipated. Soils underlying the slabs should be moisture-conditioned and compacted in accordance with the recommendations presented in this report prior to concrete placement. Joints should be constructed at intervals designed by the structural engineer to help reduce random cracking of the slab.

9.4 Pavements and Flatwork

Recommendations for pavement and exterior flatwork are presented in the following sections. A design R-value of 10 was selected considering the subsurface soil conditions encountered in the exploratory borings. The pavement subgrade should be evaluated by the geotechnical engineer during grading to check the finish subgrade for consistency with the assumed condition. Laboratory testing of R-value may be needed, based on these observations, with subsequent revision to the pavement sections.

Finish subgrade for pavement and flatwork should be scarified and moisture conditioned as needed to achieve a moisture content about 2 percentage points above the optimum, before compaction, by mechanical means, to 90 percent or more of ASTM D1557 reference density on a dry density basis for flatwork that is not subject to vehicular loading and 95 percent of the reference density for pavement and flatwork that is subject to vehicular loading. Finish subgrade for pavement should be proof-rolled with a loaded water truck under the observation of the geotechnical engineer after compaction. Yielding subgrade should be mitigated in accordance with the field recommendations of the engineer. Aggregate base and subbase for pavement and flatwork should be placed and compacted in lifts by vibrating plates, smooth drum rollers, or mechanical tampers, to 95 percent of the ASTM D1557 reference density on a dry density basis. The allowable lift thickness is influenced by the type of compaction equipment utilized but generally should not exceed 8 inches in loose thickness. Compacted subgrade and aggregate base or subbase should be maintained in a moist condition by sprinkling water. Compacted subgrade or aggregate base/subbase that has dried out and loosened or developed desiccation

cracking should be scarified and moisture-conditioned, as needed, and recompacted as per the requirements above before the material is covered with asphalt, concrete, or aggregate base/subbase.

9.4.1 Asphalt Pavement

Ninyo & Moore conducted an analysis to evaluate appropriate asphalt pavement structural sections following the methodology presented in the Highway Design Manual (Caltrans, 2020). Alternative sections were evaluated. The pavement sections were designed for a 20-year service life presuming that periodic maintenance, including crack sealing and resurfacing will be performed during the service life of the pavement. Premature deterioration may occur without periodic maintenance. Our recommendations for the pavement sections are presented in Table 6.

Paving operations and base preparation should be observed and tested by Ninyo & Moore. If subgrade enhancement geotextiles are utilized, they should be rolled out flat and tight, without folds or wrinkles, over prepared subgrade in the direction of travel. The geotextile should be pinned to the subgrade with nails and washers or u-shaped sod staples. Adjacent rolls should overlap 12 inches or more. Abutting rolls should overlap in the direction of fill placement to reduce the potential for peeling of the geotextile during fill placement. Aggregate base fill should be pushed over the geotextile into position and compacted. To reduce the potential for displacement of the geotextile or deterioration of the subgrade, construction equipment should not operate on the geotextile with less than 6 inches of aggregate base cover.

Table 6 – Asphalt Concrete Pavement Sections						
Traffic Index	R-value	Alternative 1	Alternative 2			
3	10	4 inches AC	2½ inches AC 4½ inches AB			
6	10	9 inches AC	4 inches AC 12 inches AB			
9	10	14 inches AC	5½ inches AC 19 inches AB			

Notes:

AC is Type A, Dense-Graded Hot Mix Asphalt complying with Caltrans Standard Specification 39-2 (2022).
 AB is Class II Aggregate Base complying with Caltrans Standard Specification 26-1.02 (2022).

Aggregate base for pavement should be placed in lifts of no more than 8 inches in loose thickness and compacted per Section 9.2.6. Asphalt concrete should be placed and compacted in accordance with Caltrans Standard Specification and Construction Manual; asphalt concrete should be compacted to between 92 and 96 percent of the theoretical maximum specific gravity and density (Rice gravity – ASTM D 2041) of the material.. Pavements should be sloped so that runoff is diverted to an appropriate collector (concrete gutter, swale, or area drain) to reduce the potential for ponding of water on the pavement. Concentration of runoff over asphalt pavement should be discouraged.

9.4.2 Concrete Pavement and Exterior Flatwork

Concrete thickness should be 6 inches or more for flatwork subject to vehicular traffic up to periodic garbage trucks and emergency vehicles. Flatwork subject to impact from unloading of dumpsters should be 8 inches thick or more. Pedestrian sidewalks, walkways, and other flatwork should be 4 inches thick or more. The minimum thickness of the aggregate base underlying concrete pavement and exterior flatwork should be 6 inches.

Appropriate jointing of concrete flatwork can encourage cracks to form at joints, reducing the potential for crack development between joints. Joints should be laid out in a square pattern at consistent intervals. Contraction and construction joints should be detailed and constructed in accordance with the guidelines of ACI Committee 302. The ratio of lateral spacing between contraction joints to the nominal thickness of the slab should not exceed 24 for jointed plain concrete. Contraction joints formed by premolded inserts, grooving plastic concrete, or saw-cutting at initial hardening, should extend to a depth equivalent to 25 percent of the slab thickness and 1 inch or more for thin slabs. The joint location and layout of new or reconstructed flatwork abutting existing flatwork should be consistent with joint location/layout of the existing flatwork.

Flatwork may be reinforced with distributed steel to reduce the potential for differential slab movement where cracking occurs. The distributed reinforcing steel should be terminated about 3 inches from contraction joints and should consist of No. 3 deformed bars at 18 inches on center, both ways, or with 6x6-D4/D4 welded wire fabric supplied as sheets (not rolls). Slabs reinforced with distributed steel should be 6 inches thick (or more) for No. 3 bar reinforcement and 5 inches thick (or more) for 6x6-D4/D4 reinforcement to provide adequate concrete cover for the steel. To reduce the potential for differential slab movement will be balanced by a reduction in the functionality of the contraction joint to encourage crack formation at joints. Flatwork subject to impact from unloading of dumpsters should be

reinforced with No. 4 deformed bars at 12 inches on center, both ways extending through contraction joints, if present. Masonry briquettes or plastic chairs should be used to maintain the position of the reinforcement in the upper half of the slab with 1½ inches of cover over the steel and 3 inches of cover under the steel. Root barriers adjacent to trees may be considered to reduce the potential for pavement heave from root growth.

After demolition, clearing, and rough grading, finish subgrade for flatwork should be scarified and moisture conditioned as needed to achieve a moisture content about 2 percentage points above the optimum, before compaction, by mechanical means, to 90 percent or more of ASTM D1557 reference density on a dry density basis for flatwork that is not subject to vehicular loading and 95 percent of the reference density for flatwork that is subject to vehicular loading. Aggregate base should conform to the criteria for Class 2 aggregate base in Section 26-1.02 of the California Standard Specifications (Caltrans, 2022) and should be placed and compacted in lifts by vibrating plates, smooth drum rollers, or mechanical tampers, to 95 percent of the ASTM D1557 reference density on a dry density basis after moisture conditioning to near the optimum moisture content. The allowable lift thickness is influenced by the type of compaction equipment utilized but generally should not exceed 8 inches in loose thickness. Compacted subgrade and aggregate base should be maintained in a moist condition by sprinkling water. Compacted subgrade or aggregate base that has dried out and loosened or developed desiccation cracking should be scarified and moisture-conditioned, as needed, and recompacted as per the requirements above before the material is covered with concrete or aggregate base.

9.5 Concrete

Due to the variability in the on-site soil, we recommend that Type II/V or Type V cement be used for concrete structures in contact with soil. In addition, we recommend a water-to-cement ratio of no more than 0.45. A 3-inch thick, or thicker, concrete cover should be maintained over reinforcing steel where concrete is in contact with soil in accordance with Section 20.6 of ACI Concrete Institute (ACI) Committee 318 (ACI, 2016).

To reduce the potential for shrinkage cracks in the concrete during curing, concrete for slabs and flatwork should not contain large quantities of water or accelerating admixtures containing calcium chloride. Higher compressive strengths may be achieved by using larger aggregates in lieu of increasing the cement content and corresponding water demand. Additional workability, if desired, may be obtained by including water-reducing or air-entraining admixtures. Concrete should be placed in accordance with the appropriate guidance in the ACI Manual of Concrete Practice (MCP) and project specifications. Particular attention should be given to curing techniques and

curing duration. Slabs that do not receive adequate curing have a more pronounced tendency to develop random shrinkage cracks and other defects.

9.6 Surface Drainage

Positive surface drainage should be provided to divert surface water and roof runoff away from foundations. Downspouts should be connected to a closed drainage system to discharge at a suitable location 10 feet or more away from the foundations. Runoff should be diverted by the use of swales or pipes into a collective drainage system. Drainage on the site should be provided so that water is not permitted to pond. A gradient of 2 percent or steeper on impervious surfaces and 5 percent or steeper on pervious surfaces should be maintained and drainage patterns should be established to divert and remove water from the site to appropriate outlets. Care should be taken by the contractor during grading to preserve any berms, drainage terraces, interceptor swales or other drainage devices on or adjacent to the project site. Drainage patterns established at the time of grading should be maintained for the life of the project.

9.7 Review of Construction Plans

The recommendations provided in this report are based on preliminary design information for the proposed construction. We recommend that a copy of the plans be provided to Ninyo & Moore for review before bidding to check the interpretation of our recommendations and verify that the designed improvements are consistent with our assumptions. It should be noted that, upon review of these documents, some recommendations presented in this report might be revised or modified to meet the project requirements.

9.8 Construction Observation and Testing

The recommendations provided in this report are based on subsurface conditions encountered in relatively widely spaced exploratory borings. During construction, the geotechnical engineer or his representative in the field should be allowed to check the exposed subsurface conditions. During construction, the geotechnical engineer or his representative should be allowed to:

- Observe preparation and compaction of subgrade.
- Check and test imported materials prior to use as fill.
- Observe placement and compaction of fill.
- Perform field density tests to evaluate fill and subgrade compaction.
- Observe foundation excavations prior to placement of steel reinforcing or concrete.

The recommendations provided in this report assume that Ninyo & Moore will be retained as the geotechnical consultant during the construction phase of the project. If another geotechnical consultant is selected, we request that the selected consultant provide a letter to the architect and the owner (with a copy to Ninyo & Moore) indicating that they fully understand Ninyo & Moore's recommendations, and that they are in full agreement with the recommendations contained in this report.

10 LIMITATIONS

The field evaluation, laboratory testing, and geotechnical analyses presented in this geotechnical report have been conducted in general accordance with current practice and the standard of care exercised by geotechnical consultants performing similar tasks in the project area. No warranty, expressed or implied, is made regarding the conclusions, recommendations, and opinions presented in this report. There is no evaluation detailed enough to reveal every subsurface condition. Variations may exist, and conditions not observed or described in this report may be encountered during construction. Uncertainties relative to subsurface conditions can be reduced through additional subsurface exploration. Additional subsurface evaluation will be performed upon request. Please also note that our evaluation of structural issues, environmental concerns, or the presence of hazardous materials.

This document is intended to be used only in its entirety. No portion of the document, by itself, is designed to completely represent any aspect of the project described herein. Ninyo & Moore should be contacted if the reader requires additional information or has questions regarding the content, interpretations presented, or completeness of this document.

This report is intended for design purposes only. It does not provide sufficient data to prepare an accurate bid by contractors. It is suggested that the bidders and their geotechnical consultant perform an independent evaluation of the subsurface conditions in the project areas. The independent evaluations may include, but not be limited to, review of other geotechnical reports prepared for the adjacent areas, site reconnaissance, and additional exploration and laboratory testing.

Our conclusions, recommendations, and opinions are based on an analysis of the observed site conditions. If geotechnical conditions different from those described in this report are encountered, our office should be notified and additional recommendations, if warranted, will be provided upon request. It should be understood that the conditions of a site could change with time as a result of natural processes or the activities of man at the subject site or nearby sites. In

addition, changes to the applicable laws, regulations, codes, and standards of practice may occur because of government action or the broadening of knowledge. The findings of this report may, therefore, be invalidated over time, in part or in whole, by changes over which Ninyo & Moore has no control.

This report is intended exclusively for use by the client. Any use or reuse of the findings, conclusions, and/or recommendations of this report by parties other than the client is undertaken at said parties' sole risk.

11 **REFERENCES**

- American Concrete Institute, 2016, Building Code Requirements for Structural Concrete (ACI 318-14) - Commentary on Building Code Requirements for Structural Concrete (ACI 318R-14).
- American Concrete Institute, 2017, ACI Manual of Concrete Practice.
- American Society for Testing and Materials (ASTM), 2019, Annual Book of ASTM Standards, West Conshohocken, Pennsylvania.
- American Society of Civil Engineer (ASCE), 2010, Minimum Design Loads for Buildings and Other Structures, Standard 7-10.
- Bonilla, M.G, Geologic Map of the San Francisco South, California: U.S. Geological Survey Open-File Report 98-137.
- California Building Standards Commission (CBSC), 2016, California Building Code (CBC), Title 24, Part 2, Volumes 1 and 2, based upon the 2015 International Building Code (IBC).
- California Department of Transportation (Caltrans), 2018a, Corrosion Guidelines, Version 3.0, Division of Engineering Services, Materials Engineering and Testing Services, Corrosion Branch: dated March.
- California Department of Transportation (Caltrans), 2018b, Standard Specifications.
- California Geological Survey (CGS), 1982, State of California Special Studies Zones, Newark 7.5 Minute Quadrangle, Alameda County, California, Scale: 1:24,000, dated January 1.
- California Geological Survey (CGS), 2003a, Seismic Hazard Report 090 for the Newark 7.5 Minute Quadrangle, Alameda County, California.
- California Geological Survey (CGS), 2003b, Seismic Hazard Zone Map for the Newark 7.5 Minute Quadrangle, Alameda County, California, Scale: 1:24,000, dated July 2.
- California Department of Transportation (Caltrans), 2020, Highway Design Manual, https://dot.ca.gov/programs/design/manual-highway-design-manual-hdm, dated July 1.
- California Department of Transportation (Caltrans), 2021, Corrosion Guidelines, Version 3.2, Division of Engineering Services, Materials Engineering and Testing Services, Corrosion Branch: dated May.
- California Department of Transportation (Caltrans), 2022, Standard Specifications.
- Google Earth, 2023, Google Earth 7.3.2.5776, http://earth.google.com/.
- Occupational Safety and Health Administration (OSHA), 1989, Occupational Safety and Health Standards – Excavations, Department of Labor, Title 29 Code of Federal Regulations (CFR) part 1926, dated October 31.
- California Department of Conservation (DOC) 2020, Shear Wave Velocity map in Upper 30m of Surficial Geology (Vs30).
Docusign Envelope ID: A3228B90-A416-4991-8459-8F8515CCF10B

FIGURES

Ninyo & Moore San Bruno Mountain Day Use Area, Daly City, California 404778001 April 19, 2024





Geotechnical & Environmental Sciences Consultants

404778001 I 04/24

Docusign Envelope ID: A3228B90-A416-4991-8459-8F8515CCF10B







LEGEND



Areas where historic occurrence of liquefaction, or local geological, geotechnical and groundwater conditions indicate a potential for permanent ground displacements such that mitigation as defined in Public Resources Code Section 2693(c) would be required.



EARTHQUAKE-INDUCED LANDSLIDE ZONES: Areas where previous occurrence of landslide movement, or local topographic, geological, geotechnical and subsurface water conditions indicate a potential for permanent ground displacements such that mitigation as defined in Public Resources Code Section 2693(c) would be required.



NOTE: DIMENSIONS, DIRECTIONS, AND LOCATIONS ARE APPROXIMATE I REFERENCE: CGS, 1982, 2021



FIGURE 5



LIQUEFACTION ZONES:

SAN BRUNO MOUNTAIN DAY USE AREA NEW RESTROOM AND PICNIC AREAS 555 GUADALUPE CANYON PARKWAY DALY CITY, CALIFORNIA

404778001 I 04/24

SEISMIC HAZARD ZONES

404778001.dwg 04/02/2024 AEK

Docusign Envelope ID: A3228B90-A416-4991-8459-8F8515CCF10B

APPENDIX A

Boring Logs

Ninyo & Moore | San Bruno Mountain Day Use Area, Daly City, California | 404778001 | April 19, 2024

APPENDIX A

BORING LOGS

Field Procedure for the Collection of Disturbed Samples

Disturbed soil samples were obtained in the field using the following methods.

Bulk Samples

Bulk samples of representative earth materials were obtained from the exploratory borings. The samples were bagged and transported to the laboratory for testing.

The Standard Penetration Test (SPT) Sampler

Disturbed drive samples of earth materials were obtained by means of a Standard Penetration Test sampler. The sampler is composed of a split barrel with an external diameter of 2 inches and an unlined internal diameter of 1-3/8 inches. The sampler was driven into the ground 12 to 18 inches with a 140-pound hammer freely falling from a height of 30 inches in general accordance with ASTM D 1586. The blow counts were recorded for every 6 inches of penetration; the blow counts reported on the logs are those for the last 12 inches of penetration. Soil samples were observed and removed from the sampler, bagged, sealed and transported to the laboratory for testing.

Field Procedure for the Collection of Relatively Undisturbed Samples

Relatively undisturbed soil samples were obtained in the field using the following methods.

The Modified Split-Barrel Drive Sampler

The sampler, with an external diameter of 3.0 inches, was lined with 6-inch long, thin brass liners with inside diameters of approximately 2.4 inches. The sample barrel was driven into the ground with the weight of a hammer in general accordance with ASTM D 3550. The driving weight was permitted to fall freely. The approximate length of the fall, the weight of the hammer, and the number of blows per foot of driving are presented on the boring log as an index to the relative resistance of the materials sampled. The samples were removed from the sample barrel in the brass liners, sealed, and transported to the laboratory for testing.

Г

DEPTH (feet)	Bulk SAMPLES	BLOWS/FOOT	MOISTURE (%)	DRY DENSITY (PCF)	SYMBOL	CLASSIFICATION U.S.C.S.	BORING LOG EXPLANATION SHEET
0							Bulk sample.
-							Modified split-barrel drive sampler.
-							No recovery with modified split-barrel drive sampler.
_	I						Sample retained by others.
-							Standard Penetration Test (SPT).
5-							No recovery with a SPT.
-		xx/xx					Shelby tube sample. Distance pushed in inches/length of sample recovered in inches.
-							No recovery with Shelby tube sampler.
-							Continuous Push Sample.
-			δ				Seepage.
10-			<u> </u>				Groundwater encountered during drilling.
-			-				Groundwater measured after drilling.
						SM	MAJOR MATERIAL TYPE (SOIL):
-							Solid line denotes unit change.
-						CL	Dashed line denotes material change.
							Attitudoo: Striko/Din
-							b: Bedding
45							c: Contact
15-							f: Fracture
-							F: Fault
							s: Shear
-							bss: Basal Slide Surface
							si: Shear Fracture sz: Shear Zone
							sbs: Shear Bedding Surface
20-							The total depth line is a solid line that is drawn at the bottom of the boring.



BORING LOG

	Soil Clas	sification Cl	hart	Per AST	M D 2488		Grain Size				
		iono		Seco	ndary Divisions		Doco	rintion	Sieve	Grain Siza	Approximate
F	rimary Divis	sions	Gro	up Symbol	Group Name		Desc	πριιοπ	Size	Grain Size	Size
		CLEAN GRAVEL		GW	well-graded GRAVEL		Bou	Iders	> 12"	> 12"	Larger than
		less than 5% fines		GP	poorly graded GRAVEL						basketball-sized
	GRAVEL			GW-GM	well-graded GRAVEL with silt		Cot	obles	3 - 12"	3 - 12"	Fist-sized to
	more than	GRAVEL with DUAL		GP-GM	poorly graded GRAVEL with silt						
	coarse	CLASSIFICATIONS 5% to 12% fines		GW-GC	well-graded GRAVEL with clay			Coarse	3/4 - 3"	3/4 - 3"	Thumb-sized to fist-sized
	retained on			GP-GC	poorly graded GRAVEL with		Gravel				Dec cited to
	NO. 4 SIEVE	GRAVEL with		GM	silty GRAVEL			Fine	#4 - 3/4"	0.19 - 0.75"	thumb-sized
COARSE- GRAINED		FINES more than		GC	clayey GRAVEL						Rock-salt-sized to
SOILS		12% fines		GC-GM	silty, clayey GRAVEL			Coarse	#10 - #4	0.079 - 0.19"	pea-sized
50% retained on No. 200 sieve		CLEAN SAND		SW	well-graded SAND		Sand	Medium	#40 - #10	0 017 - 0 079"	Sugar-sized to
		less than 5% fines		SP	poorly graded SAND						rock-salt-sized
				SW-SM	well-graded SAND with silt			Fine	#200 - #40	0.0029 -	Flour-sized to
	SAND 50% or more	SAND with DUAL		SP-SM	poorly graded SAND with silt					0.017	suyai-sizeu
	of coarse fraction	CLASSIFICATIONS 5% to 12% fines		SW-SC	well-graded SAND with clay		Fi	nes	Passing #200	< 0.0029"	Flour-sized and smaller
	passes No. 4 sieve			SP-SC	poorly graded SAND with clay						
		SAND with FINES more than 12% fines		SM silty SAND					Plastic	ity Chart	
				SC	clayey SAND						
				SC-SM	silty, clayey SAND		70				
				CL	lean CLAY		% 60				
	SILT and	INORGANIC		ML	SILT		(Id) 50				
	CLAY liquid limit			CL-ML	silty CLAY		A D 40			CH or C	Н
FINE-	less than 50%	ORCANIC		OL (PI > 4)	organic CLAY		¥ 30				
GRAINED SOILS		ORGANIC		OL (PI < 4)	organic SILT		LICI1 20		CL or	OL	MH or OH
50% or more passes				СН	fat CLAY		. SP 10				
No. 200 sieve	SILT and CLAY	INORGANIC		МН	elastic SILT		₽ 7 4	CL - 1	ML ML o	r OL	
	liquid limit 50% or more	ORGANIC		OH (plots on or above "A"-line)	organic CLAY		0	0 10	20 30 40	0 50 60 7	70 80 90 100
		UNGANIC		OH (plots below "A"-line)	organic SILT				LIQUII	D LIMIT (LL),	%
	Highly (Organic Soils		PT	Peat						

Apparent Density - Coarse-Grained Soil

<u> </u>	parent De	marty - Obar	se-oranie							
	Spooling C	able or Cathead	Automatic Trip Hammer			Spooling Ca	ble or Cathead	Automatic Trip Hammer		
Apparent Density	SPT (blows/foot)	Modified Split Barrel (blows/foot)	SPT (blows/foot)	Modified Split Barrel (blows/foot)	Consis- tency	SPT (blows/foot)	Modified Split Barrel (blows/foot)	SPT (blows/foot)	Modified Split Barrel (blows/foot)	
Very Loose	≤ 4	≤ 8	≤ 3	≤ 5	Very Soft	< 2	< 3	< 1	< 2	
Loose	5 - 10	9 - 21	4 - 7	6 - 14	Soft	2 - 4	3 - 5	1 - 3	2 - 3	
Medium	11 - 30	22 - 63	8 - 20	15 - 42	Firm	5 - 8	6 - 10	4 - 5	4 - 6	
Dense			0-20 15-		Stiff	9 - 15	11 - 20	6 - 10	7 - 13	
Dense	31 - 50	64 - 105	21 - 33	43 - 70	Very Stiff	16 - 30	21 - 39	11 - 20	14 - 26	
Very Dense	> 50	> 105	> 33	> 70	Hard	> 30	> 39	> 20	> 26	



USCS METHOD OF SOIL CLASSIFICATION

Consistency - Fine-Grained Soil

DEPTH (feet) Bulk SAMPLES Driven SAMPLES BLOWS/FOOT MOISTURE (%) DRY DENSITY (PCF) SYMBOL SYMBOL	CLASSIFICATION U.S.C.S.	DATE DRILLED 3/18/2024 BORING NO. B-1 GROUND ELEVATION 683' ± MSL SHEET 1 OF 1 METHOD OF DRILLING 4" SSA, B-24 Truck Mounted Rig (California Geotech) DROP 30 inches DRIVE WEIGHT 140 lbs (cathead) DROP 30 inches SAMPLED BY CJH LOGGED BY SSA REVIEWED BY MKW
 	ML	SLOPE DEBRIS AND RAVINE FILL: Dark brown, wet, stiff, sandy SILT; some clay; trace gravel. Yellowish brown and gray; moist; clayey SILT; some sand; trace gravel.
5 - 50/4"	ROCK	FRANCISCAN COMPLEX: Yellowish brown and reddish brown, moist, weathered SANDSTONE.
10		
15 50/5"		Total depth = 15 1/2 feet. Backfilled with neat cement shortly after drilling. <u>Notes:</u> Groundwater, though not encountered at the time of drilling, may rise to a higher level due to seasonal variations in precipitation and several other factors as discussed in the report. The ground elevation shown above is an estimation only. It is based on our interpretations of published maps and other documents reviewed for the purposes of this evaluation. It is not sufficiently accurate for preparing construction bids and design documents (Google, 2024).
Ninyo & Moore		FIGURE A- 1 SAN BRUNO MOUNTAIN DAY USE AREA SADDLE LOOP TRAIL, DALY CITY, CALIFORNIA

	PLES			Е,		7	DATE DRILLEDBORING NO
(feet)	SAM	-00T	KE (%)	гү (РС	Ч	S.	GROUND ELEVATION 688' ± MSL SHEET 1 OF 1
PTH		WS/F	STUF	-ISN	YMB	SIFIC J.S.C	METHOD OF DRILLING 4" SSA, B-24 Truck Mounted Rig (California Geotech)
DE	Bulk Driver	BLC	MOI	SY DE	0 0	CLAS	DRIVE WEIGHT 140 lbs (cathead) DROP 30 inches
				ä		0	SAMPLED BY LOGGED BY SSA REVIEWED BY MKW DESCRIPTION/INTERPRETATION
0						CL	SLOPE DEBRIS AND RAVINE FILL: Brown, wet, stiff, sandy silty lean CLAY with gravel.
_		<u> 15 </u>	20.3	109.1		SC	Yellowish brown and grayish brown, moist, stiff, silty clayey SAND, trace gravel.
5		24				CL	Yellowish brown, moist, lean CLAY, some silt, trace sand, trace gravel.
		50/1"				ROCK	FRANCISCAN COMPLEX: Yellowish brown and reddish brown, moist, weathered SANDSTONE.
-							Auger refusal encountered at 10 1/10 feet.
							Total depth = 10 1/10 feet.
-							Backfilled with neat cement shortly after drilling.
-							Notes:
-							Groundwater was measured at a depth of approximately 7 3/4 feet in the borehole shortly after completion of drilling.
15 —							Groundwater may rise to a level higher than that measured in borehole due to seasonal variations in precipitation and several other factors as discussed in the report.
-							The ground elevation shown above is an estimation only. It is based on our interpretations of published maps and other documents reviewed for the purpose of this evaluation. It is not sufficiently accurate for preparing construction bids and design documents (Google, 2024).
20 –							
							FIGURE A- 2
	Vin	nyo «	Ma	ore			SAN BRUND MOUNTAIN DAY USE AREA SADDLE LOOP TRAIL, DALY CITY, CALIFORNIA
G	eotechnic	al & Environm	nental Science	s Consultants			404778001 04/24

		Е.		7	DATE DRILLED	3/18/2024	BORING NO.	B-3
eet) SAMI		Y (PC	_	ATION.	GROUND ELEVATIO	ON <u>685' ± MSL</u>	SHEET	1OF1
	TURE	VSIT	'MBO	S.C.S	METHOD OF DRILLI	NG 4" SSA, B-24 Truc	k Mounted Rig (California	Geotech)
DEP	NOIS'	Y DEI	S	U.	DRIVE WEIGHT	140 lbs (catl	head) DF	OP 30 inches
	2	DR		D	SAMPLED BY	JH LOGGED BY DESCRIPTION/	SSA REVIEWED	BY <u>MKW</u>
	NE"			CL	SLOPE DEBRIS AN Yellowish brown, mo	ID RAVINE FILL: oist, stiff, lean CLA	Y, some sand, trace g	ravel.
5-50)/1"			ROCK	FRANCISCAN COM Yellowish brown, mo	<u>IPLEX:</u> oist, weathered SAI	NDSTONE.	
					Auger refusal encou	intered at 5 1/10 fee	et.	
					Total depth = 10 1/1	0 feet.		
					Backfilled with neat	cement shortly afte	er drilling.	
					Notes:			
					Groundwater, thoug level due to seasona discussed in the rep	h not encountered al variations in prec port.	at the time of drilling, ipitation and several o	may rise to a higher other factors as
					The ground elevatio interpretations of pu of this evaluation. It design documents (n shown above is a blished maps and o is not sufficiently a Google, 2024).	an estimation only. It is other documents revie ccurate for preparing	s based on our wed for the purposes construction bids and
15								
20		<u> </u>						FIGURE A- 3
Niny	0 & Ma	ore				SAD	SAN BRUNO MOUN DLE LOOP TRAIL, DA	TAIN DAY USE AREA _Y CITY, CALIFORNIA
Geotechnical & Er	nvironmental Science	s Consultants						404778001 04/24

1

Docusign Envelope ID: A3228B90-A416-4991-8459-8F8515CCF10B

APPENDIX B

Laboratory Testing

Ninyo & Moore San Bruno Mountain Day Use Area, Daly City, California 404778001 April 19, 2024

APPENDIX B

LABORATORY TESTING

Classification

Soils were visually and texturally classified in accordance with the Unified Soil Classification System (USCS) in general accordance with ASTM D 2488. Soil classifications are indicated on the logs of the exploratory borings in Appendix A.

In-Place Moisture and Density Tests

The moisture content and dry density of relatively undisturbed samples obtained from the exploratory borings were evaluated in general accordance with ASTM D 2937. The test results are presented on the logs of the exploratory borings in Appendix A.

Gradation Analysis

Gradation analysis tests were performed on selected representative soil samples in general accordance with ASTM D 422. The grain size distribution curves are shown on Figures B-1 through B-2. The test results were utilized in evaluating the soil classification in accordance with the USCS.

Atterberg Limits

Tests were performed on selected representative fine-grained soil samples to evaluate the liquid limit, plastic limit, and plasticity index in general accordance with ASTM D 4318. These test results were utilized to evaluate the soil classification in accordance with the USCS. The test results and classifications are shown on Figure B-2.

Unconfined Compression Test

Unconfined compression tests were performed on a relatively undisturbed sample in general accordance with ASTM D 2166. The test results are shown on Figure B-3.

Expansion Index Tests

The expansion index of a selected soil samples were evaluated in accordance with ASTM D 4829. The specimens were molded under a specified compactive energy at approximately 50 percent saturation (plus or minus 1 percent). The prepared 1-inch thick by 4-inch diameter specimen was loaded with a surcharge of 144 pounds per square foot and inundated with tap water. Readings of volumetric swell were made for a period of 24 hours. The results of this test are presented on Figure B-4.

		PASSING NO. 4	PASSING NO. 200	SAMPLE)
20.25		E 4	47	CANDOTONE
2.0 - 2.5	weathered SANDSTONE	54	17	SANDSTONE
	2.0 - 2.5	2.0 - 2.5 weathered SANDSTONE	2.0 - 2.5 weathered SANDSTONE 54	2.0 - 2.5 weathered SANDSTONE 54 17

PERFORMED IN ACCORDANCE WITH 1140

FIGURE B-1

NO. 200 SIEVE ANALYSIS TEST RESULTS

SAN BRUNO MOUNTAIN DAY USE AREA SADDLE LOOP TRAIL, DALY CITY, CALIFORNIA

Ningo & **Moore** Geotechnical & Environmental Sciences Consultants

404778001 | 04/24

SYMBOL	LOCATION	DEPTH (ft)	LIQUID LIMIT	PLASTIC LIMIT	PLASTICITY INDEX	USCS CLASSIFICATION (Fraction Finer Than No. 40 Sieve)	USCS
	B-2	2.5 - 3.0	35	14	21	CL	CL



PERFORMED IN ACCORDANCE WITH ASTM D 4318



FIGURE B-2

ATTERBERG LIMITS TEST RESULTS

SAN BRUNO MOUNTAIN DAY USE AREA SADDLE LOOP TRAIL, DALY CITY, CALIFORNIA 404778001 | 04/24



404778001 | 04/24

SAMPLE LOCATION	SAMPLE DEPTH (ft)	INITIAL MOISTURE (percent)	COMPACTED DRY DENSITY (pcf)	FINAL MOISTURE (percent)	VOLUMETRIC SWELL (in)	EXPANSION INDEX	POTENTIAL EXPANSION
В-3	0.0-5.0	10.1	108.3	22.4	0.0499	50	Low

PERFORMED IN ACCORDANCE WITH ASTM D 4829

FIGURE B-4

Ningo & Moore Geotechnical & Environmental Sciences Consultants EXPANSION INDEX TEST RESULTS SAN BRUNO MOUNTAIN DAY USE AREA 555 GUADALUPE CANYON PARKWAY, DALY CITY, CA 404778011 | MARCH 2024

APPENDIX C

Corrosivity Testing (CERCO Analytical)

Ninyo & Moore San Bruno Mountain Day Use Area, Daly City, California 404778001 April 19, 2024



1100 Willow Pass Court, Suite A Concord, CA 94520-1006 925 462 2771 Fax. 925 462 2775 www.cercoanalytical.com

9 April, 2024

Job No. 2404011 Cust. No.13270

Mr. Caleb Hodnett Ninyo & Moore 2149 O'Toole Avenue, Suite 30 San Jose, CA 95131

Subject: Project No.: 404778001 Project Name: San Bruno Mountain Day Use Area, 555 Guadalupe Canyon Parkway, Daly City, CA Corrosivity Analysis – ASTM Test Methods

Dear Mr. Hodnett:

Pursuant to your request, CERCO Analytical has analyzed the soil sample submitted on April 03, 2024. Based on the analytical results, this brief corrosivity evaluation is enclosed for your consideration.

Based upon the resistivity measurement, the sample is classified as "moderately corrosive". All buried iron, steel, cast iron, ductile iron, galvanized steel and dielectric coated steel or iron should be properly protected against corrosion depending upon the critical nature of the structure. All buried metallic pressure piping such as ductile iron firewater pipelines should be protected against corrosion.

The chloride ion concentration reflects none detected with a reporting limit of 15 mg/kg.

The sulfate ion concentration reflects none detected with a reporting limit of 15 mg/kg.

The pH of the soil is 7.02, which does not present corrosion problems for buried iron, steel, mortar-coated steel and reinforced concrete structures.

The redox potential is 110-mV and is indicative of potentially "moderately corrosive" soils resulting from anaerobic soil conditions.

This corrosivity evaluation is based on general corrosion engineering standards and is non-specific in nature. For specific long-term corrosion control design recommendations or consultation, please call JDH Corrosion Consultants, Inc. at (925) 927-6630.

We appreciate the opportunity of working with you on this project. If you have any questions, or if you require further information, please do not hesitate to contact us.

Very truly yours, CERÇO ANALYTICAL, INC. for

J. Darby Howard, Jr., P.E. President

JDH/jdl Enclosure

Client:	Ninyo & Moore
Client's Project No .:	404778001
Client's Project Name:	San Bruno Mountain Day Use Area, 555 Guadalupe Canyon Parkway, Daly City, CA
Date Sampled:	18-Mar-24
Date Received:	3-Apr-24
Matrix:	Soil
Authorization:	Signed Chain of Custody

CERCO a n a l y t i c a l 1100 Willow Pass Court, Suite A Concord, CA 94520-1006 925 462 2771 Fax. 925 462 2775 www.cercoanalytical.com

9-Apr-2024

Date of Report:

Job/Sample No.	Sample I.D.	Redox (mV)	рН	Conductivity (umhos/cm)*	Resistivity (100% Saturation) (ohms-cm)	Sulfide (mg/kg)*	Chloride (mg/kg)*	Sulfate (mg/kg)*
2404011-001	B-2/0.0-5.0'	110	7.02	-	5,900		N.D.	N.D.
							1	
				12				

Method:	ASTM D1498	ASTM D4972	ASTM D1125M	ASTM G57	ASTM D4658M	ASTM D4327	ASTM D4327
Reporting Limit:			10	1	50	15	15
Date Analyzed:	3-Apr-2024	4-Apr-2024	-	4-Apr-2024	_	4-Apr-2024	4-Apr-2024

* Results Reported on "As Received" Basis

N.D. - None Detected

Julia Clauson Chemist

Quality Control Summary - All laboratory quality control parameters were found to be within established limits

Docusign Envelope ID: A3228B90-A416-4991-8459-8F8515CCF10B Chain of Custody

24	0.	4	0	1	
	Page	1	of	1	



40	Job No. 04778001		CU#	20		С	lient Pro	ject I.D.		1	Schee	lule		1				Da	te Sampled	Da	te Due
Full N	lame				PI	none 40	84359	000 x 1	5329			ANALY	'SIS					ASTM			
Ca	aleb Hodnett					Fax						1						1	TT		
Comp Niny	oany and/or Mailin o & Moore 214	g Addre 9 O'Tc	ss pole Aver	nue Sui	te 30	_{Cell} San J	ose CA			ntial				100%		ation					
Samp San	le Source Bruno Mountain Day	Use Are	a, 555 Guad	lalupe Ca	nyon Pa	arkway, I	Daly City,	СА		ox Pote		ate	nide	stivity- rated		Evalua					
ab No	. Sample I.D.		Date	Time	Matrix	Contai	n. Size	Preserv.	Qtv.	Red	Hd	Sulf	Chlo	Resi		Brief					
	B-2/0.0-5.0'				S					x	x	x	x	x		x					
_																					
									1	-											
_																					
_		_																			
_																					
-										6											
-					_				_						_						
1																-				_	
D\ G\ SV	V - Drinking Water V - Ground Water V - Surface Water	LIONS	HB - Hoset PV - Petcoo	bib k Valve	EIPT	Total No	o. of Cont:	ainers		Relinc	quished	By:	Br	n ste	5		Date	3/19/2	24	Fime 1	0:00am
W	W - Waste Water	REVIA	PH - Pump RR - Restro	House om	LEREC	Conform	ns to Reco	rd		Receiv	ved By	: n	-Ne	ole.	To	ien	Date 4	1	3/2 F	lime	2:50
S - Pro	- Sludge Soil oduct	ABBF	GL - Glass PL - Plastic ST - Sterile		SAMP	Temp. a Sampler	t Lab -°C			Reling	luishec	By:		00	2 10	- yo	Date	10	י <u>י</u> ן ר	īme	
mme	nts:							1.75.1		Receiv	ved By	;				[Date		1	īme	
SRE	IS AN ADDITION	AL CHA	ARGE FOR	EXTRU	DING	SOIL F	ROM M	ETAL TUI	BES	Relinq	uished	By:				[Date		Т	īme	
ail A	ddaræssa chodnet	t@niny	oandmoo	e.com						Receiv	ved By:					E	Date		Т	ïme	

APPENDIX D

Utility Survey

Ninyo & Moore San Bruno Mountain Day Use Area, Daly City, California 404778001 April 19, 2024



No.	REVISION	DATE	ΒY	
				SAN BRUNO MOUNTAIN
				DAY USE AREA
				DALY CITY

SURVEY NOTES

- 1. DATES OF FIELD SURVEY: 03/22/2024
- HORIZONTAL & VERTICAL CONTROL WAS BASED ON A GPS SURVEY USING GNSS RTK METHODS CONNECTED TO THE LEICA SMARTNET REAL TIME NETWORK TIED INTO CALIFORNIA STATE PLANE COORDINATES NAD83, EPOCH 2023.250.

UNDERGROUND UTILITY NOTE

THE TYPES, LOCATIONS, SIZES AND/OR DEPTHS OF EXISTING UNDERGROUND UTILITIES AS SHOWN ON THIS TOPOGRAPHIC SURVEY ARE APPROXIMATE AND WERE OBTAINED FROM SOURCES OF VARYING RELIABILITY. ONLY ACTUAL EXCAVATION WILL REVEAL THE TYPES, EXTENT, SIZES, LOCATIONS AND DEPTHS OF SUCH UNDERGROUND UTILITIES. A REASONABLE EFFORT HAS BEEN MADE TO LOCATE AND DELINEATE ALL KNOWN UNDERGROUND UTILITIES. HOWEVER, THE ENGINEER CAN ASSUME NO RESPONSIBILITY FOR THE COMPLETENESS OR ACCURACY OF ITS DELINEATION OF SUCH UNDERGROUND UTILITIES WHICH MAY BE ENCOUNTERED, BUT WHICH ARE NOT SHOWN ON THIS SURVEY.

SYMBOLS & ABBREVIATIONS



ASPHALT/CONCRETE AREA DRAIN BACK FLOW PREVENTOR CEDAR TREE TREE CLUSTER CONCRETE CYPRESS TREE EDGE OF PAVEMENT EUCALYPTUS TREE FINISHED GRADE AT DOOR IRRIGATION CONTROL VALVE MISCELLANEOUS CLEANOUT SANITARY MANHOLE VAULT

LEGEND





PARTIAL TOPOGRAPHIC SURVEY	
----------------------------	--

Copyright © 2022 by Sandis

SHEET





2149 O'Toole Avenue, Suite 30 | San Jose, California 95131 | p. 408.435.9000

ARIZONA | CALIFORNIA | COLORADO | NEVADA | TEXAS | UTAH

ninyoandmoore.com



Geotechnical & Environmental Sciences Consultants

Pre-Demolition Hazardous Materials Survey and Soil Sampling for NOA

San Bruno Mountain Park 555 Guadalupe Canyon Parkway Brisbane, CA

June 7, 2024 | Report Number: R1247013

Prepared for:

Capital Program Management, Inc. 1851 Heritage Lane, Suite 210 Sacramento, CA 95815





Nationwide Terracon.com

Facilities
 Environmental
 Geotechnical
 Materials



June 7, 2024

Capital Program Management, Inc. 1851 Heritage Lane, Suite 210 Sacramento, CA 95815

Attn: Matthew Estes E: <u>matthew@capitalpm.com</u>

 RE: Pre-demolition Hazardous Materials Survey and Soil Sampling San Bruno Mountain State Park
 Day Use Area/Restroom Building
 555 Guadalupe Canyon Parkway
 Brisbane, California
 Terracon Project No: R1247013

Matthew Estes:

Terracon Consultants, Inc. (Terracon) is pleased to submit the attached report for the referenced site to **Capital Program Management, Inc**. The purpose of this report is to present the results of the predemolition hazardous materials survey performed on May 14, 2024. This survey was conducted in general accordance with Terracon's proposal PR1247013, dated January 3, 2024. We understand this survey was requested due to the planned demolition of the site structures, as well as to collect limited soil data in order to assess for the presence of naturally occurring asbestos (NOA).

Terracon appreciates the opportunity to provide this service to **Capital Program Management, Inc.** If you have any questions regarding this report, please contact our office at your convenience.

Sincerely, Terracon Consultants, Inc.

William Frieszell, CIH, CAC, CDPH Lead I/A Industrial Hygiene Consultant

Karin Achroete

Karin Schroeter, CIH, CAC, CDPH Lead I/A Industrial Hygiene Department Manager

TABLE OF CONTENTS

1.0	INTE	RODUCTI ON	1
	1.1	Project ObjectiveEr	ror! Bookmark not defined.
	1.2	Reliance	1
2.0	BUIL	LDING DESCRIPTION	2
3.0	FIEL	LD ACTIVITIES	2
	3.1	Asbestos, Lead, PCBs, and Other Hazardous Building Mate	rials2
	3.2	Visual Assessment - Asbestos	
	3.3	Physical Assessment - Asbestos	
	3.4 3.5	Sample Analysis - Asbestos	
	3.6	Visual Assessment – Lead Containing Paint and Materials	
	3.7	Physical Assessment – Lead Containing Paint and Materials	s 4
	3.8	Sample Analysis - Lead Containing Paint	
	3.9	PCBs – Sealants	
	3.10) Visual Assessment - Other Hazardous Building Materials Er	ror! Bookmark not defined.
4.0	REGI	SULATORY OVERVIEW	8
	4.1	Asbestos	
	4.2	Lead Containing Paint/Materials	
	4.3	Reirigerants Er	ror! Bookmark not defined.
	4.4		
ГО			
5.0	FINL		
	5.1	Asbestos	
	5.Z	Lead Containing Paint/Materials	
	5.3 5.4	Other Hazardous Building Materials	rorl Bookmark not defined
()	U. T		
0.U		ITATIONS/GENERAL COMMENTS	
APPEN	DIX A	A ASBESTOS ANALYTICAL LABORATORY DATA	

- APPENDIX B ASBESTOS ANALYTICAL DATA SOILS FOR NOA
- APPENDIX C LEAD ANALYTICAL LABORATORY DATA
- APPENDIX D PCB ANALYTICAL LABORATORY DATA
- APPENDIX E SAMPLE LOCATION FIGURES
- APPENDIX F CERTIFICATIONS



PRE-DEMOLITION HAZARDOUS MATERIALS SURVEY AND SOILS FOR NOA

San Bruno Mountain State Park Day Use Area/Restroom Building 555 Guadalupe Canyon Parkway Brisbane, California

Terracon Project No. R1247013 June 7, 2024

1.0 INTRODUCTION

Terracon Consultants, Inc. (Terracon) conducted a pre-demolition hazardous materials survey of the Restroom Building, which comprises a portion of the day use area of San Bruno Mountain State Park located at 555 Gudalupe Canyon Parkway in Brisbane, California (hereafter referred to as the site). The project scope also included the limited sampling of soil to a depth of thirty inches in locations throughout the day use area. The survey was conducted May 14, 2024, in general accordance with Terracon's proposal PR1247013, dated January 3, 2024, and the asbestos sampling protocols established in Environmental Protection Agency (EPA) regulation 40 Code of Regulations (CFR) Part 763 Subpart E 763.86, (Asbestos Hazard Emergency Response Act, AHERA). The survey scope included limited destructive sampling methods in interior, exterior and roofing level elevations of the affected structure. The survey included sample collection of suspect asbestos-containing materials (ACMs), naturally occurring asbestos (NOA) in site soil, lead containing paints (LCPs) and bulk materials, as well as polychlorinated biphenyls (PCBs) materials.

A total of fourteen (14) suspected asbestos containing materials (ACMs) were identified throughout the interior, exterior and roofing levels of the structure. Upon analysis, one of the sampled materials, a sink and basin undercoating, was reported to contain asbestos in concentrations exceeding the laboratory limit of detection. A further two (2) materials could not be sampled on the date of the survey, due to the level of destruction to finishes or building envelope, and have been assumed to contain asbestos in accordance with current regulatory framework.

Two (2) painted surfaces and one (1) ceramic tile glazing compound were sampled for potential lead content. One (1) of the three (3) samples collected was reported to contain lead in concentrations exceeding the laboratory limit of detection.

Six (6) samples of priority building materials established within local ordinances promulgated under the direction of the Bay Area Stormwater Management Agencies Association (BASMAA) were collected and analyzed for potential PCB content. None of the materials were reported to contain PCBs in detectable concentrations.

1.2 Reliance

This report is for the exclusive use of **Capital Program Management**, **Inc. (the client)** for the demolition of the day use area restroom structure within the San Bruno Mountain Park located 555 Guadalupe Canyon Parkway in Brisbane, California. Reliance by any other party on this report is prohibited without written authorization of Terracon and the client. Reliance on this report by



the client and all authorized parties will be subject to the terms, conditions, and limitations stated in the proposal, this report, and the project contract.

2.0 SCOPE OF WORK

The scope of the survey was as follows:

- Inspect the subject building for the presence of suspect ACMs, lead-containing paint and potential polychlorinated biphenyls (PCBs) containing materials.
- Collect samples of suspect ACMs following a National Emissions Standards for Hazardous Air Pollutants (NESHAPS) protocol for sample collection for a renovation survey.
- Asbestos bulk samples will be analyzed using polarized light microscopy (PLM) in accordance with the EPA's July 1993 method for the determination of asbestos in bulk building materials EPA 600/R-93/116.
- Collect bulk samples of primary painted surfaces and other materials suspected to be lead containing. Bulk samples will be analyzed at an accredited laboratory by Flame Atomic Absorption (AA) for Total Lead reported in parts per million (ppm).
- PCB samples will be analyzed at an accredited laboratory by Gas Chromatography and Electron Capture Detector (GC-ECD) techniques in general accordance with EPA Method 8082A.
- Perform up to six borings throughout the site in approximate locations specified by the client to conduct composite soil-sampling techniques to a depth of 30 inches.
- Submit soil samples to an accredited laboratory for analysis by CARB Method 435, which identifies asbestos aggregate fibers via enhanced PLM methodologies.
- Submit written report including analytical results, regulatory requirements and conclusions.

3.0 FIELD ACTIVITIES

3.1 Asbestos, Lead, PCBs, and Other Hazardous Building Materials

The survey was conducted by William Frieszell, Karin Schroeter and Brad Wallenberg, California Division of Occupational Safety and Health (Cal/OSHA) Certified Asbestos Consultants (CACs) and California Department of Public Health (CDPH) Lead Inspector/Assessors. Copies of pertinent training certifications are included in Appendix F. The asbestos portion of the survey was conducted in general accordance with the sample collection protocols established in EPA 40 CFR Part 763 Subpart E 763.86, AHERA. A summary of survey activities is provided below.

3.2 Visual Assessment - Asbestos

Survey activities were initiated with visual observation of the subject structures to identify homogeneous areas of suspect ACM. A homogeneous area (HA) consists of a building material that appears similar throughout in terms of color, size and texture with consideration given to the date of application.



3.3 Physical Assessment - Asbestos

A physical assessment of each HA of suspect ACM was conducted to assess the current friability and condition of the materials. A friable material is defined by the EPA as a material which can be crumbled, pulverized, or reduced to powder by hand pressure when dry. Friability was assessed by physically touching suspect materials.

Based on results of the visual observation, bulk samples of suspect ACM were collected in general accordance with EPA AHERA sampling protocols. Samples of suspect materials were collected from representative locations in each homogeneous area. Bulk samples were collected using wet methods as applicable to reduce the potential for fiber release. Samples were placed in sealable containers and labeled with unique sample numbers using an indelible marker.

The selection of sample locations and frequency of sampling were based on Terracon's observations and the assumption that like materials in the same area are homogeneous in content.

Terracon collected thirty-three (33) samples from the fourteen (14) homogeneous areas of suspect ACMs. Laboratory analysis reported that one (1) of the materials contained asbestos. A summary of the materials reported as containing asbestos is included in Tables I & II below, with full laboratory data and chains of custody provided in Appendix A.

3.4 Sample Analysis - Asbestos

Asbestos bulk samples were submitted under chain of custody to SGS Forensic Analytical Laboratories (SGS) in Carson, California for analysis by polarized light microscopy (PLM) with dispersion staining techniques per EPA methodology 600/R-93/116. The percentage of asbestos, where applicable, was determined by microscopic visual estimation. Eurofins is accredited under the National Voluntary Laboratory Accreditation Program (NVLAP) Accreditation (No. 200757-0).

3.5 Naturally Occurring Asbestos

Prior to conducting soil sampling activities, Terracon requested that USA North 811 report on any subsurface utilities within the sampling area. In addition, a private utility locator service was contracted and accompanied the technician to each of the six (6) marked locations where they were additionally cleared with a radius of five feet for shallow sampling. Finally, Terracon used a Leica wand prior to advancing the borings. A Leica is a basic locator that utilizes power and radio modes to find power lines under load and potential metallic lines. Terracon did not discover lines additional to those found and marked by the private locator firm.

During the survey, Terracon collected a total of six (6) multi-point composite samples, from approximate locations specified by the client, in order to assess for the presence of naturally occurring asbestos. Approximate coordinates for the six borings performed during the survey are as follows:

- San Bruno Mountain Park Coordinates: 37.697313, -122.433745
- San Bruno Mountain Park Coordinates: 37.697449, -122.433893
- San Bruno Mountain Park Coordinates: 37.697542, -122.434075
- San Bruno Mountain Park Coordinates: 37.697765, -122.433906



- San Bruno Mountain Park Coordinates: 37.697301, -122.432968
- San Bruno Mountain Park Coordinates: 37.697456, -122.432872

Samples were collected via manual hand auger using multi-point composite sampling techniques to a depth of approximately thirty inches below ground surface in each of the locations listed.

Soil samples were extracted by hand using a decontaminated trowel and disposable gloves and placed directly into discrete packaging containers. Each sample container was labeled with the project number, date, time, boring number, and sample number. Sample containers were placed in a chilled cooler immediately after sampling. Upon completion of soil sampling activities during each project phase, all samples were transported to Micro Analytical Laboratory (Micro) in Emeryville, California for analysis by CARB 435 PLM methodologies. Micro holds current accreditation as a California Environmental Laboratory Accreditation Program (ELAP)-certified laboratory.

At the completion of the field activities, each sample location was backfilled with soil cuttings generated from boring activities.

None of the soil samples were reported to contain asbestos in detectable concentrations. The data for soil borings is summarized in Table II, along with confirmed non-ACM building materials. Laboratory data reports and Terracon chains of custody are attached to this report in Appendix B.

3.6 Lead Containing Paint and Materials

Terracon collected paint chip, ceramic tiles, and building sealant samples to determine the lead content in parts per million (ppm) of the predominant interior and exterior surfaces and sealants associated with the structure. Suspect lead samples were collected in sealable containers and labeled with unique sample numbers using an indelible marker.

3.7 Visual Assessment – Lead Containing Paint and Materials

Inspection activities began with visual observations of painted and glazed surfaces to identify unique combinations of paint and suspect bulk materials. A unique combination of paint consists of paint that is applied to a building material and has similar color, substrate, and component. The assessment was conducted throughout the buildings. Suspect lead caulking and ceramic tiles were identified and associated with the interior and exterior of the portable restroom building.

3.8 Physical Assessment – Lead Containing Paint and Bulk Materials

A physical assessment of unique combinations of paint was conducted to assess the condition of the paint. Lead paint chip samples were collected to comply with Cal-OSHA regulations (Title 8 CCR 1532.1 – Lead Exposure in Construction) for the proposed demolition activities. Paint was sampled to identify potential worker exposure and potential disposal restrictions. All interior and exterior painted surfaces were in fair to good condition and generally intact with the building substrates at the time of the survey.

Terracon sampled two (2) painted surfaces and one (1) bulk material during the survey. One (1) of the materials sampled, a ceramic tile glazing compound, was found to contain lead concentrations in exceedance of the laboratory detection limit. A summary of suspect lead paint and materials collected during the survey is summarized in Table III.



3.9 Sample Analysis - Lead Containing Paint and Bulk Materials

Paint chip and bulk samples were submitted under chain of custody to Eurofins in Tustin and were analyzed by Flame Atomic Absorption method SW846-7000B. Eurofins is accredited by the American Industry Hygiene Association's (AIHA) Environmental Lead Laboratory Accreditation Program (ELLAP) (Laboratory Code 178697) to perform Flame Atomic Absorption analysis. The laboratory reports for the lead samples are included as Appendix C.

3.10 PCBs - Sealants

Terracon collected six (6) bulk samples from two (2) priority building materials as established under local ordinance under the direction of the Bay Area Stormwater Management Agencies Association (BASMAA). Bulk sealant samples were collected using a razor knife and were placed into individual containers. Each sample was provided a discrete sample number, which was recorded on a chain of custody form. The samples were transported under chain of custody procedures to SGS Excelchem Laboratories, Inc. (SGS) in Rocklin, California. All samples were analyzed for PCB content in accordance with EPA Method SW8082A. The laboratory reports for the PCB samples are included as Appendix D. A summary of the PCB materials samples is included in Table IV.

4.0 FINDINGS

4.1 Asbestos

Asbestos was identified or assumed in three (3) of the fourteen (14) building materials identified during the survey. A summary of confirmed ACMs is provided in Table I, below. Laboratory analytical reports are included as Appendix A. Sample locations are depicted on figures in Appendix E.

Material Description	Material Locations	Asbestos Content by Layer	NESHAP Category	Est. Qty
Mirror Mastic	Material Located at Single Mirror at Women's Restroom Area	ASSUMED	Cat. II	2 SF
Skylight Assembly Sealant/Caulk/Putty	Material is Located at Upper Roofing Level at Skylights	ASSUMED	Cat. II	10 SF
Restroom Fixture Rear Coat Mastic - Black	Material is Present behind Toilets, Urinals and Sinks throughout Restrooms	Trace Chrysotile Asbestos CO. <25% CH (verified by 400-pt Point Count analysis)	Cat. II	250 sf

Table I Asbestos Containing Materials

ND = None Detected, CH = Chrysotile, NA = Not Applicable, RACM = Regulated asbestos containing material (friable), Cat. I = Non-friable (note ACM must be reclassified as a RACM if rendered friable during removal), Cat. II = Category II Nonfriable (note ACM must be reclassified as a RACM if rendered friable during removal), LF = Linear Feet, SF = square feet, *Estimate quantity should be field verified prior to abatement or abatement design



Eleven (11) of the fourteen (14) materials identified during the survey were not reported to contain asbestos in concentrations exceeding the laboratory's limit of detection upon analysis. Additionally, Terracon performed multi-point composite sampling for naturally occurring asbestos (NOA) in six locations to a maximum depth of 30 inches. Asbestos was not reported to be present in any of the soil samples collected. The confirmed non-ACMs are summarized in Table II, below.

Material Description	Sample Locations
Roofing Field System - Asphaltic Shingle	Southeastern Corner of Upper Roof Eastern Perimeter of Building Northeastern Corner of Upper Roof
Pressure Grout in CMU Wall Systems	Northwestern Corner of Building Western Elevation at Foundation Storage/Electrical Room at Wall Penetration
Ceramic Tile Wainscot System - 4" Light Green Tile with Grout and Mortar	Men's Restroom at Divider Wall Men's Restroom at Northwestern Wall, North Side Women's Restroom at Accessible Stall
Block Filler Material at Divider Walls	Men's Restroom at Accessible Divider Men's Restroom at Single User Divider Women's Restroom at Accessible Stall
Perimeter Wall System - CMU Block with Mortar	Storage/Electrical Room at Entry Way Northwestern Corner of Building Southwestern Corner of Building
Concrete Floor/Slab System	Concrete Pad at Men's Restroom Southeastern Corner of Building Storage/Electrical Room at Entry Way
Caulking Bead at Vent Above Store Room	Storage/Electrical Room at Louver Northwestern Corner of Building
Asphalt Pavement	Pad Outside of Women's Restroom Pad at Bicycle Repair Area Pad at Exterior Sign Post
Felt Layer under Metal Roofing System	Northeastern Corner of Building at Overhang Southeastern Corner of Building at Overhang
Concrete Column Systems	Building Exterior at Southeastern Column Building Exterior at Northeastern Column Building Exterior at Southeastern Column
Restroom Door Frame Caulking - Yellow/White	Women's Restroom Entry Way Men's Restroom Entry Way
Day Use Area - Soils by CARB 435	Coordinates: 37.697313, -122.433745 Coordinates: 37.697449, -122.433893 Coordinates: 37.697542, -122.434075 Coordinates: 37.697765, -122.433906 Coordinates: 37.697301, -122.432968 Coordinates: 37.697456, -122.432872

Table II	
Non-Asbestos Containing	Materials

It should be reemphasized that although reasonable efforts were made to survey accessible suspect materials, additional suspect but un-sampled materials could be located under existing building materials, inside walls, above ceilings, in isolated areas or in other concealed areas. Therefore, if suspect materials are encountered during abatement and demolition activities that do not appear to have been characterized as ACM or non-ACM, these materials must be assumed to be ACM until samples are collected and analyzed to prove otherwise. Any assumed material



should be treated as asbestos or sampled to determine asbestos content before disturbing the material.

4.2 Lead Containing Paint/Materials

Terracon sampled two (2) predominant painted surfaces and one (1) ceramic tile glazing compound. The ceramic tile glazing compound was reported to contain lead in concentrations exceeding the laboratory detection limit. A summary of sample locations and analytical results is below in Table III. Samples reported with "<" are below the laboratory analytical reporting limit for the sample submitted. Laboratory reports are in Appendix C.

Sample #	Sample Description	Sample Location	Total Lead Result	Condition
Pb-01	Light Grey Glazing Compound on 4" Ceramic Wall Tile	Restroom Building Interior at Men's Section (also located in the women's room)	130 ppm	Intact
Pb-02	White Paint on CMU Block Exterior Wall System	Building Exterior at Northern Wall	<60 ppm	Intact
Pb-03	Grey Paint on Wooden Doorframe	Building Exterior at Sliding Door to Electrical Room	<70 ppm	Intact

Table III Lead Containing Paint/Materials

ppm = parts per million

Uncharacterized paints and/or suspect materials should be assumed to contain lead until sampling and analysis prove otherwise.

4.3 PCB Containing Materials

Terracon collected six (6) bulk samples from BASMAA priority building materials. None of the materials sampled were reported to contain PCBs in concentrations exceeding the laboratory reporting limit. A summary of PCB sample locations and analytical results is below in Table IV. Laboratory reports are provided in Appendix D.

Sample #	Material Description	Sample Location	PCB Content
PCB-1	Ventilation Louver Sealant above Door to Storage	South End of Lower Louver Edge	ND <1.96 mg/Kg
PCB-2	Ventilation Louver Sealant above Door to Storage	Approximate Center of Lower Louver Edge	ND <1.85 mg/Kg
PCB-3	Ventilation Louver Sealant above Door to Storage	North End of Lower Louver Edge	ND <1.92 mg/Kg
PCB-4	Restroom Building Doorway Caulking	Men's Restroom at Entry Way	ND <1.89 mg/Kg
PCB-5	Restroom Building Doorway Caulking	Women's Restroom at Entry Way	ND <1.92 mg/Kg

Table IV PCB Containing Materials
Pre-demolition Hazardous Materials Survey San Bruno Mountain Park - Day Use Area | Brisbane, CA June 7, 2024 | Terracon Report No. R1247013



Sample #	Material Description	Sample Location	PCB Content
PCB-6	Restroom Building Doorway Caulking	Men's Restroom at Entry Way	ND <2.00 mg/Kg

ppm = parts per million, < = less than laboratory reporting limit

5.0 REGULATORY OVERVIEW

5.1 Asbestos

The Asbestos NESHAP program in California is enforced by federal, state, and county Asbestos NESHAP Coordinators. For projects occurring in Brisbane, California, the Bay Area Air Quality Management District (BAAQMD) has been delegated authority from the EPA to enforce the Asbestos NESHAP within its respective jurisdictional boundaries, excluding tribal lands.

The asbestos NESHAP (40 CFR Part 61, Subpart M) regulates asbestos fiber emissions and asbestos waste disposal practices. The asbestos NESHAP regulation also requires the identification and classification of existing ACM according to friability prior to demolition or renovation activity. Friable ACM is a material containing more than 1% asbestos that, when dry, can be crumbled, pulverized, or reduced to powder by hand pressure. All friable ACM is considered regulated asbestos-containing material (RACM). The NESHAP regulation is implemented locally by the BAAQMD under Regulation 11, Rule 2.

The asbestos NESHAP regulation classifies ACM as either RACM, Category I non-friable ACM or Category II non-friable ACM. RACM includes all friable ACM, along with Category I and Category II non-friable ACM that has become friable, will be or has been subjected to sanding, grinding, cutting, or abrading, or ACM that has a high probability of becoming or has become crumbled, pulverized, or reduced to powder during renovation or demolition activity. Category I non-friable ACM are exclusively asbestos-containing packings, gaskets, resilient floor coverings, and asphalt roofing products that contain more than 1% asbestos. Category II non-friable ACM are all other non-friable materials other than Category I non-friable ACM that contain more than 1% asbestos.

Friable ACM, along with Category I and Category II non-friable ACM, which is in poor condition and has become friable or which will be subjected to drilling, sanding, grinding, cutting, abrading and which could be crushed or pulverized during anticipated renovation or demolition activities are considered regulated ACM (RACM).

Building materials confirmed to be ACM through the collection of bulk sampling and subsequent laboratory analysis, or presumed ACM, must be removed prior to intentional disturbance during the planned demolition activities. Asbestos abatement must be conducted by California licensed and registered abatement contractors and workers with Cal/OSHA-accredited training. Third-party air monitoring is recommended during the abatement activities.

Cal/OSHA requires that only properly licensed and certified asbestos abatement contractors be allowed to remove ACM. As per NESHAP, all RACM shall be removed from a facility being demolished or renovated before any non-burning demolition or renovation begins that would break up, dislodge, or similarly disturb the material or preclude access to the material for subsequent removal. According to the BAAQMD Regulation 11, Rule 2, if more than 100 square feet or 100 linear feet of any RACM is to be stripped, removed, dislodged, cut, drilled, or similarly disturbed, or for any demolition, the asbestos abatement contractor or facility owner must submit Pre-demolition Hazardous Materials Survey San Bruno Mountain Park - Day Use Area | Brisbane, CA June 7, 2024 | Terracon Report No. R1247013



an Asbestos Notification of Demolition and Renovation form to the BAAQMD along with the appropriate fees within at least 10 working days prior to the scheduled asbestos removal activity or demolition start date. Planned renovations that do not meet the definition of 'demolition or renovation of a facility' per BAAQMD and where no ACM exists do not require notification to NESHAP.

The Cal-OSHA asbestos standard for construction (Title 8 CCR 1529) regulates workplace exposure to asbestos. The Cal-OSHA standard requires that employee exposure to airborne asbestos must not exceed 0.1 fibers per cubic centimeter of air (0.1 f/cc) as an eight-hour time weighted average (TWA) and not exceed 1.0 fibers per cubic centimeter of air (1.0 f/cc) over a 30-minute time period known as an excursion limit (EL). The TWA and EL are known as Cal-OSHA's asbestos permissible exposure limits (PELs). The Cal-OSHA standard classifies construction and maintenance activities which could disturb ACM and specifies work practices and precautions which employers must follow when engaging in each class of regulated work.

Asbestos containing construction materials (ACCM) is a term developed by Cal/OSHA out of concern for non-hazardous building materials used inside and outside a building that contain less than 1% asbestos. The definition of ACCM includes any manufactured building material that has more than one-tenth of 1% (>0.1%) asbestos content. The BAAQMD requires point counting of friable samples of ACM at concentrations of less than 10% to determine more accurately the content of asbestos and proper classification of the material for proper abatement and disposal requirements. Alternatively, materials may be presumed as ACMs. If the material is less than one tenth of 1%, the material is not regulated by the EPA however Cal/OSHA worker protection regulations apply if any asbestos is detected.

5.2 Lead Containing Paint/Materials

Personnel performing demolition activities that may disturb painted components or materials with concentrations of lead above the designated analytical detection limit should comply with all current Cal-OSHA regulations in order to minimize employee exposure. Cal-OSHA defines lead containing paint as a paint, which contains lead, regardless of the concentration. Currently, any proposed renovation/demolition is subject to the Cal-OSHA regulations (Title 8 CCR 1532.1 – Lead Exposure in Construction). The Cal-OSHA regulation defines specific training requirements, engineering controls and working practices for construction personnel subject to this standard. Occupational exposure to lead occurring during construction work, including maintenance activities, painting, alteration, and repairs is subject to the Cal-OSHA Lead Exposure in Construction standard.

Construction work covered by Title 8 CCR 1532.1 includes any repair or renovation activities or other activities that disturb in-place lead-containing materials. Employers must assure that no employee will be exposed to lead at concentrations greater than 50 micrograms per cubic meter (μ g/m³) averaged over an eight-hour period without adequate protection. The Cal-OSHA Standard also establishes an action level of 30 μ g/m³ which if exceeded triggers the requirement for medical monitoring.

Proper waste stream categorization is required for the disposal of all lead containing materials and painted construction debris with total lead content that exceeds 50 ppm. The debris should be classified as hazardous waste if lead waste concentrations exceed either the total lead concentration or soluble lead concentration regulatory limits. Total lead concentration is determined by Total Threshold Limit Concentration (TTLC). Soluble or leachable lead is determined by the Soluble Threshold Limit Concentration (STLC, California required test) and/or

Pre-demolition Hazardous Materials Survey San Bruno Mountain Park - Day Use Area | Brisbane, CA June 7, 2024 | Terracon Report No. R1247013



Toxicity Characteristic Leaching Procedure (TCLP) (Federal EPA required test). Regulatory limits characterize a lead waste as a hazardous waste if lead concentrations exceed 1,000 ppm by TTLC or 5 milligram per liter by STLC or TCLP.

The above overview is not intended to be inclusive of all potentially pertinent regulatory information. The relevant EPA and OSHA standards should be consulted prior to undertaking activities involving the demolition, renovation, or maintenance of surfaces coated with lead containing paints.

an appropriate vendor.

5.3 PCBs

PCBs are regulated by the EPA under 40 CFR 761. The production of PCBs has been banned since 1979 and may be present in electrical capacitors, sealants, hydraulic oils, and transformers commonly found in buildings. Materials with greater than 50 ppm PCB content are considered PCB contaminated waste while materials with greater than 500 ppm PCB are considered PCB containing.

PCB containing equipment and/or contaminated materials must be removed and disposed properly prior to demolition of a building. PCB containing lighting ballasts may be present in some lighting fixtures and must be verified by labeling. All PCB containing materials must be removed and disposed prior to building demolition.

6.0 LIMITATIONS/GENERAL COMMENTS

Terracon performed limited destructive testing such as selective demolition of wall systems, dismantling of equipment or removal of protective coverings during the pre-demolition hazardous materials survey. Uncharacterized hidden materials may exist under existing finishes, equipment, or structural materials.

This hazardous materials survey was conducted in a manner consistent with the level of care and skill ordinarily exercised by members of the profession currently practicing under similar conditions in the same locale. The results, findings, conclusions, and recommendations expressed in this report are based on conditions observed during our survey at the subject site. The information contained in this report is relevant to the dates on which this survey was performed and should not be relied upon to represent conditions at a later date.

This report has been prepared on behalf of and exclusively for use by the client for specific application to their project as discussed. This report is not a bidding document. Contractors or consultants reviewing this report must draw their own conclusions regarding further investigation or remediation deemed necessary. Terracon does not warrant the work of regulatory agencies, laboratories, or other third parties supplying information which may have been used in the preparation of this report. No warranty, express or implied is made.



APPENDIX A

ASBESTOS ANALYTICAL LABORATORY DATA

Facilities | Environmental | Geotechnical | Materials



Bulk Asbestos Analysis (EPA Method 40CFR, Part 763, Appendix E to Subpart E and EPA 600/R-93-116, Visual Area Estimation)

NVLAP Lab Code: 101459-1

Terracon - Concord Karin Schroeteer 1220 Concord Ave suite 450 Concord, CA 94520					Client ID: Report Numbo Date Received Date Analyzed Date Printed: First Reported	L1969 er: B36015 : 05/15/2 I: 05/17/2 05/17/2 I: 05/17/2	3 4 4 4 4
Job ID/Site: R1247013 - San Bruno Mounta	ain Park - 555	Guadalupe Canyor	n Parkway, Brist	oane CA	SGSFL Job II): L1969	
Date(s) Collected: 05/14/2024					Total Samples Total Samples	Submitted: Analyzed:	33 33
Sample ID	Lab Numbe	Asbestos r Type	Percent in Layer	Asbestos Type	Percent in Layer	Asbestos Type	Percent in Layer
01A	12746867		ND				
Layer: 1an Roof Shingle	nononta	Ashastas (ND)	ND				
Cellulose (Trace) Fibrous Glass (40	%)	Asdestos (ND)					
01B Layer: Tan Roof Shingle Layer: Black Tar Layer: Tan Roof Shingle	12746868		ND ND ND				
Total Composite Values of Fibrous Com Cellulose (Trace) Fibrous Glass (40	ponents: %)	Asbestos (ND)					
01C Layer: Tan Roof Shingle	12746869		ND				
Total Composite Values of Fibrous Com Cellulose (Trace) Fibrous Glass (40	ponents: %)	Asbestos (ND)					
02A Layer: Grey Cementitious Material	12746870		ND				
Total Composite Values of Fibrous Com Cellulose (Trace)	ponents:	Asbestos (ND)					
02B Layer: Grey Cementitious Material Layer: Paint	12746871		ND ND				
Total Composite Values of Fibrous Com Cellulose (Trace)	ponents:	Asbestos (ND)					
02C Layer: Grey Cementitious Material Layer: Beige Cementitious Material Layer: Paint	12746872		ND ND ND				
Total Composite Values of Fibrous Com Cellulose (Trace)	ponents:	Asbestos (ND)					

Client Name: Terracon - Concord					Report Numb Date Printed:	er: B36015 05/17/2	53 24
Sample ID	Lab Numbe	Asbestos er Type	Percent in Layer	Asbestos Type	Percent in Layer	Asbestos Type	Percent in Layer
03A Layer: Beige Ceramic Tile Layer: White Grout Layer: White Mortar	12746873		ND ND ND				
Total Composite Values of Fibrous Con Cellulose (Trace)	ponents:	Asbestos (ND)					
03B Layer: Beige Ceramic Tile Layer: White Grout	12746874		ND ND				
Total Composite Values of Fibrous Con Cellulose (Trace)	ponents:	Asbestos (ND)					
03C Layer: Beige Ceramic Tile Layer: White Mortar	12746875		ND ND				
Total Composite Values of Fibrous Con Cellulose (Trace)	nponents:	Asbestos (ND)					
04A Layer: Grey Cementitious Material Layer: Paints	12746876		ND ND				
Total Composite Values of Fibrous Con Cellulose (Trace)	nponents:	Asbestos (ND)					
04B Layer: Dark Grey Cementitious Materia Layer: White Cementitious Material Layer: Paints	12746877 al		ND ND ND				
Total Composite Values of Fibrous Con Cellulose (Trace)	nponents:	Asbestos (ND)					
04C Layer: Grey Cementitious Material Layer: White Cementitious Material Layer: Paint	12746878		ND ND ND				
Total Composite Values of Fibrous Con Cellulose (Trace)	nponents:	Asbestos (ND)					
05A Layer: Dark Grey Cementitious Materia Layer: Grey Cementitious Material Layer: Paint	12746879 al		ND ND ND				
Total Composite Values of Fibrous Con Cellulose (Trace)	ponents:	Asbestos (ND)					

Client Name: Terracon - Concord					Report Numb Date Printed:	er: B36015 05/17/2	53 24
Sample ID	Lab Numbe	Asbestos r Type	Percent in Layer	Asbestos Type	Percent in Layer	Asbestos Type	Percent in Layer
05B Layer: Dark Grey Cementitious Material Layer: Grey Cementitious Material Layer: Paint	12746880		ND ND ND				
Total Composite Values of Fibrous Com Cellulose (Trace)	ponents:	Asbestos (ND)					
05C Layer: Dark Grey Cementitious Material Layer: Grey Cementitious Material Layer: Paint	12746881		ND ND ND				
Total Composite Values of Fibrous Com Cellulose (Trace)	ponents:	Asbestos (ND)					
06A Layer: Grey Cementitious Material	12746882		ND				
Total Composite Values of Fibrous Com Cellulose (Trace)	ponents:	Asbestos (ND)					
06B Layer: Grey Cementitious Material	12746883		ND				
Total Composite Values of Fibrous Com Cellulose (Trace)	ponents:	Asbestos (ND)					
06C Layer: Beige Cementitious Material Layer: Paint	12746884		ND ND				
Total Composite Values of Fibrous Com Cellulose (Trace)	ponents:	Asbestos (ND)					
08A Layer: Black Mastic	12746885		ND				
Total Composite Values of Fibrous Com Cellulose (Trace)	ponents:	Asbestos (ND)					
08B Layer: Black Mastic	12746886	Chrysotile	Trace				
Total Composite Values of Fibrous Com Cellulose (Trace) Comment: This comment applies to the	ponents: Black Masti	Asbestos (Trac	e) ent material for	r additional a	analyses.		
08C Layer: Black Mastic	12746887		ND				
Total Composite Values of Fibrous Com Cellulose (Trace)	ponents:	Asbestos (ND)					
09A Layer: Brown Non-Fibrous Material Layer: Paint	12746888		ND ND				
Total Composite Values of Fibrous Com Cellulose (Trace)	ponents:	Asbestos (ND)					

Client Name: Terracon - Concord					Report Numb Date Printed:	er: B36015 05/17/2	53 24
Sample ID	Lab Number	Asbestos r Type	Percent in Layer	Asbestos Type	Percent in Layer	Asbestos Type	Percent in Layer
09B Layer: Brown Non-Fibrous Material Layer: Paint	12746889		ND ND				
Total Composite Values of Fibrous Con Cellulose (Trace)	ponents:	Asbestos (ND)					
10A Layer: Black Asphalt with Debris	12746890		ND				
Total Composite Values of Fibrous Com Cellulose (Trace)	ponents:	Asbestos (ND)					
10B Layer: Black Asphalt	12746891		ND				
Total Composite Values of Fibrous Com Cellulose (Trace)	ponents:	Asbestos (ND)					
10C Layer: Black Asphalt	12746892		ND				
Total Composite Values of Fibrous Con Cellulose (Trace)	ponents:	Asbestos (ND)					
11A Layer: Black Felt	12746893		ND				
Total Composite Values of Fibrous ComCellulose (50 %)Synthetic (15 %)	ponents:	Asbestos (ND)					
11B Layer: Black Felt	12746894		ND				
Total Composite Values of Fibrous ComCellulose (50 %)Synthetic (15 %)	ponents:	Asbestos (ND)					
12A Layer: Grey Cementitious Material Layer: Paints	12746895		ND ND				
Total Composite Values of Fibrous Com Cellulose (Trace)	ponents:	Asbestos (ND)					
12B Layer: Grey Cementitious Material Layer: Paints	12746896		ND ND				
Total Composite Values of Fibrous Com Cellulose (Trace)	ponents:	Asbestos (ND)					
12C Layer: Grey Cementitious Material Layer: Paints	12746897		ND ND				
Total Composite Values of Fibrous Com Cellulose (Trace)	ponents:	Asbestos (ND)					

Client Name: Terracon - Concord					Report Numbe Date Printed:	r: B360153 05/17/24	
Sample ID	Lab Number	Asbestos Type	Percent in Layer	Asbestos Type	Percent in Layer	Asbestos Type	Percent in Layer
14A Layer: Tan Non-Fibrous Material Layer: Paint	12746898		ND ND				
Total Composite Values of Fibrous Con Cellulose (Trace)	nponents: As	sbestos (ND)					
14B Layer: Tan Non-Fibrous Material Layer: Paint	12746899		ND ND				
Total Composite Values of Fibrous Con Cellulose (Trace)	nponents: As	sbestos (ND)					

Em Cenelis

Eric Cerecedo, Laboratory Supervisor, Carson Laboratory

Note: Limit of Quantification ('LOQ') = 1%. 'Trace' denotes the presence of asbestos below the LOQ. 'ND' = 'None Detected'.

Analytical results and reports are generated by SGS Forensic Laboratories (SGSFL) at the request of and for the exclusive use of the person or entity (client) named on such report. Results, reports or copies of same will not be released by SGSFL to any third party without prior written request from client. This report applies only to the sample(s) tested. Supporting laboratory documentation is available upon request. This report must not be reproduced except in full, unless approved by SGSFL. The client is solely responsible for the use and interpretation of test results and reports requested from SGSFL. This report must not be used by the client to claim product endorsement by NVLAP or any other agency of the U.S. Government. SGSFL is not able to assess the degree of hazard resulting from materials analyzed. SGS Forensic Laboratories reserves the right to dispose of all samples after a period of thirty (30) days, according to all state and federal guidelines, unless otherwise specified. All samples were received in acceptable condition unless otherwise noted.

04 С

Material Location:

Women's Restroom at Accessible Stall

20 Concord Avenue Suite 450, Concord, Californai									nai	ACM BULK SAMPLE DATA SHEET * PLM Analysis Page 1 of 4					EET
PM- W wmfrie com	. Friesze eszell@t	ell erracon.	x	PM <u>Kar</u>	- K. Sc in@rg	Schroeter PM - K. Pilgrim rgaenv.com Ken@rgaenv.com		Stop Analysis at First Positive X Analyze All Samples							
PM - T. Kattchee PM - B, Gils Tedd@rgaenv.com Bob@rgaenv.com			<u>om</u>	F	PM - M. Bryant nariin bryant@rgaenv.com	Point Count Analysis (400-point)									
Project	Name	e/Addre	ss	San I	Bruno	Mo	untain Park	555	5 Guadalupe Cany	on Parkway	, Brisban	e CA			
Project	Numb	ber		R1247	013		Sampled	Ву	Karin Schroeter		Samplin	ng Date	May	14, 2024	1.00
Labora	tory	SGS F	orensi	nsic X Other Turn		Turn Around Time	n Around Time		X	Other (Specify)	P			
HM# Sample	01 ID	Ma	Sample Location & Material Location									_			
	01 /	A Res	Restroom Building – SE Corner of Upper Roof												
	01	B Res	Restroom Building – E Perimeter												
	01 (C Res	Restroom Building – NE Corner of Upper Roof												
Material L	ocation:														
HM#	02	Ma	terial	Des	riptio	on:	Pressur	re G	irout Inside Block						
Sample	ID	Sar	nple L	ocat	ion &	Mat	terial Locat	ion	1	Quantity:					_
	02	A Res	troon	n Bui	lding	-NV	V Corner of	Bui	ilding at 4.5'						
	02	B Res	troon	n Bui	lding	– We	est Elevation	n at	Foundation						
Material	02 Locatio	C Res	troon	n Bui	lding	- Sto	orage Room	/Ele	ectrical Panel Roor	n at Wall P	enetratio	'n	-		-
HM#	03	Ma	terial	Dese	riptic	on:	Cerami	ic Ti	le Wainscoting, Gi	out, Morta	ar Bed – L	ight Gre	en		
Sample	ID	Sar	nple L	ocat	ions					Quantity:					
11.1	03	A Me	n's Re	stro	om at	Low	er End of D	ivid	ler Wall						
	03	B Me	n's Re	stro	om at	Nor	th end of N	W	Wall						
	03	C Wo	men's	s Res	troon	n – A	ccessible St	tall					-		
Material	Locatio	n:													
HM#	04	Ma	terial	Des	riptie	on:	Block F	iller	r at Divider Walls i	n Restroon	15				
			Sample Location & Material Location						Quantity:						

	NAME:	SIGNATURE:	COMPANY:	DATE:
Relinquished By:	Karin Schroeter	Jana Schraute	Terracon Consultants	May 14, 2024
Received By:	VIUNUW (UNINEIDI	Amme (nothe-	Cas	5-1-24 9.
Relinquished By:		V	1	100 State (1997)
Received By:	RECEVED	2		
	MAY 1 9 2024	nelg: J	ade ceuks 5 2:04	214124
	BY:00 1030			

20 Concord	DCC Avenue S	DN uite 450, Concord, Ca	alifornai	ACM BULK SAMPLE DATA SHEET						
PM- W. Friesze wmfrieszell@te	rracon. X	PM - K. Schroeter Karin@rgaenv.com	PM - K. Pilgrim Ken@rgaenv.com	Stop	Stop Analysis at First Positive					
PM - T. Kattche Tedd@rgaenv.o	e <u>om</u>	PM - B. Gils Bob@rgaenv.com	PM - M. Bryant marlin bryant@rgaenv.com	Poi	Point Count Analysis (400-point)					
Project Name	/Address	San Bruno Mountain Pa	rk 555 Guadalupe Cany	on Parkwa	y, Brisbane	e CA				
Project Numb	er	R1247013 Sample	d By Karin Schroeter		Samplin	g Date	May 14, 2024	-		
Laboratory	Micro	X Other	Turn Around Time	0 - 1	5 Day	x	Other (Specify)	1		
FAX OR E-M	AIL REPORT	TTO: SEE ABOVE PROJEC	CT MANAGER (PM) *	-						
HM# 05	Materia	al Description: CMU	I/Block - Grout							
Sample ID	Sample	Location & Material Loc	ation	Quantity:				_		
05 A	Restroo	m Building – Storage Roo	om/Electrical Panel Roo	m						
05 B	Restroo	Restroom Building – NW Corner of Building								
05 C	Restroo	m Building – SW Corner								
Material Location:										
HM# 06	Materia	al Description: Conc	crete Floor	Sec. 19. 19						
Sample ID	Sample	Location & Material Loc	ation	Quantity:	-					
06 4	Restroo	om Building – Concrete Pa	ad in Men's Room							
06 E	Restroo	om Building – SE Corner o	f Pad							
06 (Restroo	om Building – Storage Roo	om/Electrical Panel Room	m						
Material Location	:									
HM# 07	Materia	al Description: Mirr	or Mastic							
Sample ID	Sample	Locations		Quantity:						
Assumed	Assume	ed								
Material Location										
HM# 08	Materia	al Description: Sink	Undercoating							
Sample ID	Sample	Location & Material Loc	ation	Quantity:						
08 /	Restroo	m Building – Men's at Ur	inal							
08 6	Restroo	om Building – Men's Singl	e User Restroom							
08 (Restroo	m Building – Women's R	estroom Eastern Stall							
1 COLT - 6 6 2 10 10 10	-									

NAME:	SIGNATURE:	COMPANY:	DATE:	
Karin Schroeter	Bann Schreete	Terracon Consultants	May 14, 2024	
VICTIMINA (UNIVERIA	frome anto	595	5-17-24 9	
V	V			
CAL SD3				
MAY 1 5 2024				
	NAME: Karin Schroeter	NAME: SIGNATURE: Karin Schroeter MAX 1 2 2024	NAME: SIGNATURE: COMPANY: Karin Schroeter MAY 1 2 2024	

BY. JE 1030

20 Concord /	Avenue Su	JIA uite 450, Concord, Ci	alifornai	ACM BULK SAM	PLE DATA SHEET Page <u>3 of 4</u>			
PM- W. Frieszel wmfrieszell@te com	rracon. X	PM - K. Schroeter Karin@rgaenv.com	PM - K. Pilgrim Ken@rgaenv.com	 Stop Analysis at First Positiv X Analyze All Samples 	re			
PM - T. Kattchee Tedd@rgaenv.c	e om	PM - B. Gils Bob@rgaenv.com	PM - M. Bryant marlin bryant@rgaenv.com	Point Count Analysis (400-point)				
Project Name,	Address	San Bruno Mountain Pa	ark 555 Guadalupe Canyo	n Parkway, Brisbane CA				
Project Numb	er	R1247013 Sample	ed By Karin Schroeter	Sampling Date	May 14, 2024			
Laboratory	Micro	X Other	Turn Around Time	Time 5 Day X Other (Specify)				
*FAX OR E-M	AIL REPORT	TO: SEE ABOVE PROJE	CT MANAGER (PM) ***					
HM# 09	Materia	I Description: Caul	king Bead at Vent Above	Storeroom				
Sample ID	Sample	Location & Material Loc	cation	Quantity:				
09 A	Restroor	n Building – Storage Roo	om/Electrical Panel Room					
09 B	Restroor	n Building – NW Corner	of Building					
Waterial Location:	i, i							
HM# 10	Materia	I Description: Aspt	nalt Floor					
Sample ID	Sample	Location & Material Loc	cation	Quantity:				
10 A	Restroor	n Building – Pad Outside	e of Women's Room					
10 B	Restroor	n Building – Pad at Bike	Repair Area					
10 (Restroor	n Building – Pad at Sign	Post					
material Location								
HM# 11	Materia	I Description: Roof	fing Felt Under Metal Roo	of				
Sample ID	Sample	Locations		Quantity:				
11 A	Restroom	n Building – NE Corner o	of Building at Overhang					
11 P	Bastroo	I building - St Corner o	or Building at Overhang					
11 E	B Restroom							
11 E Material Location 4M# 12	Restroom	I Description: Colu	umn Concrete					
11 E Material Location HM# 12 Sample ID	B Restroom	I Description: Colu	imn Concrete	Quantity:				
11 E Material Location HM# 12 Sample ID 12 A	B Restroom	I Description: Colu Location & Material Loc n Building – SE Corner C	umn Concrete cation	Quantity:				
11 E Material Location HM# 12 Sample ID 12 A 12 E	B Restroom Materia Sample Restroom Restroom	I Description: Colu Location & Material Loc m Building – SE Corner C m Building – NE Corner (imn Concrete cation Column Column	Quantity:				
11 E Material Location HM# 12 Sample ID 12 A 12 E 12 C	B Restroom Materia Sample Restroom Restroom Restroom	I Description: Colu Location & Material Loc n Building – SE Corner C n Building – NE Corner C n Building – SE Corner C	imn Concrete cation Column Column Column	Quantity:				
11 E Material Location HM# 12 Sample ID 12	B Restroom Materia Sample A Restroom B Restroom C Restroom	I Description: Colu Location & Material Loc m Building – SE Corner C m Building – NE Corner C m Building – SE Corner C	imn Concrete cation Column Column Column	Quantity:				
11 E Material Location HM# 12 Sample ID 12 12 12 12 12 12 12 12 12 12 12 12 12 13	B Restroom Materia Sample Restroom Restroom Materia	I Description: Colu Location & Material Loc m Building – SE Corner C m Building – NE Corner C m Building – SE Corner C I Description: Skyli	imn Concrete cation Column Column Column ight Putty	Quantity:				
11 E Material Location HM# 12 Sample ID 12 A 12 C Material Location HM# 13	B Restroom Materia Sample A Restroom B Restroom C Restroom Materia Materia	I Description: Colu Location & Material Loo m Building – SE Corner C m Building – NE Corner C m Building – SE Corner C I Description: Skyli e	imn Concrete cation Column Column Column ight Putty	Quantity:				
11 E Material Location HM# 12 Sample ID 12 A 12 C 12 C Material Location HM# 13 Assume	B Restroom Materia Sample Restroom Restroom Restroom Materia Materia	I Description: Colu Location & Material Loo m Building – SE Corner C m Building – NE Corner C m Building – SE Corner C I Description: Skyli e NAME:	imn Concrete cation Column Column Column ight Putty SIGNATURE:	Quantity:	DATE:			
11 E Material Location HM# 12 Sample ID 12 F 12 C Material Location HM# 13 Assume Relinquished B	By: Ka	I Description: Colu Location & Material Loo m Building – SE Corner C m Building – NE Corner C m Building – SE Corner C I Description: Skyli IE NAME: min Schroeter	imn Concrete cation Column Column Column ight Putty SIGNATURE:	Quantity: COMPANY:	DATE: May 14, 2024			
11 E Material Location HM# 12 Sample ID 12 12 E 12 E 12 C Material Location HM# 13 Assume Relinquished B Received Burn	Restroom Materia Sample A Restroom Restroom Restroom Restroom Materia Materia A	I Description: Colu Location & Material Loo m Building – SE Corner (m Building – NE Corner (m Building – SE Corner (I Description: Skyli I NAME: NAME: IT IN IN (A) (I M V) M	imn Concrete cation Column Column Column Column ight Putty SIGNATURE:	Quantity: COMPANY: Terracon Consultants	DATE: May 14, 2024			

Rela:

Vs silulzy

W

ac

<u>r</u>	RECTYNOR
	MAY 1 0 2024
	BYSC LOSD

Received By:

PM- W. Friesze wmfrieszell@te	ll erracon.	K PN	1 - K. Schr rin@rgae	oeter nv.com	PM - K. Pilgrim Ken@rgaenv.com	* PLN 	* PLM Analysis Page <u>4 of 4</u> Stop Analysis at First Positive			
PM - T. Kattche Tedd@rgaenv.c	e com	PN Bo	1 - B. Gils b@rgaen	v.com	PM - M. Bryant marlin bryant@rgaenv.o		oint Count An	alysis (40	00-point)	
Project Name	/Address	San	Bruno M	Mountain Park	555 Guadalupe Ca	nyon Parkw	ay, Brisban	e CA		
Project Numb	er	R124	7013	Sampled	By Karin Schroe	ter	Samplin	ng Date	May 14, 2024	
Laboratory	Micro		XC	Other	Turn Around Ti	me	5 Day	x	Other (Specify)	
Sample ID 14 A 14 B	Sampl Restro	e Local om Bui om Bui	ilding – ilding –	Material Locat Women's RR I Men's RR Doo	ion Door r	Quantity				
Material Location:		ial Dec								_
Sample ID	Sampl	e Local	tion & M	ı. Material Locat	ion	Quantity	:			
Material Location										
HM#	Mater	ial Des	cription	1:						
Sample ID	Sampl	e Locat	tions			Quantity				
Material Location										
HM#	Mater	ial Des	cription	n:						
Sample ID	Sampl	e Local	tion & N	Material Locat	ion	Quantity				
		-								-
Material Location	-									
Material Location										

	NAME:	SIGNATORE.	COMPANY:	DATE:
Relinquished By:	Karin Schroeter	Barin Schreets	Terracon Consultants	May 14, 2024
Received By:	samme contrerat	Imme antes	SAS	9:300mit
Relinquished By:		Var min		
Received By:	RECTX 8043			
	MAY 1 0 2071 BY JC 1080	Rila	1. Jack Could 3101	us s/14/21 ppm

Pre-demolition Hazardous Materials Survey San Bruno Mountain Park - Day Use Area | Brisbane, CA May 24, 2024 | Terracon Report No. R1247013



APPENDIX B

ASBESTOS ANALYTICAL LABORATORY DATA - NOA

Page 1 of 2

MICRO ANALYTICAL LABORATORIES, INC. **BULK ASBESTOS ANALYSIS - PLM ARB 435**

Micro Log In 314947

1023 Karin Schroeter Terracon Consultants, Inc. 1220 Concord Ave., Ste 450 Concord, CA 94520

PROJECT: JOB NO. R1247013 SAN BRUNO MOUNTAIN PARK **555 GUADALUPE CANYON PARKWAY BRISBANE, CA**

Total Samples 6

Date Sampled 05/14/2024 Date Received 05/14/2024 Date Analyzed 05/14/2024

	SAMPLE INFORMATION	ASBESTOS INFORMATION QUANTITY (AREA %) / TYPES / LAYERS / DISTINCT SAMPLES	DOMINANT OTHER MATERIALS
Client #: Micro #: 314947	S-1 7-01 Analyst: AF OSITE TO 30" DEPTH	ND	2 % CELLULOSE
SAN BRUNO MOUN COORDINATES: 37 Asb. / Total Pts.	ATAIN PARK - .697313, -122.433745 Matrix Removed Sensitivity		Matrix CLAY Type: CARBONATE OPAQUES
Client #:	S-2	ND	2 % CELLULOSE
SITE SOILS- COMP SAN BRUNO MOUN COORDINATES: 37 Asb. / Total Pts.	OSITE TO 30" DEPTH ITAIN PARK - .697449, -122.433693 Matrix Removed Sensitivity		Matrix CLAY Type: CARBONATE OPAQUES
Client #:	0% 0.250% S-3	ND	2 % CELLULOSE
Micro #: 314947 siTE SOILS- COMP SAN BRUNO MOUN COORDINATES: 37 Asb. / Total Pts.	7-03 Analyst: AF OSITE TO 30" DEPTH ITAIN PARK - 6897542, -122.434075 Matrix Removed Sensitivity		Matrix CLAY Type: CARBONATE OPAQUES
Client #:	S-4	ND	2 % CELLULOSE
Micro #: 314947 SITE SOILS- COMPI SAN BRUNO MOUN COORDINATES: 37. Asb. / Total Pts. 0 / 400	7-04 Analyst: AF OSITE TO 30" DEPTH ITAIN PARK - 697765, -122.433906 Matrix Removed Sensitivity 0% 0.250%		Matrix CLAY Type: CARBONATE OPAQUES
Client #: Micro #:314947	S-5 7-05 Analyst: AF	ND	2 % CELLULOSE
SITE SOILS- COMPO SAN BRUNO MOUN COORDINATES: 37. Asb. / Total Pts. 0 / 400	05/TE TO 30" DEPTH TAIN PARK - 697301, -122.432968 Matrix Removed Sensitivity 0% 0.250%		Matrix CLAY Type: CARBONATE OPAQUES

Technical Supervisor:

Baojia Ke, Ph.D.

5/14/2024 Date Reported

Analyses use Polarized Light Microscopy (PLM), Micro Analytical SOP PLM-101 for building materials. Basic techniques follow California ARB 435 (1991) for applicable soil, rock, or aggregate samples. NOTES: Weight % cannot be determined by this method. Asbestos fibers with diameter below ~1 µm may not be detected by PLM. Only dominant non-asbestos materials are indicated. This report must not be interpreted as a conclusive identification of non-asbestos (fibrous or not). Preparation (all samples): grinding, milling; teasing bundles apart; drying, if needed, by hotplate. Various sample interferences may prevent detection of small asbestos fibers, and hinder determination of some optical properties. Notes are made if point counting is used; otherwise, asbestos is quantified by calibrated visual estimation. Detection limit is material dependent. Detection of absestos traces (<0.25%) may not be field forms. Laboratory sample descriptions may differ from descriptions given by the client. Quality Control (QC): all results have been determined to be within acceptance limits prior to reporting. Samples that were received are denoted by two sets of analyst initials. Unless otherwise stated in this report, all samples were received in full, without the approval of Micro Analytical Laboratories, Inc., and pertains only to the samples analyzed as received. ND = NO ASBESTOS DETECTED.

Page 2 of 2

MICRO ANALYTICAL LABORATORIES, INC. BULK ASBESTOS ANALYSIS - PLM ARB 435

1023 Karin Schroeter Terracon Consultants, Inc. 1220 Concord Ave., Ste 450 Concord, CA 94520

PROJECT: JOB NO. R1247013 SAN BRUNO MOUNTAIN PARK **555 GUADALUPE CANYON PARKWAY BRISBANE, CA**

	21	
£	47	ß
	:स्	Ľ

Micro Log In 314947 **Total Samples** 6 Date Sampled 05/14/2024 Date Received 05/14/2024 Date Analyzed 05/14/2024

SAMPLE INFORMATION	ASBESTOS INFORMATION QUANTITY (AREA %) / TYPES / LAYERS / DISTINCT SAMPLES	DOMINANT OTHER MATERIALS
Client #: S-6	ND	2 % CELLULOSE
Micro #: 314947-06 Analyst: AF		
SITE SOLLS- COMPOSITE TO 30" DEPTH SAN BRUNO MOUNTAIN PARK - COORDINATES: 37.697456, -122.432872		Matrix CLAY Type: CARBONATE
Asb. / Total Pts. Matrix Removed Sensitivity 0 / 400 0% 0.250%		OPAQUES

Technical Supervisor 5/14/2024 Date Reported Baojia Ke, Ph.D. Dr

Analyses use Polarized Light Microscopy (PLM), Micro Analytical SOP PLM-101 for building materials. Basic techniques follow California ARB 435 (1991) for applicable soil, rock, or aggregate samples. NOTES: Weight % cannot be determined by this method. Asbestos fibers with diameter below ~1 µm may not be detected by PLM. Only dominant non-asbestos raterials are indicated. This report must not be interpreted as a conclusive identification of non-asbestos (fibrous or not). Preparation (all samples): grinding, milling; teasing bundles apart; drying, if needed, by hotplate. Various sample interferences may prevent detection of small asbestos fibers, and hinder determination of some optical properties. Notes are made if point counting is used; otherwise, asbestos is quantified by calibrated visual estimation. Detection limit is material dependent. Detection of asbestos races (<0.25%) may not be reliable or reproducible by PLM. Lower quantitation limit (reporting limit) of this method is 0.25%. Clients are solely responsible for identification and description of bulk materials listed on field forms. Laboratory sample descriptions may differ from descriptions given by the client. Quality Control (QC): all results have been determined to be within acceptance limits prior to reporting. Samples that were reanalyzed are denoted by two sets of analyst initials. Unless otherwise stated in this report, all samples were received in acceptable condition for analysis. This report must not be used to claim product endorsement by NIST or any U.S. Government agency. This report shall not be reproduced except in full, without the approval of Micro Analytical Laboratories, Inc., and pertains only to the samples analyzed as received. ND = NO ASBESTOS DETECTED.

220 Concord Av	acon enue, Suite 450, Concord, California M - K. Schroeter Karin@rgaenv.com	ACM BULK SAMPLE DATA SHEET (ARB 435 [4PP] * PLM Analysis at First Positive
		Point Count Analysis (400-point)
Dupingt Name / A		
Terracon Droject	Number R1247013 Sampled Br. W Kitigen	arkway, Brisbane, California
Laboratory	Adicro X Other Turn Around Time	5 Days X Other (Specify)
***FAX OR E-M	AIL REPORT TO: SEE ABOVE PROJECT MANAGE	<i>R</i> (<i>PM</i>) ***
HM# 1	Material Description: Site Soils - Composite	to 30" Denth
Sample ID	Sample Location & Material Location	Quantity:
S-1	San Bruno Mountain Park - Coordinates: 37.	697313, -122,433745
S-2	San Bruno Mountain Park - Coordinates: 37	697449, -122,433893
S-3	San Bruno Mountain Park - Coordinates: 37	697542 -122 434075
S-4	San Bruno Mountain Park - Coordinates: 37	697765 -122 433906
S-5	San Bruno Mountain Park - Coordinates: 37	697301 -122 432968
S-6	San Bruno Mountain Park - Coordinates: 37	697456 -122 432872
Material Location:	San Stand Houmann Funt Coordinates. 51.	07,150, 122.152072
HM#	Material Description:	
Sample ID	Sample Location & Material Location	Quantity:
А		
В		
C		
Material Location:		
HM#	Material Description:	
Sample ID	Sample Locations (Quantity:
Material Location:		
HM#	Material Description:	
Sample ID	Sample Location & Material Location	Quantity:
А		
В		
C		
Material Location:		
HM#	Material Description:	
Sample ID	Sample Location & Material Location (Quantity:
A		
В		
С		

-	NAME:	SIGNATURE:	COMPANY:	DATE:
Relinquished By:	William Frieszell	Will Fright	Terracon Consultants	May 14, 2024
Received By:	MS	- All		5/14/24 1310
Relinquished By:	mto.			on yet ibit
Received By:				

Pre-demolition Hazardous Materials Survey San Bruno Mountain Park - Day Use Area | Brisbane, CA May 24, 2024 | Terracon Report No. R1247013



APPENDIX C

LEAD ANALYTICAL LABORATORY DATA



Metals Analysis of Bulks - TTLC (AIHA-LAP, LLC Accreditation, Lab ID #101762)

Terracon - Concord					Client ID:	L1969				
Karin Schroeter					Report Num	ber: M260006				
1220 Concord Ave					Date Receive	d: 05/15/24				
suite 450					Date Analyzo	ed: 05/17/24				
Concord, CA 94520					Date Printed	: 05/17/24				
					First Report	ed: 05/17/24				
Job ID / Site: R1247013 - San Bruno Mountain Park, 555 Guadalupe Parkway Road, Brisbane, SGSFL Job ID: L1969 CA										
CA										
CA Date(s) Collected: 05/12/2024	4				Total Sample	es Submitted: 1				
CA Date(s) Collected: 05/12/2024	4				Total Sample Total Sample	es Submitted: 1 es Analyzed: 1				
CA Date(s) Collected: 05/12/2024 Sample Number	4 Lab Number	Analyte	Result	Result Units	Total Sample Total Sample Reporting Limit*	es Submitted: 1 es Analyzed: 1 Method Reference				
CA Date(s) Collected: 05/12/2024 Sample Number PB-01	4 Lab Number 30939205	Analyte Pb	Result 130	Result Units ppm	Total Sample Total Sample Reporting Limit* 6	es Submitted: 1 es Analyzed: 1 Method Reference EPA 3050B/7000B				

* The Reporting Limit represents the lowest amount of analyte that the laboratory can confidently detect in the sample, and is not a regulatory level. The Units for the Reporting Limit are the same as the Units for the Final Results.

Kevin Poon

Kevin Poon, Laboratory Supervisor, Hayward Laboratory

Analytical results and reports are generated by SGS at the request of and for the exclusive use of the person or entity (client) named on such report. Results, reports or copies of same will not be released by SGS to any third party without prior written request from client. This report applies only to the sample(s) tested. Supporting laboratory documentation is available upon request. This report must not be reproduced except in full, unless approved by SGS. The client is solely responsible for the use and interpretation of test results and reports requested from SGS. SGS is not able to assess the degree of hazard resulting from materials analyzed. SGS reserves the right to dispose of all samples after a period of thirty (30) days, according to all state and federal guidelines, unless otherwise specified. Any modifications that have been made to referenced test methods are documented in SGS Standard Operating Procedures Manual. Sample results have not been blank corrected. Quality control and sample receipt condition were acceptable unless otherwise noted.

Note* Sampling data used in this report was provided by the client as noted on the associated chain of custody form.



Metals Analysis of Paints (AIHA-LAP, LLC Accreditation, Lab ID #101762)

Terracon - Co	oncord					Client ID:	L1969			
Karin Schroet	ter					Report Num	ber: M260005			
1220 Concord	d Ave					Date Receive	ed: 05/15/24			
suite 450						Date Analyz	ed: 05/17/24			
Concord, CA	94520					Date Printed	1: 05/17/24			
						First Report	ed: 05/17/24			
Job ID / Site: R1247013 - San Bruno Mountain Park, 555 Guadalupe Parkway Road, Brisbane, SGSFL Job ID: L1969 CA										
Date(s) Colle	cted: 05/12/2024	4				Total Sampl	es Submitted: 2			
						Total Sampl	es Analyzed: 2			
Sample Number Lab Number		Lab Number	Analyte	Result	Result Units	Reporting Limit*	Method Reference			
PB-02		30939206	Pb	< 60	ppm	60	EPA 3050B/7000B			
Comment:	RESTROOM E	XT N EXT WALL								
PB-03		30939207	Pb	< 70	ppm	70	EPA 3050B/7000B			
		50757201		10	PP	10	LITI SOSOD, TOOOD			

* The Reporting Limit represents the lowest amount of analyte that the laboratory can confidently detect in the sample, and is not a regulatory level. The Units for the Reporting Limit are the same as the Units for the Final Results.

Kevin Poon

Kevin Poon, Laboratory Supervisor, Hayward Laboratory

Analytical results and reports are generated by SGS at the request of and for the exclusive use of the person or entity (client) named on such report. Results, reports or copies of same will not be released by SGS to any third party without prior written request from client. This report applies only to the sample(s) tested. Supporting laboratory documentation is available upon request. This report must not be reproduced except in full, unless approved by SGS. The client is solely responsible for the use and interpretation of test results and reports requested from SGS. SGS is not able to assess the degree of hazard resulting from materials analyzed. SGS reserves the right to dispose of all samples after a period of thirty (30) days, according to all state and federal guidelines, unless otherwise specified. Any modifications that have been made to referenced test methods are documented in SGS Standard Operating Procedures Manual. Sample results have not been blank corrected. Quality control and sample receipt condition were acceptable unless otherwise noted.

Note* Sampling data used in this report was provided by the client as noted on the associated chain of custody form.

Terracon 1220 Concord Avenue, Suite 450, Concord, CA 94520						LEAD SAMPLE DATA SHEET * Lead Analysis _X_ Flame AA (EPA 7420)						
	<u>Steff@rgaenv.c</u>	om	^	Karin@rgaen	v.com		Ken@rgaen	v.com			Other	
	Fax: 510 899.70	051	\rightarrow	Fax: 510.899.	7063 701	┟──┟╵	Fax: 510.899	9.7053				Page 1 of 1
	Tedd@rgaenv.c	tom		wmfrieszell@	terracon.							
,	Fax: 510.899.70	070		com						,		
P	Project Name	/Address	Sa	an Bruno M	lountain	Park 55	5 Guadalu	pe Parkwa	ay Road	d, Brisbane, CA	1	
P	roject Numb	er	R1	247013	Samp	oled By	Karin S	chr o eter		Sampling Da	ite May 12, 2	024
L	aboratory	SGS Fore	nsic	Ot	her		Turn Arou	und Time		2 day ×	Other (5pecil	īγ)
Se	ample ID	Paint	Desc	cription a	nd Sam	iple Lo	cation		Pe	r Kavin s[1s]	Jace Jul 1:34	Condition
	Dh 01	Col	or:	Light	Grey	Subs	strate:	Ceram	nic	Component:	Wall	
	PD-01	Bid	lg:	Rest	room	U	nit:	Interi	ог	Room:	Men's	
		Paint Color:		: w	White Substrate:		strate:	СМ		Component:	Wall	
-	Pb-02	Bld	lg:	Rest	room	Unit:		Exteri	or	Room:	North Exterior Wall	
		Paint (Color	: Grey		Subs	strate:	Woo	d	Component:	Door	
	Pb-3	Bidg:		Rest	room	U	nit:	Exteri	or	Room:	Sliding Door to Electrical Room	
		Paint (Color	:		Subs	strate:			Component:		
		Bldg:				U	nit:			Room:		
		Paint Color:		:		Subs	strate:			Component:		
		Bld	lg:			U	nit:			Room:		
		Paint (Color	:		Subs	strate:			Component:		
		Bld	lg:			U	nit:			Room:		

	NAME:	SIGNATURE:	COMPANY:	DATE:
Relinquished By:	Karin Schroeter	Kavin - Schreete	Terracon Consultants	May 14, 2024
Received By:				
Relinquished By:				
Received By:				
R	ECELVED	,		•
	MAY 1 5 2024			
BY	JC 1020			

Pre-demolition Hazardous Materials Survey San Bruno Mountain Park - Day Use Area | Brisbane, CA May 24, 2024 | Terracon Report No. R1247013



APPENDIX D

PCBS ANALYTICAL LABORATORY DATA

SGS EXCELCHEM Laboratories, Inc.

1135 W Sunset Boulevard Suite A Rocklin, CA 95765 Phone# 916-543-4445

22 May 2024 Nicole Adams SGS Carson 20535 Belshaw Ave Carson, CA 90746 RE: L1969

Work order number:2405078

Enclosed are the results of analyses for samples received by the laboratory on 05/16/24 09:52. All Quality Control results are within acceptable limits except where noted as a case narrative. If you have any questions concerning this report, please feel free to contact the laboratory.

Sincerely,

ig Sel

Doug Selby, Technical Director



ELAP Certificate No. : 2119

SGS Carson	Project:	L1969	
20535 Belshaw Ave	Project Number:	L1969	Date Reported:
Carson, CA 90746	Project Manager:	Nicole Adams	05/22/24 14:24

ANALYTICAL REPORT FOR SAMPLES

Sample ID	Laboratory ID	Matrix	Date Sampled	Date Received
PCB 1	2405078-01	Paint Chips	05/14/24 08:00	05/16/24 09:52
PCB 2	2405078-02	Paint Chips	05/14/24 08:00	05/16/24 09:52
PCB 3	2405078-03	Paint Chips	05/14/24 08:00	05/16/24 09:52
PCB 4	2405078-04	Paint Chips	05/14/24 08:00	05/16/24 09:52
PCB 5	2405078-05	Paint Chips	05/14/24 08:00	05/16/24 09:52
PCB 6	2405078-06	Paint Chips	05/14/24 08:00	05/16/24 09:52

SGS Excelchem Laboratories, Inc.

1 el g 2 L

The results in this report apply to the samples analyzed in accordance with the chain of custody document. This analytical report must be reproduced in its entirety.

SGS Carson	Project:	L1969	
20535 Belshaw Ave	Project Number:	L1969	Date Reported:
Carson, CA 90746	Project Manager:	Nicole Adams	05/22/24 14:24

PCB 1 2405078-01 (Paint Chips)

Analyte	Result	Reporting Limit	Units	Batch	Date Prepared	Date Analyzed	Method	Notes
PCBs by GC/ECD								
Aroclor 1016	ND	1.96	mg/kg	AHE0160	05/20/24	05/20/24	PCBs BY EPA 8082	C-03, C-04, R-02
Aroclor 1221	ND	1.96	"	"	"	"	"	C-03, C-04, R-02
Aroclor 1232	ND	1.96	"	"	"	"	"	C-03, C-04, R-02
Aroclor 1242	ND	1.96	"	"	"	"	"	C-03, C-04, R-02
Aroclor 1248	ND	1.96	"	"	"	"	"	C-03, C-04, R-02
Aroclor 1254	ND	1.96	"	"	"	"	"	C-03, C-04, R-02
Aroclor 1260	ND	1.96	"	"	"	"	"	C-03, C-04, R-02
Surrogate: Decachlorobiphenyl	82.0 %	% Recovery Limits		60-140			"	C-03, C-04
Surrogate: Tetrachloro-meta-xylene	84.9 %	% Recovery Limits		60-140			"	C-03, C-04

SGS Excelchem Laboratories, Inc.

ug - Sel /

The results in this report apply to the samples analyzed in accordance with the chain of custody document. This analytical report must be reproduced in its entirety.

SGS Carson	Project:	L1969	
20535 Belshaw Ave	Project Number:	L1969	Date Reported:
Carson, CA 90746	Project Manager:	Nicole Adams	05/22/24 14:24

PCB 2 2405078-02 (Paint Chips)

Analyte	Result	Reporting Limit	Units	Batch	Date Prepared	Date Analyzed	Method	Notes
PCBs by GC/ECD								
Aroclor 1016	ND	1.85	mg/kg	AHE0160	05/20/24	05/20/24	PCBs BY EPA 8082	C-03, C-04, R-02
Aroclor 1221	ND	1.85	"	"	"	"	"	C-03, C-04, R-02
Aroclor 1232	ND	1.85	"	"	"	"	"	C-03, C-04, R-02
Aroclor 1242	ND	1.85	"	"	"	"	"	C-03, C-04, R-02
Aroclor 1248	ND	1.85	"	"	"	"	"	C-03, C-04, R-02
Aroclor 1254	ND	1.85	"	"	"	"	"	C-03, C-04, R-02
Aroclor 1260	ND	1.85	"	"	"	"	"	C-03, C-04, R-02
Surrogate: Decachlorobiphenyl	72.7 %	% Recovery Limits		60-140			"	C-03, C-04
Surrogate: Tetrachloro-meta-xylene	74.9 %	% Recovery Limits		60-140			"	C-03, C-04

SGS Excelchem Laboratories, Inc.

eig / el

The results in this report apply to the samples analyzed in accordance with the chain of custody document. This analytical report must be reproduced in its entirety.

SGS Carson	Project:	L1969	
20535 Belshaw Ave	Project Number:	L1969	Date Reported:
Carson, CA 90746	Project Manager:	Nicole Adams	05/22/24 14:24

PCB 3 2405078-03 (Paint Chips)

Analyte	Result	Reporting Limit	Units	Batch	Date Prepared	Date Analyzed	Method	Notes
PCBs by GC/ECD								
Aroclor 1016	ND	1.92	mg/kg	AHE0160	05/20/24	05/20/24	PCBs BY EPA 8082	C-03, C-04, R-02
Aroclor 1221	ND	1.92	"	"	"	"	"	C-03, C-04, R-02
Aroclor 1232	ND	1.92	"	"	"	"	'n	C-03, C-04, R-02
Aroclor 1242	ND	1.92	"	"	"	"	"	C-03, C-04, R-02
Aroclor 1248	ND	1.92	"	"	"	"	"	C-03, C-04, R-02
Aroclor 1254	ND	1.92	"	"	"	"	"	C-03, C-04, R-02
Aroclor 1260	ND	1.92	"	"	"	"	"	C-03, C-04, R-02
Surrogate: Decachlorobiphenyl	77.2 %	% Recovery Limits		60-140			"	C-03, C-04
Surrogate: Tetrachloro-meta-xylene	79.2 %	% Recovery Limits		60-140			"	C-03, C-04

SGS Excelchem Laboratories, Inc.

eig _ el

The results in this report apply to the samples analyzed in accordance with the chain of custody document. This analytical report must be reproduced in its entirety.

SGS Carson	Project:	L1969	
20535 Belshaw Ave	Project Number:	L1969	Date Reported:
Carson, CA 90746	Project Manager:	Nicole Adams	05/22/24 14:24

PCB 4 2405078-04 (Paint Chips)

Analyte	Result	Reporting Limit	Units	Batch	Date Prepared	Date Analyzed	Method	Notes
PCBs by GC/ECD								
Aroclor 1016	ND	1.89	mg/kg	AHE0160	05/20/24	05/20/24	PCBs BY EPA 8082	C-03, C-04, R-02
Aroclor 1221	ND	1.89	"	"	"	"	"	C-03, C-04, R-02
Aroclor 1232	ND	1.89	"	"	"	"	"	C-03, C-04, R-02
Aroclor 1242	ND	1.89	"	"	"	"	"	C-03, C-04, R-02
Aroclor 1248	ND	1.89	"	"	"	"	"	C-03, C-04, R-02
Aroclor 1254	ND	1.89	"	"	"	"	"	C-03, C-04, R-02
Aroclor 1260	ND	1.89	"	"	"	"	"	C-03, C-04, R-02
Surrogate: Decachlorobiphenyl	81.0 %	% Recovery Limits		60-140			"	C-03, C-04
Surrogate: Tetrachloro-meta-xylene	76.3 %	% Recovery Limits		60-140			"	C-03, C-04

SGS Excelchem Laboratories, Inc.

eig _ el

The results in this report apply to the samples analyzed in accordance with the chain of custody document. This analytical report must be reproduced in its entirety.

SGS Carson	Project:	L1969	
20535 Belshaw Ave	Project Number:	L1969	Date Reported:
Carson, CA 90746	Project Manager:	Nicole Adams	05/22/24 14:24

PCB 5 2405078-05 (Paint Chips)

Analyte	Result	Reporting Limit	Units	Batch	Date Prepared	Date Analyzed	Method	Notes
					-	-		
PCBs by GC/ECD								
Aroclor 1016	ND	1.92	mg/kg	AHE0160	05/20/24	05/20/24	PCBs BY EPA 8082	C-03, C-04,
Aroclor 1221	ND	1.92	"	"	"	"	"	C-03, C-04, R-02
Aroclor 1232	ND	1.92	"	"	"	"	"	C-03, C-04, R-02
Aroclor 1242	ND	1.92	"	"	"	"	"	C-03, C-04, R-02
Aroclor 1248	ND	1.92	"	"	"	"	"	C-03, C-04, R-02
Aroclor 1254	ND	1.92	"	"	"	"	"	C-03, C-04, R-02
Aroclor 1260	ND	1.92	"	"	"	"	"	C-03, C-04, R-02
Surrogate: Decachlorobiphenyl	106 %	% Recovery Limits		60-140			"	C-03, C-04
Surrogate: Tetrachloro-meta-xylene	76.6 %	% Recovery Limits		60-140			"	C-03, C-04

SGS Excelchem Laboratories, Inc.

eig _ sel

The results in this report apply to the samples analyzed in accordance with the chain of custody document. This analytical report must be reproduced in its entirety.

SGS Carson	Project:	L1969	
20535 Belshaw Ave	Project Number:	L1969	Date Reported:
Carson, CA 90746	Project Manager:	Nicole Adams	05/22/24 14:24

PCB 6 2405078-06 (Paint Chips)

Analyte	Result	Reporting Limit	Units	Batch	Date Prepared	Date Analyzed	Method	Notes
PCBs by GC/ECD								
Aroclor 1016	ND	2.00	mg/kg	AHE0160	05/20/24	05/20/24	PCBs BY EPA 8082	C-03, C-04, R-02
Aroclor 1221	ND	2.00	"	"	"	"	"	C-03, C-04, R-02
Aroclor 1232	ND	2.00	"	"	"	"	"	C-03, C-04, R-02
Aroclor 1242	ND	2.00	"	"	"	"	"	C-03, C-04, R-02
Aroclor 1248	ND	2.00	"	"	"	"	"	C-03, C-04, R-02
Aroclor 1254	ND	2.00	"	"	"	"	"	C-03, C-04, R-02
Aroclor 1260	ND	2.00	"	"	"	"	"	C-03, C-04, R-02
Surrogate: Decachlorobiphenyl	77.6 %	% Recovery Limits		60-140			"	C-03, C-04
Surrogate: Tetrachloro-meta-xylene	74.7 %	% Recovery Limits		60-140			"	C-03, C-04

SGS Excelchem Laboratories, Inc.

eig _ el

The results in this report apply to the samples analyzed in accordance with the chain of custody document. This analytical report must be reproduced in its entirety.

SGS Carson	Project:	L1969	
20535 Belshaw Ave	Project Number:	L1969	Date Reported:
Carson, CA 90746	Project Manager:	Nicole Adams	05/22/24 14:24

PCBs by GC/ECD - Quality Control

Analyte	Result	Reporting Limit	Units	Spike Level	Source Result	%REC	%REC Limits	RPD	RPD Limit	Notes
Batch AHE0160 - PCBs BY EPA 8082										
Blank (AHE0160-BLK1)				Prepared &	Analyzed:	05/20/24				
Surrogate: Decachlorobiphenyl	0.0235		mg/kg	0.0261		90.0	60-140			C-03, C-04,
	0.0225			0.0261		06.0	60.140			Z-01
Surrogate: Tetrachloro-meta-xylene	0.0225			0.0261		86.3	60-140			<u>C-03, C-04,</u> <u>7-01</u>
Aroclor 1016	ND	0.0912	"							C-03, C-04,
										Z-01
Aroclor 1221	ND	0.0912	"							C-03, C-04,
Angelon 1929	ND	0.0012								Z-01
Aroclor 1252	ND	0.0912								C-03, C-04, Z-01
Aroclor 1242	ND	0.0912								C-03, C-04,
										Z-01
Aroclor 1248	ND	0.0912	"							C-03, C-04,
										Z-01
Aroclor 1254	ND	0.0912	"							C-03, C-04,
Aroclor 1260	ND	0.0912								Z-01 C-03 C-04
AI00101 1200	ND	0.0912								Z-01
LCS (AHE0160-BS1)				Prepared &	Analyzed:	05/20/24				
Surrogate: Decachlorobiphenvl	0.0220		mg/kg	0.0262		83.9	60-140			C-03. C-04
Surrogate: Tetrachloro-meta-xylene	0.0203		"	0.0262		77.6	60-140			C-03. C-04
Aroclor 1016	0.977	0.0916	"	1.31		74.6	50-150			C-03, C-04
Aroclor 1260	1.23	0.0916	"	1.31		94.0	50-150			C-03, C-04
LCS Dup (AHE0160-BSD1)				Prepared &	Analyzed:	05/20/24				
Surrogate: Decachlorobiphenvl	0.0229		mg/kg	0.0261		87.8	60-140			C-03, C-04
Surrogate: Tetrachloro-meta-xylene	0.0203		"	0.0261		77.7	60-140			C-03, C-04
Aroclor 1016	1.01	0.0914	"	1.31		77.4	50-150	3.44	50	C-03, C-04
Aroclor 1260	1.27	0.0914	"	1.31		97.4	50-150	3.30	50	C-03, C-04
Matrix Spike (AHE0160-MS1)		Source: 240508.	3-01	Prepared &	Analyzed:	05/20/24				
Surrogate: Decachlorobiphenyl	0.0187		mg/kg	0.0266		70.0	60-140			C-03, C-04
Surrogate: Tetrachloro-meta-xylene	0.0178		"	0.0266		67.0	60-140			C-04, C-03
Aroclor 1016	1.34	0.0933	"	1.33	ND	100	50-150			C-03, C-04
Aroclor 1260	1.02	0.0933	"	1.33	ND	76.7	50-150			C-03, C-04

SGS Excelchem Laboratories, Inc.

/ ez sel 2 L

The results in this report apply to the samples analyzed in accordance with the chain of custody document. This analytical report must be reproduced in its entirety.

SGS Carson	Project:	L1969	
20535 Belshaw Ave	Project Number:	L1969	Date Reported:
Carson, CA 90746	Project Manager:	Nicole Adams	05/22/24 14:24

PCBs by GC/ECD - Quality Control

Analyte	Result	Reporting Limit	Units	Spike Level	Source Result	%REC	%REC Limits	RPD	RPD Limit	Notes
Batch AHE0160 - PCBs BY EPA 8082										
Matrix Spike Dup (AHE0160-MSD1)		Source: 240508	3-01	Prepared &	Analyzed:	05/20/24				
Surrogate: Decachlorobiphenyl	0.0197		mg/kg	0.0266		74.2	60-140			C-04, C-03
Surrogate: Tetrachloro-meta-xylene	0.0190		"	0.0266		71.3	60-140			C-03, C-04
Aroclor 1016	1.44	0.0931	"	1.33	ND	109	50-150	7.67	50	C-03, C-04
Anaplan 1260	1.07	0.0021		1 22	ND	00.0	50 150	5.05	50	G 02 G 04

SGS Excelchem Laboratories, Inc.

sug -Sel /

The results in this report apply to the samples analyzed in accordance with the chain of custody document. This analytical report must be reproduced in its entirety.

SGS Cars	on Project:		L1969		
20535 Be	lshaw Ave	Project Number:	L1969	Date Reported:	
Carson, C	A 90746	Project Manager:	Nicole Adams	05/22/24 14:24	
		Notes and I	Definitions		
Z-01	Silica gel/florisl clean-up performed twice	ce on sample			
R-02	Elevated Reporting Limits due to limited	l sample volume.			
C-04	To reduce matrix interference, the sampl compound contamination.	e extract has undergo	ne florisil clean-up, which is specific to non-polar		
C-03	To reduce matrix interference, the sampl compound contamination.	e extract has undergo	ne silica-gel clean-up, which is specific to polar		
ND	Analyte not detected at reporting limit.				
NR	Not reported				

SGS Excelchem Laboratories, Inc.

SGS Excelchem Laboratories, Inc.

/ ez sel 2 L

The results in this report apply to the samples analyzed in accordance with the chain of custody document. This analytical report must be reproduced in its entirety.

SG

SGS Excelchem Laboratories, Inc.

SGS Carson	Project:	L1969	Date Reported: 05/22/24 14:24
20535 Belshaw Ave	Project Number:	L1969	
Carson, CA 90746	Project Manager:	Nicole Adams	

Subcontract Sample Release Form & Receipt

	Attn: Jo 1135 W Suite A Rocklin	e Trapasso . Sunset Blvd. , CA 95765	Finn Fax: Emai	916-543-4449 916-543-4449 I: joseph.trapasso@	@sgs.com
Date:	05/15/2024		From Pare	: Terracon - Conc nt RN:	cord
	SGS Forensic custody:	Laboratories on th	his day releases custody and control of	of the following same	ple(s) presently in our
	FALI Job	P.O.	Analysis Requested		Due Date
	L1969	U003063	PCB - NO SPECIAL REPORTI TAT (Due 5/23) - Email Lab Do ENV.LosAngeles.PM@sgs.com	NG LIMITS - 5 Day	6 – 5 DAY TAT
PLEAS	SE USE Report	U003063 as the I	Purchase Order (PO).		
Sneeial	Handling. Se	nd results and	invoice only to ENV. Hayway	rd.SUBCONTR	ACT@sgs.com
	and the second				
		_			
		SU	BMISSION LOCATION (Ple	ase circle)	
SGS		1	SGS		SGS
3777 De	anot Road Suite	400	20535 S Belshaw Avenu	ie	3626 Sunset Rd., Suite 10
Havavar	d CA 94545	102	Carson CA 90746		Las Vegas NV 89120
Tayward	u, CA 94545		Carson, Crr 50740		Eus regus, mr synas
		60			
		DO			2405078
	+	DO			2405078
Release	ed By:	PO		(2007)	2405078
Release	ed By:		S)		2405078
Release	ed By:		S)		2405078
Release	ed By:	(SG:	S)		2405078 BIN B-3
Release	ed By:	Fodda (sos	s)		2405078 BIN B-3
Release	ed By:	Fade (s) CUD/45	الله (کار ال	2405078 BIN 13-3 2:020 pm
Release	ed By:	Jade (2405078 BIN 13-3 2:02 pm
Release	ed By:	Jade ((Print N	s) DODAS	<u>اللہ (مارا</u> ع	2405078 BIN 3-3 2:02 pm
Release	ed By:	Jade ((Print N	s) DODAS	<u>الم</u> (مع (مع	2405078 BIN B-3 2:01e pm te/Time)
Release	ed By:	Jade ((Print N	s) DODAS	होरि) <u>अत्</u> (Da	2405078 BIN 13-3 2:02 pm te/Time)
Release	ed By:	Jade ((Print N	s) DODAS	5/15/24 (Da	2405078 RIN 13-3 2:02 pm
Release	ed By:	Jade ((Print N	s) DODACS	<u>ह] ७) २५</u> (Da	2405078 RIN 13-3 2:02 pm
Release	ed By:	Jade ((Print N	s) DODAS Jame)	<u>الم (ما</u> ع) (Da	2405078 BIN B-3 2:02 pm te/Time)
Release	ed By:	Jade (Print N	s) DODIES Jame)	<u>ह]।८] २५</u> _{(Da}	2405078 RIN B-3 2:02 pm te/Time)
Release	ed By:	Jade (Print N (Client / Vondor I	s) Lame) SQS Representative)	ह] <u>८) २५</u> (Da	2405078 RIN 13-3 2:02 pm te/Time)
Release	ed By:	Jade (Print N (Client / Vendor I	s) SOULS Jame) SOS Representative)	5/15/24 (Da	2405078 RIN 13-3 2:02 pm te/Time)
Release	ed By:	(SGS)	S) SOOVES Jame) SOS Representative)	<u>اللہ (ما</u> ع (Da	2405078 BIN B-3 2:02 pm te/Time)
Release	ed By:	(SGS Jade ((Print N (Client / Vendor I	S) DODAS Jame) SOS Representative)	5/15/24 (Da	2405078 RIN B-3 2:02 pm te/Time) 9:52
Release	ed By:	(Client / Vondor I	s) DODAS Jame) SQS Representative)	5/14/24	2405078 RIN 13-3 2:02 pm te/Time) 9:52
Release	ed By:	(Client / Vondor I (Print N (Client / Vondor I (Client / Vondor I (Print N	s) SOULS Jame) SOS Representative) Asch_ Jame)	5/14/24 (Da	2405078 RIN 13-3 2:020 pm te/Time) 9:52 te/Time)
Release	ed By:	(Client / Vondor I AC AUAC (Print N	S) SOOVES Jame) SOS Representative) ASCA Jame)	5/14/24 (Da	2405078 RIN B-3 2:02 pm te/Time) 9:52 te/Time)
Release Receive	ed By:	(SGS Jade ((Print N (Client / Vendor I (Client / Vendor I (Client / Vendor I (Print N	S) DODAS Jame) SOS Representative) ASEA Jame)	5/14/24 (Da	2405078 RIN B-3 2:01e pm te/Time) 9:52 te/Time) Page
Release Receive	ed By:	(Client / Vendor I Client / Vendor I Client / Vendor I Che Albeet (Print N	s) SOOLS Jame) SOS Representative) Asch Jame)	5/14/24 (Da	2405078 RIN B-3 2:020 pm te/Time) 9:52 te/Time) Page
Release Receive	ed By:	(Client / Vondor I (Print N (Client / Vondor I (Print N	s) SOULS Jame) SOS Representative) Asch Jame)	5/14/24 (Da	2405078 RIN 3-3 2:020 pm te/Time) 9:52 te/Time) Page

SGS Excelchem Laboratories, Inc.

Sel e 2 L

The results in this report apply to the samples analyzed in accordance with the chain of custody document. This analytical report must be reproduced in its entirety.

SGS Carson	Project:	L1969	
20535 Belshaw Ave	Project Number:	L1969	Date Reported:
Carson, CA 90746	Project Manager:	Nicole Adams	05/22/24 14:24

20 Concord Av	enue, Sui	te 450, Con	cord, Califo	ornia 94520	PCB A	NALYSI	s	P	nge <u>1</u> of <u>1</u>
PM - S. Steiner Steff@rgaenv.com Fax: 510 899.705		PM - K. Schr Karin@rgaen Fax: 510.899	noeter nv.com 0,7063	PM - K. Pilgrim <u>Ken@rgaenv.com</u> Fax: 510.899.7053					
PM - T. Kattchee Tedd@rgaenv.com Fax: 510.899.707	n D	PM - B. Gils Bob@rgaenv Fax: 510.899	7.com 0.7050	PM - M. Bryant marlin.bryant@rgacny.com Fax: 510,899.7062					
Project Name/Ad	Idress	San Bruno Me	ountain Park -	- 555 Guadalupe Parkw	ay, Brisbane, C	CA			
RGA Project Nu	nber	R1247013	Sampled I	By Karin Schroeter, V	WF, BW	Sampling	Date	May 14, 20	24
Laboratory S	GS Forensi	c X O	ther	Turn Around Time		5 Days	х	Other (Specify)
**FAX OR E-M	AIL REI	PORT TO: S	SEE ABOVE	PROJECT MANAG	ER (PM) **	*			
PCB#	Materia	I Descriptio	on: Venti	lation Louver Caul	k Above D	oor to S	torage	Ð	
Sample ID	Sample	Location &	& Material L	ocation	Quantity:		_		
PCB 1	Restro	om Buildin	g - South	End of the Lower	Edge of L	ouver			
PCB 2	Restro	om Buildin	g – Center	of the Lower Edg	ge of Louv	er			
PCB 3	Restro	om Buildin	g – North	End of the Lower	Edge of L	ouver			
Material Location:					0				
PCB#	Materia	I Descriptio	on: Restr	oom Building Doo	or Caulking				
Sample ID	Sample	Location a	& Material L	Destroom Door	Quantity:				
PCB 4	Restro	om Buildin	g - Men's	Restroom Door	25				
PCB 5	Restro	om Buildin	g - Wonie	Pestroom Door	,				
Material Location:	Restro	on Buildin	g nens	Restroom boor					
PCB#	Materia	I Descriptio	on:						
Sample ID	Sample	e Locations			Quantity:				
Α		100							
В	-						_		
С									
Material Location:									
PCB#	Materia	al Descriptio	on:		1 Ourseffe				
Sample ID	Sample	e Location d	& Material I	Location	Quantity	-			
A									
В									
Material Location:									
000#							_		
PCB#	Materia	al Descriptio	on:	lacation	Quantita				
Commenter ID	Sample	e Locauon	o, Material I	Location	quantity				
Sample ID									
Sample ID A									
Sample ID A C	-								

	TAPRIVARS.	OTOTAT OTTAT		
Relinquished By:	Karin Schroeter	Barn Schrocte	Terracon Consultants	May 14, 2024
Received By:				
Relinquished By:	RECEN	FICIDID 2		
Received By:	New York	apaos		
	MAY 1 5 2	024 00,19	: Tade Cooks	5/15/24 1:570m
	BYENC, 112	37	1 0	

SGS Excelchem Laboratories, Inc.

0 2 L

The results in this report apply to the samples analyzed in accordance with the chain of custody document. This analytical report must be reproduced in its entirety.

Section 1 – Sample Arrival Information	Project Number: Project Manager:	L1969 Nicole Adams WORK	D: 05/ Sample In FN: F-ROC-SC- Rev. Date: 11/0 Page ORDER:	ate Reported: /22/24 14:24
Sample Integrity Date Received: 516(24) Section 1 - Sample Arrival Information Sample Transport: ONTRAC, LIPS, 1	Project Manager:	Nicole Adams	05 SGS - R Sample In FN: F-ROC-SC- Rev. Date: 11/0 Page ORDER:	/22/24 14:24 cocklin tegrity 001-00 1/2023
Sample Integrity Date Received: 516(24 Section 1 - Sample Arrival Information Sample Transport: ONTRAC, LIPS, 1		WORK	SGS - R Sample In FN: F-ROC-SC- Rev. Date: 11/0 Page ORDER:	ocklin tegrity 001-00 1/2023
Date Received: 516(24 Section 1 - Sample Arrival Information				B
Section 1 – Sample Arrival Information		Company N New Client	lame: <u>SGS-</u> Hayword : Y	d
Sample Transport: ONTRAC LIPS I				
Transported In: Ice Chest Box Ha Packing materials: Bubble Wrap Fo Has chilling process begun? Y N Temperature of Samples (°C): 21- Thermometer ID: DT-16 CF (°C)	JSPS Walk-In E. am Packing Pean Samples R Ice Chest	XCELCHEM Courier Fed-E Paper Other: ceceived: Chilled to Touch Temperature(s) (°C):	Ambient / On Ice	
Section 2 – Bottle/Analysis Info.		Yes No N/A	Comments	
Did all bottles arrive unbroken and in	ntact?			
Did all bottle labels agree with COC	ta ata ang ang ata 10			
Were correct containers used for the	tests requested?			
Was a sufficient amount of sample so	ent for tests indicat	ed?		
Were bubbles present in VOA Vials?: (Vol	atile Methods Only)			
Is there head space in the VOA vials? (Vol	atile Methods Only)			
Section 3– COC Information	No Comments		Ves	No
COC Received	tio comments	Analysis Requested		
Date Sampled		Samples arrived within hold	ing time	
Time Sampled		Hold times less than 72 hour	rs T	/
Sample ID		Client Name		
Rush Turn Around Time	5	Client Contact Information		
pH Chlorine Corrosivity Coliform	SHORT HOLI Dissolved Oxygen	Odor Nitrate Nitri	te Ortho-phosphate	
MB Asbestos Settable Solids Turbidity	Biochemical Oxygen De	mand HPC Color Tedl	ars Ammonia/TKN (unpreserve	:d)
			1	
Section 4 – Comments / Discrepanc	ies			
Client notified of discrepancies: Yes / No	Notified by:	11-100 6-52		
	C			
Din Number/ Leastion	Filled out have	1	Deter	
Bin Number/ Location: 13-3	Fined out by:		Date:5) [[4]	524
COC Scanned/Attached by:	1 1		Time: 9:0	4
Samples labeled by:	[6			
Sample labels reviewed by:	1			
4 47	1			

SGS Excelchem Laboratories, Inc.

sug -Sel /

The results in this report apply to the samples analyzed in accordance with the chain of custody document. This analytical report must be reproduced in its entirety.
SGS Excelchem Laboratories, Inc.

SGS Carson	Project:	L1969	
20535 Belshaw Ave	Project Number:	L1969	Date Reported:
Carson, CA 90746	Project Manager:	Nicole Adams	05/22/24 14:24

1

Albertsen, Kathryn (Rocklin)

From:	Adams, Nicole (Carson)
Sent:	Thursday, May 16, 2024 11:26 AM
То:	Albertsen, Kathryn (Rocklin)
Cc:	Trapasso, Joseph (Rocklin)
Subject:	Re: Missing Sampling time for Project L1969

Hello Kate,

Here is the info, per Karin Schroeter w/ Terracon. LMK if you need anything else.

Terracon - PCB | RI247013 Sample Date & Time: May 14th 8-11am

Good day! Nicole

*Upcoming PTO 5/21/24 SGS North America Inc. Nicole Adams Sr. Project Manager SGS Built Environment 20535 S. Belshaw Avenue Carson, CA 90746 Nicole.Adams@sgs.com

From: Albertsen, Kathryn (Rocklin) <Kathryn.Albertsen@sgs.com>
Sent: Thursday, May 16, 2024 10:32 AM
To: Adams, Nicole (Carson) <Nicole.Adams@sgs.com>
Cc: Trapasso, Joseph (Rocklin) <Joseph.Trapasso@sgs.com>
Subject: RE: Missing Sampling time for Project L1969

Thank you. I appreciate it.

Kate Albertsen Industries and Environment Client Services Representative

SGS Excelchem Laboratories, Inc. 1135 W. Sunset Blvd, Suite A Rocklin CA 95765 USA

Phone: +01 916 543 4445 ext 102 E-mail: Kathryn.Albertsen@sgs.com

SGS Excelchem Laboratories, Inc.

el

The results in this report apply to the samples analyzed in accordance with the chain of custody document. This analytical report must be reproduced in its entirety.

PM S. Steiner PM - K. Schroeter PM - K. Pilgrim PCB ANALYSIS Page 1 or PM - S. Steiner V PM - K. Schroeter PM - K. Pilgrim Ken@rgaenv.com Pax: 510.899.7053 Page 1 or PM - T. Kattchee PM - B. Gils PM - B. Gils PM - B. Gils PM - M. Bryant PM - M. Bryant Fax: 510.899.7070 PM - B. Gils PM - B. Gils PM - Fax: 510.899.7062 PM - M. Bryant Fax: 510.899.7070 Fax: 510.899.7050 Fax: 510.899.7062 PM - M. Bryant PM - M. Bryant Project Name/Address San Bruno Mountain Park - 555 Guadalupe Parkway, Brisbane, CA Karin Schroeter, WF, BW Sampling Date May 14, 2024	/T f <u>1</u>
PM - S. Steiner PM - K. Schroeter PM - K. Pilgrim Steff@rgaenv.com Karin@rgaenv.com Ken@rgaenv.com Fax: 510 899.7051 PM - B. Gils Fax: 510.899.7053 PM - T. Kattchee PM - B. Gils PM - M. Bryant Tedd@rgaenv.com Bob@rgaenv.com Fax: 510.899.7053 Fax: 510.899.7070 PM - B. Gils PM - M. Bryant Bob@rgaenv.com Fax: 510.899.7050 Fax: 510.899.7062 Fax: 510.899.7070 Bob@rgaenv.com Fax: 510.899.7062 Fax: 510.899.7050 Fax: 510.899.7062 Fax: 510.899.7062	
Steff@rgaenv.com Fax: 510 899.7051 Karin@rgaenv.com Fax: 510.899.7063 Ken(@rgaenv.com Fax: 510.899.7053 PM - T. Kattchee Tedd@rgaenv.com Fax: 510.899.7070 PM - B. Gils Bob@rgaenv.com Fax: 510.899.7050 PM - M. Bryant marlinbryant@rgaenv.com Fax: 510.899.7062 Project Name/Address San Bruno Mountain Park - 555 Guadalupe Parkway, Brisbane, CA RGA Project Number R1247013 Sampled By Karin Schroeter, WF, BW Sampling Date May 14, 2024	
Pax: 510 899.7051 Pax: 510.899.7051 PM - R. Gils PM - M. Bryant PM - T. Kattchee PM - B. Gils Bob@rgaenv.com PM - M. Bryant Tedd@rgaenv.com Bob@rgaenv.com Fax: 510.899.7062 PM - M. Bryant Fax: 510.899.7070 Fax: 510.899.7050 Fax: 510.899.7062 PM - M. Bryant Project Name/Address San Bruno Mountain Park - 555 Guadalupe Parkway, Brisbane, CA RGA Project Number R1247013 Sampled By Karin Schroeter, WF, BW Sampling Date May 14, 2024	<u> </u>
Image: Project Name/Address Bob@rgaenv.com Fax: 510.899.7070 marlinbryant@rgaenv.com Fax: 510.899.7062 Project Name/Address San Bruno Mountain Park - 555 Guadalupe Parkway, Brisbane, CA RGA Project Number R1247013 Sampled By Karin Schroeter, WF, BW Sampling Date May 14, 2024	
Fax: 510.899.7070 Fax: 510.899.7030 Fax: 510.899.7030 Fax: 510.899.7030 Project Name/Address San Bruno Mountain Park - 555 Guadalupe Parkway, Brisbane, CA RGA Project Number R1247013 Sampled By Karin Schroeter, WF, BW Sampling Date May 14, 2024	
Project Name/AddressSan Bruno Mountain Park - 555 Guadalupe Parkway, Brisbane, CARGA Project NumberR1247013Sampled ByKarin Schroeter, WF, BWSampling DateMay 14, 2024	
RGA Project Number R1247013 Sampled By Karin Schroeter, WF, BW Sampling Date May 14, 2024	_
Laboratory SGS Forensic X Other Turn Around Time 5 Days X Other (Specify)	
*** <u>FAX OR E-MAIL REPORT TO</u> : SEE ABOVE PROJECT MANAGER (PM) ***	
PCB# Material Description: Ventilation Louver Caulk Above Door to Storage	
Sample ID Sample Location & Material Location Quantity:	
PCB 1 Restroom Building – South End of the Lower Edge of Louver	
PCB 2 Restroom Building – Center of the Lower Edge of Louver	
PCB 3 Restroom Building – North End of the Lower Edge of Louver	
Material Location:	
PCB# Material Description: Restroom Building Door Caulking	
Sample ID Sample Location & Material Location Quantity:	<u> </u>
PCB 4 Restroom Building – Men's Restroom Door	
PCB 5 Restroom Building – Women's Restroom Door	
PCB 6 Restroom Building – Men's Restroom Door	
Material Location:	
PCB# Material Description:	
Sample ID Sample Locations Quantity:	
В	
C	
Material Location:	
PCB# Material Description:	
PCB# Material Description: Sample ID Sample Location & Material Location Quantity:	
PCB# Material Description: Sample ID Sample Location & Material Location Quantity: A	
PCB# Material Description: Sample ID Sample Location & Material Location Quantity: A	
PCB# Material Description: Sample ID Sample Location & Material Location Quantity: A	
PCB# Material Description: Sample ID Sample Location & Material Location Quantity: A	
PCB# Material Description: Sample ID Sample Location & Material Location Quantity: A	
PCB# Material Description: Sample ID Sample Location & Material Location Quantity: A	
PCB# Material Description: Sample ID Sample Location & Material Location Quantity: A	
PCB# Material Description: Sample ID Sample Location & Material Location Quantity: A	

	NAME:	SIGNATURE:	COMPANY:	DATE:
Relinquished By:	Karin Schroeter	Kom Sensete.	Terracon Consultants	May 14, 2024
Received By:				
Relinquished By:	RECEN	TRO		
Received By:	MAYA	1002		
	MAY 1 J 2	1024 Rela	: Jade Works	5/15/24 1:571
	BY L L	20		

Pre-demolition Hazardous Materials Survey San Bruno Mountain Park - Day Use Area | Brisbane, CA May 24, 2024 | Terracon Report No. R1247013



APPENDIX E

SAMPLE LOCATION FIGURES







Forracon	Site Soil Sampling Event	Sample Lo San Bruno Mountai Brisbar	cation Drawing n State and County Park ne, California	Not to Scale
	SURVEY DATE: May 14, 2024	PROJECT NO.:	R1247013	FIGURE: 2

Pre-demolition Hazardous Materials Survey San Bruno Mountain Park - Day Use Area | Brisbane, CA May 24, 2024 | Terracon Report No. R1247013



APPENDIX F

CERTIFICATIONS

STATE OF CALIFORNIA

DEPARTMENT OF INDUSTRIAL RELATIONS **Division of Occupational Safety and Health-Asbestos Certification** 1750 Howe Avenue, Suite 460 Sacramento, CA 95825 (916) 574-2993 Office <u>http://www.dir.ca.gov/dosh/asbestos.html</u> <u>actu@dir.ca.gov</u>





201234853C

359

Bastion EHS William M Frieszell 2950 Merced Street, Suite 220 San Leandro CA 94577 February 27, 2024

Dear Certified Asbestos Consultant or Technician:

Enclosed is your certification card. To maintain your certification, you must abide by the rules printed on the back of the certification card.

Your certification is valid for a period of one year. If you wish to renew your certification, you must apply for renewal at least 60 days <u>before</u> the expiration date shown on your card. [8 CCR 341.15(h)(1)].

Please hold and do not send copies of your required AHERA refresher renewal certificates to our office until you apply for renewal of your certification.

Certificates must be kept current if you are actively working as a CAC or CSST. The grace period is only for those who are not actively working as an asbestos consultant or site surveillance technician.

Please contact our office at the above address or email w any changes in your contact/mailing information within 15 days of the change.

Sincerely,

Kethulis

Kevin Graulich Principal Safety Engineer

Attachment: Certification Card

cc: File



Occupational Safety and Health as authorized by Sections 7180 at seq. of the Business and Professions Code.





State of California





STATE OF CALIFORNIA DEPARTMENT OF PUBLIC HEALTH



LEAD-RELATED CONSTRUCTION CERTIFICATE

INDIVIDUAL:

CERTIFICATE TYPE:

NUMBER:

EXPIRATION DATE:



Lead Inspector/Assessor

LRC-00000210

5/4/2025

Brad Wallenberg

Disclaimer: This document alone should not be relied upon to confirm certification status. Compare the individual's photo and name to another valid form of government issued photo identification. Verify the individual's certification status by searching for Lead-Related Construction Professionals at www.cdph.ca.gov/programs/clppb or calling (800) 597-LEAD

Docusign Envelope ID: A3228B90-A416-4991-8459-8F8515CCF10B

State of California Division of Occupational Safety and Health Certified Asbestos Consultant



Karin M. Schroeter

Certification No. ____92-0772

Expires on _____01/15/25

This certification was issued by the Division of Occupational Safety and Health as authorized by Sections 7180 et seq. of the Business and Professions Code.



STATE OF CALIFORNIA DEPARTMENT OF PUBLIC HEALTH



LEAD-RELATED CONSTRUCTION CERTIFICATE

INDIVIDUAL:	CERTIFICATE TYPE:	NUMBER:	EXPIRATION DATE:
	Lead Inspector/Assessor	LRC-00005265	4/6/2025

Karin Schroeter

Disclaimer: This document alone should not be relied upon to confirm certification status. Compare the individual's photo and name to another valid form of government issued photo identification. Verify the individual's certification status by searching for Lead-Related Construction Professionals at www.cdph.ca.gov/programs/clppb or calling (800) 597-LEAD

Exhibit 3 SCOPE OF WORK

Docusign Envelope ID: A3228B90-A416-4991-8459-8F8515CCF10B CONCEPTUAL DESIGN



Date: May 28, 2024

Subject: Romtec Scope of Work & Conceptual Design

Name of Project: San Bruno Mountain Park Restroom Replacement Project

1. Romtec Scope of Work

Romtec will work through the complete design of the structure as outlined in the County RFP and the Romtec Technical Proposal. This includes all County Parks review, County Planning Department Review, and County Building department review. Romtec has provided a detailed design and review process, based on the County RFP, in our Technical Proposal. Once the designs are complete and approved Romtec will produce the building package and deliver it to the site for installation. The complete production and installation process is outlined in the Romtec Technical Proposal and Project Schedule.

2. Romtec Scope of Materials Supply

- A. Structure
 - 1. Romtec's proposal includes the specific *color* selections noted below. Deviation from these may result in a price increase.
 - 2. The following building package structural items will be supplied by Romtec.
 - a. Concrete Masonry Units CMU
 - i. Exterior walls will be constructed of split-face, mortar joint, concrete masonry units (concrete blocks). Base Bid includes siding package.
 - ii. Block color will be Split-face tan.
 - Add Alternate 1:
 - i. Split-face *dark tan* CMU block for lower courses and top courses to be split-face *light tan* CMU block.
 - Add Alternate 2:
 - ii. Smooth-face grey CMU with exterior finish to be fiber cement, Hardi, Cedarmill board and batten siding with stone veneer accent.
 - iii. [stone veneer siding not selected]
 - b. Interior Paint: Sherwin Williams *Note: County to select color.*
 - c. Natural Hues, eco-body ceramic monochromatic look, Q3685 6x8 cove base. Color: Carnation Glass QH48* (2)
 - d. SpecChem, Polycure concrete protective sealer.
 - e. Doors, frames, and hardware
 - i. Steelcraft, F16 standard double rabbet frame with optional 14-gauge closer reinforcement.
 - ii. Steelcraft, SL18 standard laminated Honeycomb core doors.
 - iii. Doors and frames are powder coated **black**.
 - iv. Hager, WTBB1168, five knuckle, ball bearing, heavy weight wide throw, 4-1/2 x 6 door hinges.
 - v. 18'x18 kick proof door louver
 - vi. All rooms shall have LCN-4110 extra duty with hold open arms, hurricane rated door closers.

- vii. Schlage L Series, Grade 1 mortise locks with full size interchangeable core (FSIC).
 - a. Public-Use Doors: LV 9056R 06A 626 x 09-509 x L583-363 x L283-722
 b. Chase Doors: LV 9056R 06A 626 x 09-509 x L583-363
- viii. All public-use doors shall also have Schlage B-Series Auxiliary Locks to be used for maintenance shut down purposes, they will not be operable by the public.
 - a. Grade 1 deadbolts, B662R 626 with full size interchangeable construction core (FSIC).
- f. Roofing materials
 - i. Wood truss roof system.
 - ii. Owens Corning Kraft Faced—EcoTouch® Batt Insulation R30 (no insulation over mechanical rooms)
 - iii. Solatube 160 DS
 - iv. AC Sanded Plywood, Douglas Fir.
 - v. Roofing will be Fabral, 26-gauge, Horizon 16 standing seam roof panels with 16" coverage. Color: **to be selected by owner.**
- g. Cabinet, heavy duty metal, 24"D x 48"W x 72"H storage cabinet.
- h. Romtec manufactured steel bulletin board, polycarbonate with cork, MI-000-1027, powder coated **black**.
- B. Plumbing Fixtures and Accessories
 - 1. The following items relative to the fixtures will be supplied by Romtec:
 - a. Wall mount, ProFlo, 1.28 gpf, high efficiency flush valve, vitreous china bowl.
 - i. ProFlo, PFTSCOF2000WH, Commercial toilet seat.
 - ii. Sloan Flushometers, 1.6/1.1 gpf, Polished Chrome Finish, Fixture Connection Top Spud, Dual Flush, UPPERCUT® Exposed Manual Water Closet Flushometer.
 - b. Wall mount, ProFlo, PF5414 lavatory.
 - i. Chicago Faucets, MVP faucets 333-665PSHABCP.
 - c. Bradley, 1-1/2" O.D. heavy-duty stainless-steel grab bar with concealed mounting.
 - i. 812 Grab bar, 001-18"
 - ii. 812 Grab bar, 001-36"
 - iii. 812 Grab bar, 001-42"
 - d. Bradley, 781-024300 channel frame mirror.
 - e. Bradley, Model 377 13-gallon Capacity trash can with lid.
 - f. Romtec manufactured, Wall mount, stainless steel 3-roll toilet paper dispensers, MI-011-5003.
 - g. Wall mount, stainless steel, Bradley 8496 soap dispensers.
 - h. Surface-mount, Bradley 9631 baby changing station. Color: Light Grey Bradex
 - i. Elkay Outdoor EZH20 bottle filling station wall mount bi-level non-filtered nonrefrigerated, 316 stainless, model LK4409BF. Color: **Black (BLK)**.
 - i. Brey Krause, D-7867-SS grab bars at drinking fountain location.
 - j. Mop Service Basin 63M.
 - i. Service Sink Faucet, 63.600A.
 - ii. Hose and Hose Holder, 65.700.
 - iii. Mop Hanger, 65.600
 - iv. Bumper Guards, 63.401 20-1/4" length
 - v. Duraguard Wall Guards, 67.2424
 - vi. Romtec manufactured mop faucet mounting bracket, PL-000-1042, powder coated **black**.

- k. Building signage:
 - i. Compliance Signs, ADA All Gender Restroom signage, RRE-31960-99 *white on black* CS794340.
 - ii. Compliance Signs, ADA Family Restroom signage, RRE-170 *white on black* CS984064.
 - iii. Compliance Signs, Baby Changing Area signage, RRE-175 white on black CS416754.
- I. Floor drain with mesh debris basket.
- C. Electrical Fixtures
 - 1. The following electrical fixtures will be supplied by Romtec:
 - a. Exterior lighting
 - i. Soffit mounted exterior lights.
 - ii. Single roof-mounted photocell for all exterior lights.
 - b. Interior Restroom, Shower Room & Public-Use Room Lighting
 - i. Lithonia Lighting, 48" surface/suspended mount WL4 LED light fixtures with integrated occupancy sensor and photocell.
 - c. Interior Mechanical Room Lighting
 - i. Lithonia Lighting, 48" ceiling mount, XVML LED Vapor Tight light fixtures, XVML L48 3500LM MVOLT 40K 80CRI.
 - d. TA-SB, Brushed Stainless Steel ThinAir® HAND DRYER.
 - e. Broadford, 80 gallon, 240V, 5.5 KW, ALTE8040R024000 water heater.
 - f. Romtec exhaust Air Assembly system to include 12" exhaust fans.
 - g. BRP30B200R 200AMP, single phase, rain-tight, breaker panel.
- 3. Conceptual Building and Site Designs

The following plan view, elevation, and site plan drawings outline Romtec's conceptual plan for each of the proposed buildings. Each building drawing and site plan drawing address the floor plans, retaining walls, parking stalls, and any other work necessary to complete each building. See below.

- i. Lithonia Lighting, 48" surface/suspended mount WL4 LED light fixtures with integrated occupancy sensor and photocell.
- c. Interior Mechanical Room Lighting
 - i. Lithonia Lighting, 48" ceiling mount, XVML LED Vapor Tight light fixtures, XVML L48 3500LM MVOLT 40K 80CRI.
- d. Rinnai Sensei, RU199i, internal (indoor) condensing tankless water heater. (only for buildings with showers)
 - i. Ferguson, Legend valve and fitting, FIP brass curb stop, #314-258NL. Note: Valve will be sized based on waterline size.
- e. Romtec exhaust Air Assembly system to include 14" exhaust fans.
- 3. Conceptual Building and Site Designs

The following plan view, elevation, and site plan drawings outline Romtec's conceptual plan for each of the proposed buildings. Each building drawing and site plan drawing address the floor plans, retaining walls, parking stalls, and any other work necessary to complete each building. See Exhibit & Schematic Design Documents. Docusign Envelope ID: A3228B90-A416-4991-8459-8F8515CCF10B ____NS & ASSUMPTIONS



Date: May 28, 2024

Subject: Exclusions & Assumptions

Name of Project: San Bruno Mountain Park Restroom Replacement Project.

Romtec is providing our proposal based on key assumptions and exclusions for this project. Below you will find a list of the key items Romtec has either assumed based on the County RFP or excluded from our scope. Many of these items were not defined in the RFP and should be addressed.

Exclusions:

- 1. Abatement of hazardous material is excluded from our scope of work. Romtec did not receive a Hazardous Material Report prior to bidding. If abatement is required Romtec can include abatement of hazardous material.
- 2. Romtec is excluding the new invert into existing Lift Station. This work shall be completed by others and extend a min of 10' outside of existing Lift Station.
- 3. Romtec excludes the work of any existing irrigation equipment. Romtec's proposal only includes the connection of the existing irrigation equipment.
- 4. A new meter base is excluded from the Romtec scope of work. Romtec's proposal does include the upgrade from the transformer to the new restroom.
- 5. Romtec excluded the rebuilding of the existing transformer shed.
- 6. Romtec excludes the design, supply, and installation of the generator pad.
- 7. Romtec excludes ADA accessibility walking paths and parking stalls to the building. In other words, Romtec is only providing 5' sidewalks around the building. Romtec can bid the walking paths and parking stalls if necessary, however, they are not included in the Romtec proposal.

Assumptions:

- 1. It is assumed that the project will be permitted and constructed in one phase.
- 2. Unless otherwise addressed, existing utilities have adequate capacity to serve the proposed improvements, that they are adjacent to the site frontage and do not require main extensions, and that utility system capacity studies are not required.
- 3. Construction staking scope and fee is based on the assumption that BKF will be provided, by the Owner, a site benchmark and adequate topographic survey control to accurately place site feature within each area. Boundary resolutions, and re-establishment of survey control prior to the start of construction surveying will be considered extra work.
- 4. Geotechnical report to be provide by the owner.
- 5. Romtec is not providing any lift station or pump station designs.
- 6. Romtec is not constructing or performing any work related to improving access roads.

Exhibit 4 PRICE PROPOSAL

Docusign Envelope ID: A3228B90-A416-4991-8459-8F8515CCF10B



18240 North Bank Rd. F: 541-496-0803 E: service@romtec.com Date

5/28/2024

PROPOSAL/PO

San Bruno Mountain Replacement

Customer: County of San Mateo Mario Nastari 455 County Center, 4th Floor Redwood City, CA 94063

Quantity	San Bruno Mountain Res	an Bruno Mountain Restroom replacement											
1	Romtec Restroom Building	\$	942,012.63										
	Installation Per Romtec Tec												
	Add Alternate 1	\$	4,080.00	Add Alternate 2	\$	8,160.00							
	Add Alternate 3	\$	74,573.28	Add Alternate 4	\$	59,266.00							
			Freight to:	Brisbane, CA		\$6,141							

	\$0)111
ROMTEC INC. PURCHASE ORDER TOTAL	\$ 948,153.63
ESTIMATED TAX BASED ON BASE BID PRICE: \$37,680.67	
NOTE: ESTIMATED TAX IS NOT INLCUDED IN ROMTEC INC. PURCHASE ORDER TOTAL	\$ -
AMOUNT	

	Add Alternate Descriptions
Base Bid	Block color will be Tan Split-face CMU Block. Steel post for roof extension area.
Add Alternate #1	
	Bottom 5 Courses of CMU block to be Dark Tan with remaining courses to be light tan.
Add Alternate #2	Exterior finish to be fiber cement, Hardi, Cedarmill board and batten siding with stone
	veneer accent.
Add Alternate #3	Romtec is proposing two (2) Roll up doors & CMU walls for the covered area of the
	building. Pricing inlcudes roll up doors, and CMU block for additional walls. The pricing
	would be in addition to the base bid and any add alternate pricing.
Add Alternate #4	Romtec is proposing two (2) Barn styledoors & CMU walls for the covered area of the
	building. Pricing inlcudes roll up doors, and CMU block for additional walls. The pricing
	would be in addition to the base bid and any add alternate pricing.

NOTE: ADD ALTERNATE PRICES ARE NOT INCLUDED IN THE PURCHASE ORDER TOTAL AMOUNT. THE PURCHASE ORDER TOTAL AMOUNT IS FOR THE ROMTEC "BASE BID" BUILDINGS. IF ADD ALTERNATES ARE SELECTED THERE WOULD BE A CHANGE TO THE PURCHASE ORDER TOTAL.

EXHIBIT 4B

San Mateo County Parks - San Bruno Mountain Park Restroom Replacement Project <u>ROMTEC CONTRACT AMOUNT BREAKDOWN</u>

Romtec Restroom Building Base Bid	\$	942,013	
(Design, Supply, Demolition & Installation)			
Add Alternate #2	\$	8,160	
(Exterior Finish Hardie -Board & Stone Vaneer)			
Sub-Total	\$	950,173	
Design and Supply Sub-Total (Including Add Alt #2)	¢	370 567	39%
Installation Sub-Total (Including Add Alt #2)	Ś	579.606	61%
Total Design and Supply Sub-Total (Including Add Alt #2)	\$	950,173	
San Mateo County Sales Tax (9.375%) (Design and Supply Portion Only)	\$	34,741	
Freight	\$	6,141	
Total Fixed Fee	\$	991,055	
10% Design Phase	\$	99,105	
90% Construction Phase	\$	891,949	
Owner Construction Contingency	\$	50,000	5%
Agreement Total Amount	\$	1,041,055	

Exhibit 5

PERSONNEL AND EQUIPMENT

Docusign Envelope ID: A3228B90-A416-4991-8459-8F8515CCF10B



Date: May 28, 2024

Subject: Romtec Technical Proposal

Name of Project: San Bruno Mountain Park Restroom Replacement Project.

Thank you for allowing Romtec the opportunity to bid on the San Bruno Mountain Park Restroom Replacement Project. Based on the RFP release May 3, 2024 and appendix's Romtec has put together a technical proposal outlining our project team, expertise, process, and strategic plan. The following information is how Romtec will proceed and what Romtec feels is the best value to the County for this project. Romtec will be teaming with BKF for site design, Precision Structural Engineering for final structural design, and Double E Engineering for the building electrical design.

- 1. Proposed Project Team and Expertise
 - A. Please see page 11 of the Technical Proposal for the Romtec organizational chart defining the Romtec design building team.
 - B. The following defines Romtec's key employees and sub consultants that will be working on the San Bruno Mountain Park Restroom Replacement Project. This team has worked together on the previous County of San Mateo projects and many projects before. The Romtec employees who manage and oversee the project from start to finish including our sales department, project management, construction management, engineering team, and Romtec's president, Ben Cooper. We have partnered with a subconsultant, Precision Structural Engineering, who stamps all Romtec plan sets. In addition, Romtec has partnered with BKF Civil Engineering to complete the site design for each building location. Below is a more defined outline of what each team member will bring to the project.
 - i. Dave Smith, Romtec Construction and Project Manager. Dave has a Bachelor's Degree in Construction Management and manages over 20 Romtec installations per year. He will be the Romtec lead and point of contact during the project during production and construction. Dave will be overseeing production and scheduling installation of the restroom buildings.
 - ii. Lindsay Bogan, Romtec Project Manager. Lindsay will be assisting Dave in the review, production, and construction phase of this project. Lindsay will be assisting in sending out submittals, plan sets, coordinating building department review, and helping with all aspects of the project. Lindsay also handles the majority of Romtec projects each year and is very familiar with the Design Build Process.
 - iii. Ken Perry, Romtec Contractor. Ken Perry, of Perry's Precision Concrete, installs almost all Romtec's restrooms on turnkey or design build projects. Ken has partnered with Romtec and has installed our buildings for over 20 years. Ken will be managing his crew on site and handling any on site construction aspects of this project. Ken and his crews will be preforming or coordinating all the demolition, site work, and installation of the building.
 - iv. Dalton Deeks, Romtec Engineering Manager. Dalton has a Bachelor's Degree in Civil Engineer along with many years of experience on building projects. Dalton

will manage and coordinate our internal engineering team and any engineering subconsultants for this project. Dalton will be responsible for coordinating the building plan sets, demolition plans, and site plans to the Romtec team and will oversee all engineering related aspects of the project.

- v. Ralph Hall & Nabil Taha, Precision Structural Engineering. Ralph and Nabil have been Romtec's structural engineer for over 10 years and have "sealed" the majority of Romtec's plan set. Romtec will provide the design of each building for Precision Engineering to review and "stamp" on all projects in California. Precision Engineering is licensed in California and have an ample amount of experience in the Bay Area.
- vi. Joe Young & Melinda Thomas, BKF Civil Engineering. Romtec will again be partnering up with BKF Engineering for as we did for past restroom replacement projects. BKF is located in the bay area and has worked with Romtec on several other projects. BKF and Romtec have a great understanding of the project after completing phase I and will be working together on the demolition and site plans for each location.
- vii. Greg Pride, Double 'E' Engineering. Romtec has worked directly with Greg Pride for over 20 years to provide electrical engineering services on projects in California. Greg knows the Romtec process and product very well and is a key part of our team.
- viii. Dayna Lewis, Romtec Chief Financial Officer. Dayna oversees the Romtec accounting department. Dayna will be responsible for obtaining all insurance and bonding requirements, billing, and other accounting related items. Dayna and the accosting team will also be responsible for invoicing and collection for all Romtec project.
- ix. Mark Sheldon, Vice President of Operations. Mark has been with the company since the beginning and runs Romtec's day to day operations. Mark's primary role is to oversee project management, engineering, production, and operations of Romtec. It is his goal to make sure Romtec is working smoothly and efficiently during the engineering and production of each project
- x. Ben Cooper, President. Ben will be overseeing all Romtec employees and Romtec subconsultants for this project. During this project Ben will be involved in every step of the project to ensure the project is running smooth and efficient to ensure all parties are happy with final product.
- C. The following will demonstrate how the proposed project team will communicate, collaborate, and work together with the County. The Romtec team has worked with CPM and the County on past projects and hopes the process with be familiar for all. Romtec will be the lead on all design and construction related items for the project. Romtec will work with the County to finalize the building design, building materials, and site design for each location. Romtec will be in direct contact with the County while managing our team and subconsultants throughout the entire design building process. Romtec would like to request a weekly meeting with the County to discuss the project and status of the project. We will ultimately work together to obtain the demolition permits, building permits, and scheduling of construction for the buildings.
- D. Romtec has been a leader in the public building industry for over forty years specializing in the design, supply, and construction of public restrooms and similar type park and recreation structures. The most recent example of our design and construction experience is on the Memorial Park Restroom Replacement Phase II Project. Romtec worked directly with the County of San Mateo and CPM to Design, Supply, and Construct 7 restroom buildings. This project was over \$3M in value an includes demolition, site design, building design, site construction and building construction. This is one of many examples where Romtec has worked with our team to design, supply, and construct our buildings.

Romtec, Inc. | 18240 North Bank Road | Roseburg, Oregon 97470 | 541-496-3541 | sales@romtec.com

- E. Romtec's proposed team will result in the best value for the County in many ways. By having a company like Romtec, who specializes in restroom type structures, we know the most efficient and comprehensive way to design and install restroom shower buildings. This results in the best, most robust, restroom building for the most competitive pricing. In addition to the Romtec building design we have brought in a very well-known civil engineering team to help provide the most efficient site design. By having a well-designed site, the Romtec contractor can prepare the site and install the buildings quickly and more cost effective. I believe we were able to prove Romtec is the best value to the County with the completion of the Memorial Park Restroom Replacement Phase I project.
- F. Romtec's Installation crew includes graduates of apprenticeship programs. The work is being performed by Romtec certified installers and installation of the building's makeup more than 60% of the complete project.
- 2. Method and Strategic Plan
 - A. Reduced Technical and Managerial Approach Romtec will be coordinating the site civil and building design together per the County's RFP document released May 3, 2024. Per the RFP, Romtec will work though the design documents with the County prior to any demolition or construction work on the project. Upon being provided the "Intent to Award" by the County of San Mateo, Romtec will proceed immediately on the Schematic Design phase for the project. The Following details Romtec's technical and managerial approach to the design and construction phases of the project.
 - B. Design Capabilities

It's Romtec's intent to follow the design processed outlined in the County's RFP. Romtec will work with the county through the schematic design, design development, and 100% construction documents. The following outlines Romtec's Design Capabilities and professional services for the project.

- i) Schematic Design Phase
 - (1) Building schematic Design. Romtec will produce a submittal for each building which will include the following key documents:
 - i. Preliminary plan view and elevation drawings for each location
 - ii. Data sheets and specifications for all key components included in each building.
 - iii. Color Selection charts

This submittal will be provided to the county for review, comment, and approval. Romtec will formally respond in writing to all comments received from the County on the Schematic Design submittal and will incorporate those defined comments and changes in the Design Development phase.

The intent of the Schematic Design review is to approval all design options, floor plans, site design, building components, and any other items needed for approval. Once approved the county is confirming no more design changes will be taking place.

- (2) "Schematic" Civil Plans. Accompanying the Romtec Building Schematic Design Submittal will be the "schematic" site civil design. The following scope will be included in the "schematic" civil design:
 - i. Drawings. A one (1) sheet per restroom location that will include the following details:
 - 1. Existing Conditions
 - 2. Grading and Drainage
 - 3. Utilities (water, sewer, storm drain)
 - 4. To scale equipment layout of all Romtec structures.
 - ii. Plan for NPDES C.3 Compliance. Based on a preliminary review, it is assumed that the building site will be viewed by the County as separate areas with regards to the MRP Section C3.i requirements. Based on our preliminary review each site does not create and/or replace more than 2,500sf of impervious surface and thus does not trigger the need to implement post construction stormwater treatment facilities. All water will be designed to drain to vegetated areas.
 - iii. Romtec will respond in writing to all County comments on the schematic site plan and Plan View and Elevation submittal
 - iv. Incorporating all comments. Romtec will incorporate all County comments on the Schematic Design submittal in the Design Development building plans
- ii) Design Development

Upon County approval of the Schematic Design Phase, Romtec/BKF will prepare Design Development Drawings for the County and County Planning Department to Review. This phase will include the Romtec architectural, structural, mechanical, electrical, plumbing, and the BKF civil plans.

(1) Design Development for County Park & Rec. Approval.

- Building Plans: Romtec will produce an engineered plan set for the building i. which will include the following key documents:
 - (i) Romtec "unsealed" plan set including all structural, mechanical, electrical, and plumbing plans for the building.
 - (ii) Data sheets and specifications for all key components included in each building.
- Civil Plans for County Park & Rec. Approval. Accompanying the Romtec ii. Building Design Development Submittal will be the site civil design. The following scope will be included in the civil design:
 - (i) Complete Civil Site Designs for each location, including the following details:
 - 1. Existing Conditions
 - 2. Grading and Drainage
 - 3. Utilities (water, sewer, storm drain)
 - 4. To scale equipment layout of all Romtec structures.

(2) Design Development for Planning Department Review.

Romtec, Inc. | 18240 North Bank Road | Roseburg, Oregon 97470 | 541-496-3541 | sales@romtec.com

<u>Note</u>: Upon County Park acceptance of the Design Development submittal (Unsealed plans), Romtec will then submit separately to the County Planning department for approval. This will occur prior to submitting to the building department for permits.

- i. Final "Sealed" Building Plans for review by the County of San Mateo Building Department. Included will be the following:
 - (i) Sealed building plans (35-50 sheets)
 - (ii) Sealed structural calculations for each building to satisfy all permitting requirements. (30-50)
- ii. Final "Construction" Site Plans.
 - (i) Drawings. The following Sheet layout is anticipated:
 - 1. Existing Conditions
 - 2. Site Improvement Plan (Paving, Grading and Drainage) 7 Sheets
 - 3. Erosion Control Plan 7 Sheets
 - 4. Construction Details 2 Sheets
- iii. Demolition Plans
 - (i) Existing Conditions and Demolition Drawing
 - 1. All General Notes related to demolition and tree protection

iii) 100% Construction Documents

Upon the County's Parks and Planning Department acceptance of the Design Development Documents, Romtec will provide the 100% Construction Documents. Romtec will meet with the County prior to having the Construction Documents submitted to the building department. Per the County RFP, it is Romtec's intention to submit for a separate demolition and building permit. This will allow Romtec to start the demolition of the building prior to starting construction.

(1) 100% Construction Documents

- i. Final "Sealed" Building Plans for review by the County of San Mateo Building Department, including the following:
 - (i) Sealed building plans (35-50 sheets)
 - (ii) Sealed structural calculations for each building *to satisfy all permitting requirements. (30-50)*
- ii. Final "Construction" Site Plan(s).
 - (i) Drawings. The following sheet layout is anticipated:
 - 1. Existing Conditions and Demolition Plan 7 Sheets
 - 2. Site Improvement Plan (Paving, Grading and Drainage) 7 Sheets
 - 3. Erosion Control Plan 7 Sheets
 - 4. Construction Details 2 Sheets
 - (ii) Calculations: BKF will update the calculations performed during the Schematic Development phase to reflect the construction level design.
 - (iii) NPDES C.3 Compliance: Based on comments received from the County, BKF will further develop the methods to meet the NPDES requirements for post-construction storm water discharge. BKF will work with the Romtec to implement the site water quality features.

Romtec, Inc. | 18240 North Bank Road | Roseburg, Oregon 97470 | 541-496-3541 | sales@romtec.com

- Project Approval: BKF and Romtec will provide 100% level construction (iv)documents for use in the obtaining all local-jurisdictional agency plan approvals and permitting.
- iii. **Demolition Plans**
 - (i) Existing Conditions and Demolition Drawing
 - 1. All General Notes related to demolition and tree protection
- C. Construction Means and Methods

Romtec intends to produce building packages and construct the buildings in accordance with the agreed to timelines for the project, see "Project Schedule" for further detail. The following outlines the Romtec Production and construction process along with staging and phasing delivery of the building packages.

- (1) Building Production. With the County's approval, and Notice to Proceed on production, Romtec will start production of the building packages:
 - Romtec will begin purchasing the building material starting with any long i. lead time items.
 - ii. Once all material is delivered to the Romtec facility our team will package the building kits into stages. Romtec packages our building kits on pallets, in stages, to help with delivery, storage, and speed during construction. By having the buildings organized in stages it allows us to install building more efficiently and safely store the remaining material until needed.
 - Along with the building package, Romtec provides a Bill of Material (BOM) iii. outline all material include in each stage of the building package. This ensures Romtec is supplying the proper material for each building.
- (2) Delivery, Staging, and Site Security. Once the building packages are produced and ready for delivery Romtec will schedule "phased" deliveries of the building packages. Below is how Romtec plans to deliver the buildings:
 - Romtec have the building packages delivered to the park using a semi-truck ί. and flatbed trailers. The use of a "hot shot" may be used during for certain deliveries.
 - ii. The buildings will be delivered to the park entrance and then offloaded by the Romtec Construction team.
 - iii. Building packages will be delivered to an agreed to staging area where they will be stored until the building material is needed. Currently Romtec plans to store the building packages in camp spots closest to the building locations.
 - iv. The building packages are completely wrapped to protect the material from the elements and add a layer of security.
 - Romtec's crews will also plan on camping on site to provide on site security V. during the entire project.
- (3) Demolition. Once the Demolition and Building Permits are approved Romtec can start on the demolition, site work, and construction of the restroom building. Below is an outline on how Romtec plans to schedule the demo and construction phases:
 - Once demolition permits have been approved, we will start removing the i. existing building. This will start with tree protection around all project sites and disconnecting all existing utilities.

- ii. If required, Romtec will work with our abatement sub-contractor to start the demolition and disposal of the existing buildings. (please note this is excluded due to lack of having a hazardous report provided prior to bid).
- iii. Most of this work will be completed via an excavator and dump truck where site access allows.
- (4) Construction. Once demolition of the exiting structure is complete, Romtec will start with the site preparation for the new building. In general, the construction work will start with site work and rough in. Romtec will then start coordination the foundations and vertical construction at each site, throughout this process Romtec will be calling for all necessary inspections for the buildings:
 - i. Site preparation will include grading, setting gravel, preparing for any retaining walls, drainage, and more.
 - ii. Romtec will completer the new building site to be ready for a new building to be installed.
 - iii. All site aspects for the new building location will be considered during the "site work" phase. This includes walking paths and more.
 - iv. Romtec will then begin the "rough in" phase of the construction. During this phase the footings will be dug, and the footings and foundations will be formed with rebar set.
 - v. Following the rough in work Romtec can start on the vertical construction of the building. This includes the CMU walls and trusses for each building.
 - vi. Once the Walls and roof structure are complete Romtec can work on the "finish out" work for the building. This will include all plumbing, electrical, fixture installation, and so on.

Note: A 6-day work week with available work hours of 7:00 AM – 7:00 PM. Reductions in this schedule will impact the schedule and proposal value.

Romtමප්රීමාණ්ණික් Organizational Chart



Exhibit 6 SCHEDULE

Docusign Envelope ID: A3228B90-A416-4991-8459-8F8515CCF10B

(8) PROPOSED SCHEDULE

ID Task Mode	Task Name	Duration	Start	Finish	Predecessors	Resource Names	Q	tr 3, 2024	C	Qtr 4, 2024	Neu	Qtr 1, 2025	L Colo Ma	Qtr 2, 2025	Mari	Qtr 3, 2025		Qtr 4, 2025
0	Romtec Design, Supply,	315 day	rs Thu	Wed			May Jun	Jui Aug	Sep	Uct		ec Jan	Feb Ma	r Apr	Мау	Jun Jui	Aug	Sep Oct
	& Install	-	5/30/24	8/13/25														
1 🚽	Design Phase Services	207 days	Thu 5/30/2	4Fri 3/14/25									1					
2 -	Receive Intent to Award	1 day	Thu 5/30/24	Thu 5/30/24	-	County of San Ma	County of San N	Mateo - Contracting										
3 -5	Schematic Design	32 days	Fri 5/31/24	Mon 7/15/24	2	DBE	DBE											
	Design Submittal Package	10 0033	111 3/ 31/ 24	1110 0/ 13/ 24	2													
5 🔩	County Parks to review	10 days	Fri 6/14/24	Thu 6/27/24	4	County of San	Co	ounty of San Mateo - Pro	oject Managment	t								
	Package and Provide approval with Comments. No Additional County Parks Schematic Design					Managment												
	comments.																	
6 嘱	Romtec to Respond to Comments	10 days	Fri 6/28/24	Thu 7/11/24	5	DBE		DBE										
7 📑	Romtec to Revise and County to provide Formal Approval of Schematic Design.	2 days	Fri 7/12/24	Mon 7/15/24	6	County of San Mateo - Project Managment		County of San Ma	ateo - Project Ma	nagment								
8 -5	Design Development	97 days	Tue 7/16/24	Wed 11/27/2	4			r			1							
9 -	Romtec to Produce Unsealed Construction Documents for County Parks Review	20 days	Tue 7/16/24	Mon 8/12/24	7	DBE		DBE	E									
10	County Parks to Review Unsealed Design Development Documents and Provide Approval with Comments. No Additional County Parks Design Development Comments.	5 days	Tue 8/13/24	Mon 8/19/24	9	San Mateo County Planning Department,Coun of San Mateo - Project Managment		*	San Mateo Count	y Planning Depai	tment,County of San	Mateo - Project Mana	gment					
11 🔩	Romtec to Respond to	10 days	Tue 8/20/24	Mon 9/2/24	10	DBE			DBE									
12	County Parks Comments Romtec to Revise and	15 days	Tue 9/3/24	Mon 9/23/24	11	DBE				DBE								
	Provide Sealed Plans Based on County Parks Comments																	
13 📭	Romtec to Submit Parks Approved Sealed Design Development Drawings to Planning Department for Review	1 day	Tue 9/24/24	Tue 9/24/24	12	DBE			ř	DBE								
14 🔩	County Planning Department Review and Provide Approval with Comments. No Additional Planning Department Comments.	20 days	Wed 9/25/24	Tue 10/22/24	13	San Mateo County Planning Department			*	San	Mateo County Plann	ing Department						
15 🔩	Romtec to Respond to Planning Department Comments	10 days	Wed 10/23/24	Tue 11/5/24	14	DBE				+	DBE							
16 🔩	Romtec to Revise and Submit for the Record Based on Planing Department Comments. No Additional Review or Comments	15 days	Wed 11/6/24	Tue 11/26/24	15	DBE					DBE							
17 🔩	Owner to Release Romtec Into Production for the	1 day	Wed 11/27/24	Wed 11/27/24	4 16	County of San Mateo - Project Management					County	of San Mateo - Project	t Managment					
18 📑	Submit for Demolition Permit	41 days	Wed 11/27/24	Wed 1/22/25	16	wanagment						1						
	,					,	I _ I										I	
	Task		Summa	iry		Inactive Milestone	Duration-only		Start-only	С	External Milestone	♦ Manu	ual Progress					
Project: Romte Date: Thu 5/23	2C Design, Supply Split Milestone	•	Project	Summary Task	00	Inactive Summary	Manual Summary F	Rollup	Finish-only External Tasks	3	Deadline	÷						
	mestone	•	mactive					- •	Page 1									
									raye i									

Docusign Envelope ID: A3228B90-A416-4991-8459-8F8515CCF10B

SCHEDULE CHART

ID	Task	Task Name	Duration	Start	Finish	Predecessors	Resource Names		Qtr 3, 2	024	•	6	Qtr 4, 2024			Qtr 1	, 2025		Qtr 2, 2
19	Mode	Romtec to Submit Approved Design Development Drawing for Demolition Permit Review	1 day	Wed 11/27/24	Wed 11/27/24	16	DBE	May	un	Jul	Aug	<u>Sep</u>	Oct	Nov	DBE	<u> </u>	Jan Fé	<u>b M</u>	ar
20		County Building Department to provide Demolition Approval with Comments	6 wks	Thu 11/28/24	Wed 1/8/25	19	San Mateo County - Building Department	2									San Mateo Coun	ty - Building D	Department
21	->	Romtec to Revise and	10 days	Thu 1/9/25	Wed 1/22/25	20	DBE - Project									1	DBE - Proj	ect Manager	
22		100% Construction	56 days	Wed	Wed 2/12/25	16	wanager								r 				
23		Documents Romtec to provide	1 day	11/27/24 Wed	Wed 11/27/24	16	DBE								DBE				
		"Stamped & Seal" Plans fo Building Department Review	r	11/27/24															
24		Building Department Review and Approval/Permit with Comments. No Additional Changes from County Building Department.	8 wks	Thu 11/28/24	Wed 1/22/25	23	County of San Mateo - Building Department										County of	San Mateo - E	Building Depa
25	-	Romtec Respond and Revise Sealed Plan Set. Romtec to Provide For the Record Plan Set to Building Department	15 days	Thu 1/23/25	5 Wed 2/12/25	24	DBE - Engineering										*	DBE - Enginee	ring
26	-5	Romtec Production & Delivery	77 days	Thu 11/28/24	Fri 3/14/25	17									r				
27	-5	Production of Romtec Building Packages	15 wks	Thu 11/28/24	Wed 3/12/25	17	DBE - Production Team	n							•			I	DBE - Product
28	-5	Final Delivery of Romtec	2 days	Thu 3/13/25	5 Fri 3/14/25	27	DBE												DBE
29	-5	Mobilize and Install	93 davs	Mon 4/7/25	Wed 8/13/25														_
30		Customer Provides Mobilization Payment No Later Than	1 day	Mon 4/7/25	Mon 4/7/25		County of San Mateo - Contracting												L C
31		Construction Start Date - Cre to Mobilize, Installation Tree Protection and Errosion Control	w 2 days	Tue 4/8/25	Wed 4/9/25	30	DBE - Install Crev	w											
32	-5	Demolition & Site Utility Wor	k 15 days	Thu 4/10/25	Wed 4/30/25	31	DBE - Install Crev	w											*
33		Romtec have Permits and Begin Construction - Rough in Footings, Foundation, and Slab No Later Than	10 days n,	Thu 5/1/25	Wed 5/14/25	32	DBE - Install Crev												
34	-5	Erection of Masonry Walls	10 days	Thu 5/15/25	Wed 5/28/25	33	DBE - Install Crev	w											
35		Roof Framing, Stone Siding and Sidewalks	10 days	Thu 5/29/25	Wed 6/11/25	34	DBE - Install Crev	w											
36	-5	Complete Stone Siding, Hard Siding and Soffits	i 5 days	Thu 6/12/25	6 Wed 6/18/25	35	DBE - Install Crev	w											
37		Plumbing, Mechanical and Interior Paint	10 days	Thu 6/19/25	Wed 7/2/25	36	DBE - Install Crev	w											
38	-5	Complete Interior Paint and Exterior Paint	5 days	Thu 7/3/25	Wed 7/9/25	37	DBE - Install Crev	w											
39	-5	Final Grade	10 days	Thu 7/10/25	6 Wed 7/23/25	38	DBE - Install Crev	w											
40	->	Installation of Accessories, Testing and Cleaning of Buildings	10 days	Thu 7/24/25	5 Wed 8/6/25	39	DBE - Install Crev	w											
41	-5	Walk through with County ar Complete Punchlist	nd 5 days	Thu 8/7/25	Wed 8/13/25	40	DBE & County of San <u>Mateo Rep</u> .												
42	->	County Parks to Provide Signoff/Acceptance	0 days	Wed 8/13/25	Wed 8/13/25	41	DBE - Install Crev	w											
Proje Date	ect: Romteo : Thu 5/23,	c Design, Supply /24		Summ	ary I Summary I		Inactive Milestone Inactive Summary	Durat Manu	ion-only al Summary Rollup		Start-only Finish-only	y netes	C 3	External Mile Deadline	estone	>	Manual Progre	55	
		Milestone	•	inactiv	e idsk		ividhual Lask	Manu	a summary	•	∎ Externai la	Page 2		Progress					
												i uge z							

2tr 2, 2025	Mair	l tur	Qtr 3, 2025			6	Qtr 4, 2025
Apr	мау	Jun	Jui		Aug	Sep	Uct
nt							
Department							
duction Team							
County of Sa	n Mateo - (Contracting		I			
DBE - Install	Crew						
DE	BE - Install	Crew					
-	DBE -	Install Crew					
	¥	DBE - Install C	rew				
	ì	DBE - Ir	stall Crew				
		DBE	- Install Crew				
				_			
			DBE - Install	Crew			
			DBE - Ins	tall Cre	w		
			DE	BE - Ins	tall Crew		
			•	DI	3E - Install Cre	N	
				+	DBE & Count	y of San	Mateo Rep.
					0/13		
					0/13		
-							

Exhibit 7

SCHEMATIC DESIGN DOCUMENTS







SAN BRUNO MOUNTAIN PARK SAN BRUNO, CALIFORNIA

0 2024 ROMTEC, INC. ALL RIGHTS RESERVED. THESE PLANS AND DRAWINGS MAY NOT BE REPRODUCED, ADAPTED OR FURTHER DISTRIBUTED, AND NO BUILDINGS MAY BE CONSTRUCTED FROM THESE PLANS, WITHOUT THE WRITTEN PERMISSION OF ROMTEC, INC.





SAN BRUNO PARK: FLOOR PLAN



JOB CREATED: 5/15/2024 - LAST SAVED BY: jeschonauer ON 5/15/2024 - LAST PRINTED: 5/15/2024 - LOCATION: O:\Engineering RI\Sales Drawings\Projects\2000 - 2499\2411- San Bruno1 🛱 2411- San Bruno-2999, CA 240514.dug
Docusign Envelope ID: A3228B90-A416-4991-8459-8F8515CCF10B

SAN BRUNO PARK: ELEVATIONS



PRELIMINARY THESE PLAN VIEW AND ELEVATION DRAWINGS ARE A PRELIMINARY ARCHITECTURAL REPRESENTATION OF THE BUILDING. ALL DIMENSIONS FEATURES AND COMPONENTS SHOWN ON THESE PRELIMINARY DRAWINGS MAY OR MAY NOT BE PART OF THE QUOTE. PLEASE REFER TO THE "SCOPE OF SUPPLY AND SERVICES" LETTER PROVIDED WITH YOUR QUOTE FOR ROMTEC'S PROPOSED SCOPE OF SUPPLY



FURTHER ROMTEC, TED OR F PLANS AND DRAWINGS MAY NOT BE REPRODUCED, ADAP TED FROM THESE PLANS, WITHOUT THE WRITTEN PERMIS: THESE ЩŚ RESERV MAY BE HTS GS RIGH LDIN(ALL BUI C, IN AND Щ Ш I RON) 2024 | ISTRIF

NC

METAL ROOFING-"ROOFING SHALL HAVE A CLASS ASTM E108/UL 190 CLASS A FIRE RESISTANCE, TYP

		SH	EET	NO.
--	--	----	-----	-----

REVISIONS:

MOUNTAIN PARK

BRUNO

Z ₹

ທີ

PROJECT #:

DRAWN BY:

DATE:

IFORNIA

∏ ∢

υ

BRUNO

Z ₹

ທີ

DATE:

< NEN

<u>S</u>

ELEVATION HEET TITLE

2411

JS

BY:

02

5/14/24

Docusign Envelope ID: A3228B90-A416-4991-8459-8F8515CCF10B

SAN BRUNO PARK: ADD/ALT 1



PRELIMINAR THESE PLAN VIEW AND ELEVATION DRAWINGS ARE A PRELIMINARY ARCHITECTURAL REPRESENTATION OF THE BUILDING. ALL DIMENSIONS FEATURES AND COMPONENTS SHOWN ON THESE PRELIMINARY DRAWINGS MAY OR MAY NOT BE PART OF THE QUOTE. PLEASE REFER TO THE "SCOPE OF SUPPLY AND SERVICES" LETTER PROVIDED WITH YOUR QUOTE FOR ROMTEC'S PROPOSED SCOPE OF SUPPLY.



RIGHTS RESERVED. THESE PLANS AND DRAWINGS MAY NOT BE REPRODUCED, ADAPTED OR FURTHER LDINGS MAY BE CONSTRUCTED FROM THESE PLANS, WITHOUT THE WRITTEN PERMISSION OF ROMTEC, INC. IC. ALL NO BUII EC, IN

METAL ROOFING-"ROOFING SHALL HAVE A CLASS ASTM E108/UL 790 CLASS A FIRE RESISTANCE, TYP



Docusign Envelope ID: A3228B90-A416-4991-8459-8F8515CCF10B

SAN BRUNO PARK: ADD/ALT 2



PRELIMINAR THESE PLAN VIEW AND ELEVATION DRAWINGS ARE A PRELIMINARY ARCHITECTURAL REPRESENTATION OF THE BUILDING. ALL DIMENSIONS FEATURES AND COMPONENTS SHOWN ON THESE PRELIMINARY DRAWINGS MAY OR MAY NOT BE PART OF THE QUOTE. PLEASE REFER TO THE "SCOPE OF SUPPLY AND SERVICES" LETTER PROVIDED WITH YOUR QUOTE FOR ROMTEC'S PROPOSED SCOPE OF SUPPLY.



N ADAPTED OR FURTHER ERMISSION OF ROMTEC, BE REPRODUCED, JT THE WRITTEN PE S. WITHOU THESE PLANS AND DRAWINGS A STRUCTED FROM THESE PLANS ЩÔ ₩ B MAXES RIGHTS LDINGS I ALL BU) 2024 ROMTE ISTRIBUTED.

METAL ROOFING-"ROOFING SHALL HAVE A CLASS ASTM E108/UL 790 CLASS A FIRE RESISTANCE, TYP



PARK